ENVIRONMENTAL AND SOCIAL IMPACT ASSESSMENT(COMPREHENSIVE PROJECT REPORT) FOR THE PROPOSED KAMARIGA OMENA DRYING FACILITY, WEST UYOMA WARD OF RARIEDA SUBCOUNTY, SIAYA



CPC- KENYA CLIMATE SMART AGRICULTURE PROJECT, SIAYA COUNTY, P.O. BOX 4-40600, SIAYA

APRIL 12, 2021

Certification

This is to certify that the Lead Expert hereunder as per the requirements of the National Environment Management and Coordination Act, 1999, carried this Environmental and Social Impact Assessment.

Name of The Expert/ Firm of Experts	NEMA Reg.NO.	Contact Details	Signature and Date	
Dr. Volenzo Tom Elijah	7332	+254724222154	Jakin_	

Proponent

I, the undersigned, certify that all kinds of information provided for the purpose of this Environmental Impact Assessment project report, without any prejudice, is true to the best of my Knowledge

Willis Atiang

Project Coordinator,

KSCAP, Siaya County

Date and Seal: April 12. 2021

Certification	i
TABLE OF CONTENTS	ii
List of plates/ photos	vii
List of Acronyms	viii
CHAPTER ONE: INTRODUCTION	1
1.1 Project Background	1
1.2. Purpose and Nature of the Project	2
1.4. Methodology	2
1.4.1. Environmental Screening and Scoping	3
1.4.2. Desktop Study	3
1.4.3. Site Assessment	3
1.4.4. Public Consultation	3
1.4. 6. Questionnaires	4
1.5. Data Analysis	4
1.7. Organization of the ESIA Report	4
2.0. NATURE OF PROPOSED PROJECT AND PROPOSED ACTIVITIES	6
2.1. Location and Scale of Project	6
2.1.1: Project Construction works	7
2.1.2 Construction Activities	7
2.1. 3 Methods of drying	9
2.1.3 Sanitary facilities	13
2. 1. 4 Storage	13
2.1.5 General Approach to Site Management	13
3.0 ENVIRONMENTAL AND SOCIAL BASELINE INFORMATION	15
3.1 Environmental conditions	15
3.1.1. Topography	15
3.1.2 Soils	15
3.1.3 Hydrology	16
3.1.4 Climate	16
3.2.1 Land Use patterns	18
3.2.2. Livestock Production	18
3.2.5. Industrial Activities	19
3.2.6. Tourism	19
3.2.7 Education	19
3.2.8 Religious beliefs and cultural practices	20
~ •	20
	-

3.2.9. Housing and house types	20
3.2. 10. Vulnerable and Marginalized Groups (VMGs)	20
3.2.11. Land tenure	20
3.2.12. Communication	20
3.2.13. HIV/AIDS prevalence, Knowledge, Attitudes and Practices	21
3.2.14. Energy sources and their accessibility	21
3.2.15. Water Access and Sanitation	21
3.3 Conflict and Grievance Resolution Mechanism (GRM)	22
4.0 ANALYSIS OF PROPOSED PROJECT ALTERNATIVES	23
4.1 Site alternatives	23
4.2. No project scenario	23
4.4. Scale and demand alternatives	24
5.0. POLICY, LEGAL AND ADMINISTRATIVE FRAMEWORK	25
5.1 General overview	25
5.2. National Policies	25
5.2.1 Vision 2030	25
5.2.2 Sessional Paper No. 10 of 2012 on Kenya Vision 2030	26
5.2.3. National Policy on Water Resources Management and Development	26
5.2.4 Kenya Youth Development Policy 2019	26
5.2.5. National Gender and Development Policy, 2011	26
5.2.6 National Policy for Prevention and Response to Gender Based Violence 2014	26
5.2.7 Beach Management Units (BMU) Regulations, 2017	27
5.2.9 Kenya climate Smart Agriculture Strategy 2017-2026	27
5.2.10 Agricultural Sector Transformation and Growth Strategy 2019-2029	28
5.3. Legal framework	28
5.3.1 Constitution of Kenya 2010	28
5.3.2 The Environment Management and Coordination Act, 1999 (Revised 2015)	29
5.3.3: The Environment (Impact Assessment and Audit) Regulations, 2003	29
5.3.4. Environment management and coordination (waste management) regulations 2006	30
5.3.5. Environment management and coordination (water Quality) Regulations 2006	30
5.3.6 Environmental Management and Coordination (Conservation of Biodiversity, Access Genetic Resources and Benefit Sharing) Regulations 2006	; to 31
5.3.7. Environmental Management and Coordination (Wetlands, River Banks, Lake Shore Sea Shore Management) Regulation, 2009.	s and 31
5.3.8. The Fisheries Management and Development Act No.35 of 2016	31
5.3.9. The Public Health Act (Cap. 242)	32
5.3.10. The county Government Act No. 17 of 2012	32
5.3.12. The Penal Code (Cap. 63)	33

5.3.13. Occupational Health and Safety Act 2007	33
5.3.14. The Land Planning Act (cap 303)	33
5.3.15. Physical Planning Act, 1999	34
5.3.16. The Wildlife Conservation and Management Act, Cap 376	34
5.3.17. The Land Registration Act, 2012	34
5.3.18. Labour Laws of Kenya including Employment Act 2007	35
5.3.19. The Sexual Offences Act, 2006	35
5.3.20. The Standards Act Cap 496	35
5.3.21. The Energy Act, 2019	36
5.3.22. National Construction Authority (2011)	36
5.4. World Bank Environmental and Social Safeguards	36
5.4.1 OP/BP 4.01 (Environmental Assessment)	36
5.4.2. OP/BP 4.04 Policy on Natural Habitats	37
5.4.3. Pest Management (Operational Policy, OP/BP 4.09)	37
5.4.4 OP/BP 4.11 (Physical Cultural Resources)	37
5.5. Relevant International Conventions and Treaties	38
5.5.1. International Convention on Biological Diversity (1992)	38
5.5.2. Ramsar Convention 1971	38
5.5.3. The United Nations Framework Convention on Climate Change (UNFCCC)	39
5.6. Institutional framework	39
6.0. PUBLIC PARTICIPARTION AND STAKEHOLDER CONSULTATION	40
6.1 Overview	40
6.2. Methodology used in the Consultative Public Participation	41
6.2.1 Stakeholders Public Meetings	41
6.4 Analysis of public consultation and participation	
7.0 ANTICIPATED IMPACTS AND MITIGATION MEASURES	44 лл
7.2 Impacts during the construction phase	
7.2.1 Anticipated positive impacts	
7.2.1 Creation of employment opportunities	44
7.2.2 Demand for locally available construction materials	44
7.2.3 Improved infrastructure through development of foot paths	44
7.3 Anticipated Negative Environmental impacts	44
7.3.1 Hydrological impacts	44
7.3.2. Landscape disturbance, erosion and Vegetation loss	45
7.3.3 Surface Run-off Management	45
7.3.4. Noise and Vibration Generation	45
7.3.5 Air pollution and Aerial Emissions	46

7.3.6 Dust generation	46
7.3.7 Effluents and pollution of water sources	46
7.3.8. Aerial emissions	47
7.3.9 Oil spills/Fuels and Lubricants	47
7.3.11 Limitations of access and interruption of services	48
7.4 Anticipated negative Social and health impacts during construction phase	48
7.4.1. Labour Influx Effects	48
7.4.2 Spread of COVID-19 amongst community members during consultations and	
construction	48
7.4.3 Gender based violence and sexual harassment (GBV/SH)	49
7.4.4 Child Abuse	50
7.4.5 Sexual Exploitation and Abuse (SEA)	50
7.4.6 Risk of Increased incidences of HIV/AIDS and STIs	51
7.4.7. Increased Occupational health and safety (OHS) risks	51
7.4.8. Risk of Accidents	52
7.4.9 Risks of increased spread of COVID-19 at work sites	52
7.5 Impacts during the operation phase	53
7.5.1 Anticipated positive impacts during operation phase	53
7.5.2 Reduced post-harvest losses	53
7.5.3 Resilience building in the community	53
7.5.4 Improved food and nutritional security	53
7.5.5. Livelihood diversification	53
7.5.6 Biodiversity conservation and enhancement	53
7.5.7 Improved beach hygiene and public health	54
7.5.8 Enhanced food safety	54
7.6 Anticipated negative environmental impacts during operation phase	54
7.6.1. Health risks and Environmental contamination from pesticides	54
7.6.2. Increased water usage	54
7.6.3 Increased energy use	55
7.6.4 Fire	55
7.6.5 Solid wastes	55
7.6 Anticipated negative social impacts during operation phase	55
7.6.1 Gender based violence and sexual harassment (GBV/SH)	55
7.6.2 Sexual Exploitation and Abuse by project workers against community members	56
7.6.3 Risk of Increased incidences of HIV/AIDS and STIs	56
7.6.4 Grievances/conflicts	57
7.6.5 Child abuse	

7.6.6 Risk of increased spread of COVID-19	58
7.6.7 Impacts related to occupational and public/community safety and health	58
7.7. Project Lifespan and Decommissioning	60
7.4.1 Negative Environmental and Social impacts and Mitigation measures during decommissioning	60
7.5. Positive Impacts during decommissioning	61
8.0. ENVIRONMENTAL AND SOCIAL MANAGEMENT AND MONITORING PLAN	62
9.0 CONCLUSION AND RECOMMENDATIONS	80
9.1 Conclusion	80
9.2 Recommendations	80
9.2.1 Compliance with the Water Quality Regulations	80
9.2.3 Emergency Response Plan (ERP)	81
References	82
APPENDIXES	83
Appendix I: Environmental Social Management Framework Checklist and Appraisal	83
Appendix II : Assessment Checklist and Findings	87
Appendix III: Practising Licences	90
Appendix IV : Ground Layout Of The Proposed Kamariga Fish Drying	91
Appendix V: Evidence of landing ownership title deed	93
Appendix VI: Record of Issues (FGD) Meetings	94
Appendix VII: Evidence of Public Consultation (Attendance and Sample Questionnaire).	95
Appendix VIII : Minutes Of Community Validation Barazaon Proposed Project Planning	105
Appendix IX : Key Informant (KI) Interview Schedule	108
APPENDIX XI : chance Finding Procedures	109
APPENDIX XII: Disclosure of ESMP at the Proposed Site	111

List of Tables

Table 1: Omena Drying Machine Specifications	10
Table 2 Agro -Ecological Zonation Of Siaya County	16
Table 3: Population and Key Indicators Welfare in Proposed Project Area Uyoma	17
Table 4: Housing type in the Project Area	20
Table 5 : ACCESS Ownership and communication	21
Table 6 : Energy and their Sources in Subproject Site, Central Sakwa	21
Table 7: summary of major issues during public participation	43
Table 8: Summary of Environmental and Social Management Monitoring Plan- Kamariga BMU fish	
drying plant	63

List of plates/ photos

Plate	1: Surface runoff soil types and sanitation Challenges in the proposed site	8
Plate	2 :LANDING SITE KAMARIGA	14
Plate	3: LAKE SHORE EROSION AT LANDING SITES	15
Plate	4: Housing settlement and sanitation challenges near Kamariga beach	18
Plate	5: FLORA AND FAUNA IN THE PROPOSED PROJECT SITE	19
Plate	6: A focus Group Discussion at BMU office, this was done before Covid 19 outbreak	40
Plate	7 : Community engagement at the proposed project site	40
Plate	8: Community member Group Photo After consultation	42

List of Acronyms

AEZ	Agro- Ecological Zone
AIDS	Acquired Immune Deficiency Syndrome
BMU	Beach Management Unit
CBO	Community Based Organization
C-ESMP	Contract specific Environment and Social Management Plan
CESSCO	County Environment and social Officer
CIDP	County Integrated Development Plan
CITES	Convention on International Trade in Endangered Species
COVID 19	Corona Virus disease
CPP	Consultation and Public Participation
CPU	County Project Coordination Unit
CSA	Climate-Smart Agriculture
EA	Environmental Audit
FHS	Environmental Health and Safety
FIΔ	Environmental Impact Assessment
EMCA	Environmental Management and Co-ordination Act
EMCA	Environmental Management Plan
EMS	Environmental Management System
	Environmental and Social Impact Assessment
ESIA	Environmental and Social Management Plan
	Environmental and Social sofoguards
	Emorgoneo Posponso Dion
CRV	Conder Based Violence
CDP	Gross Domestic Product
CHC ^o	Groenhouse georg
	Covernment of Venue
U.O.K	Howerd Analysis Critical Control Daint
HALLP	Hazard Analysis Chucal Control Point
GPS	Uumon Immuno deficiency Virus
HIV	Human Immunodeficiency virus
	International Labour Organization
KUSAP	Kenya Chimate Smart Agriculture Project
KBS VES	Kenya Bureau ol Standards
NF5 KWC	Kenya Forest Service
KWS	Kenya wildlife Service
Mg/L	Milligrams per Litre
MOU	Memorandum of Understanding
NEMA	National Environment Management Authority
NGUS	Non-Governmental Organizations
NLC	National Land Commission
NOX	Nitrogen Oxides
OSHA	Occupational Safety and Health Act
PAPs	Project Affected Persons
PCB	Polychlorinated biphenyls
PCR	Physical Cultural Resources
PPE	Personal Protective Equipment
KAP	Relocation Action Plan
SEA	Sexual exploitation and Abuse
SH	Sexual Harassment
SOP	Standard Operating Procedure
TIMPs	Technology, Innovation and Management Practices

W.H.O	World Health Organization
WRA	Water Resources Authority

Executive Summary

Kenya is predominantly an agro based economy where small-scale farmers dominate with about 75% of the populations' livelihoods directly linked to agriculture. Agriculture is thus key to the overall national development, equity objectives and sustainable growth. Intuitively, weather-related disasters, particularly droughts, present a major challenge to the predominant rain fed agricultural production system with profound adverse impact on the economy. The adverse effects negatively affect foreign exchange earnings, food security and nutrition, employment and rural livelihoods. Adaptation to extreme weather impacts is thus a priority under Kenya's National Adaptation Programme Action plans (NAPAs (2018-2030) and updated National Determined Contributions . Among other objectives, NAPAs envisages improved crop productivity through irrigation. Building farmer resilience to climate change risks is the main objective under the Agricultural Sector Transformation and Growth Strategy, which in agriculture operationalize the Kenya's climate change Act.

The Kenya Climate Smart Agriculture Project (KCSAP) under the Support of World Bank (WB) intends to increase agricultural productivity and build resilience to climate change risks in the targeted smallholder farming and pastoral communities in Kenya, and in the event of an eligible crisis or emergency, to provide immediate and effective response. The Project activities will contribute to these objectives by up-scaling Climate-Smart Agriculture (CSA) Practices and supporting smallholder farmers to adopt integrated climate-smart Technology, Innovation and Management Practices (TIMPs) and also support investments through a community driven development approach in smallholder agro-pastoral production systems as well as extensive pastoral production systems in the selected 24 counties of Kenya. It's against this backdrop that the County government of Siaya, one of the recipient counties is seeking to undertake Environmental and Social Impact Assessment for sub-projects which fall under the second Schedule of Environment Management Act (EMCA).

The proposed project is categorized as a Low-Risk Project under the Legal Notice No. 31, Legislative Supplement No. 16 published in the Kenya Gazette Supplement No. 62 on 30th April 2019 i.e. Amendment of the Second Schedule which lists the projects to undergo EIA [Section 58 (1) of EMCA, 1999 (Cap. 387) Revised 2015]. The proposed project falls under the category described as 'Medium scale processing and manufacturing industries including-fish processing plants' [Sub-section 1 (c) of this Legal Notice] as it focusses on construction of Omena drying facility. The proposed project on the basis of its potential to pose both environmental and social impacts requires undertaking of an ESIA before it is implemented in accordance to Section 31 (3) (a) (i) and (ii), of the Environmental (Impact Assessment and Audit) Regulations, 2003. The main purpose of the Comprehensive Project Report is to assist the Proponent, NEMA and all other stakeholders in understanding the proposed project and its potential environmental consequences and thus provide a basis for making informed decisions on the project.

. The objective for this is to integrate environmental and social concerns in the project planning and implementation processes. This ESIA has considered all the relevant legal, policy and institutional framework, key among them; the World Bank Environment and Social Safeguards Policies, the existing environmental regulatory framework EMCA Cap 387 and the Environmental (Impact Assessment and Audit) Regulations of June 2003, Occupational Health and Safety Act (2007), the Physical planning Act, County Government Act, wastes disposal regulation of 2006, environmental standards, and sustainable use of natural resources principles. Other relevant legislations to this ESIA that were considered include the public health, physical planning, land planning Acts and gender promotion, HIV/AIDS prevention and control Act, and sexual offences Act.

On methodology, a screening exercise was done to ascertain whether, the project warrants an ESIA and it was found to be necessary. A general public meeting (baraza) was held on 4th of March 2020 and 62 persons attended. This was followed by focused group discussions with 32 persons. Key Informant Interviews were conducted for a total of 10 persons (lead agencies) from the following institutions, Kenya Wildlife Service (KWS), Kenya Forest Service (KFS), County Environment Officers, Department of Agriculture, National Museums of Kenya (NMS), Governor office/ Ward office, WRA (Water Resources Authority), Water Department, Fisheries Department, Chief- Kagwa location The proposed project is basically involves setting up a new drying facility and auxiliary equipment such as the compressor.

The project will cost KES 41,820,000 funds requested from KCSAP was KES 37,320,000, while community contribution is KES Ksh 4,500,000

Though the analysis from the assessment reveals positive livelihood and environmental impacts, a number of social risks such as sexual exploitation of women are likely. Minor environmental impacts such as solid waste, public health and safety, emissions to air and increased energy and water consumption are also anticipated. The proponent who is Kamariga Omena vendors/ fishers on behalf of Kamariga Beach Management Unit (BMU), as agreed during the public participation is expected to review and incorporate agreed changes in the procurement plan, as well coordinate and oversee the implementation of the outlined mitigation actions/ measures

The main issues and concerns discussed during public consultation and meetings relate to Gender based violence (GBV), sexual exploitation, lake shore erosion, employment of the locals by the contractor and design issues such as effluent control. These issues were discussed by various stakeholders including the project engineer who was tasked with the revision of the bill of quantities to incorporate the need for lake shore protection.

The following negative impacts are likely to occur during various stages the proposed Omena drying facility project implementation are : loss of vegetation, increased surface runoff soil erosion. noise levels, air pollution, dust generation and solid wastes are likely to increase. Social impacts such as gender based violence and child labor, HIV/AIDs prevalence is likely to increase. High risks of contracting the dreaded Covid 19 virus., contamination from pesticides, increased use of energy, risks of accidents and fire outbreaks Some of the proposed mitigation measures to the impacts are; undertaking landscaping and planting grass and indigenous trees where there is destruction of vegetation, use of less vibrating machines to reduce the noise levels and encourage workers to wear ear masks, to control noise pollution. Dust arresters must be placed all over to and face masks be provided to the workers in the factory. On Occupational Health and Safety the proponent shall provide personal protective equipment's such as suitable gloves, footwear, and googles and head coverings. The proponent shall provide solid waste collection facilities and sensitization of construction workers on proper disposal of solid wastes. Temporary latrines will be provided on site to be used by construction workers Oils and greases emanating from repair and maintenance activities will be collected in containers to avoid entry into local drainage channels. On Covid 19 the proponent shall ensure adherence to health guidelines provided like wearing of face masks, social distance, hand washing, use of sanitizers. The proponent shall put in place solar panels to provide energy to run the entire facility. Fire extinguishers shall be put in all areas that have a potential for fire to erupt accidentally. The CPCU, BMU and the contractor will in coordination with the local leadership undertake community awareness on GBV and sexual exploitation and put in place grievance redress mechanisms(GRM) for tracking and resolving any emerging issues during the project implementation. The proponent shall ensure that labour laws are strictly followed to avoid issues of Child labour and policies on gender mainstreaming and HIV/AIDs will be adhered strictly followed

The review of this ESIA is undertaken during the era of the Corona virus disease 2019 (COVID-19) pandemic outbreak. As such, specific mitigation measures have been introduced to prevent the spread of the pandemic during the construction period. Moreover, consultations are required to adhere to all COVID-19 the mitigation measures, such as during training on E&S issues, to reduce risk of infection to communities. Such measures include adherence to Ministry of Health Standard operation procedures (SOPs) on social distancing, open air congregation, and use of masks, hand washing and limiting the number of participants per activity.

Based on the assessment, therefore the subproject is recommended for approval by the National Environment Management Authority (NEMA). The conditional license will be tracked through regular monitoring on the implementation of the ESMP, and undertaking of annual environmental and social audits after operating for one year. The implementation of ESMP will cost Ksh. 1.221M and shall be part of the project contract. The proponent should share the ESMP with the selected Contractor and the latter will be required to develop and implement a Contractor-Specific ESMP (C-ESMP). The CPCU will follow up and monitor implementation of the ESMP. The CPCU/ CESSCO, contractor, the supervising water engineer, the proponent and the Siaya county environmental committee will be required to ensure that the mitigation measures proposed for the construction, operation and decommissioning phases in the ESMP are followed.

CHAPTER ONE: INTRODUCTION

1.1 Project Background

The overall national development objectives of the Government of Kenya are accelerated economic growth and rising productivity of all sectors, equitable distribution of national income, alleviation of poverty through provision of basic needs, enhanced agricultural production, industrialization, accelerated employment creation and improved rural-urban balance. These objectives are facilitated through a myriad of interventions that are funded through local (central government and devolved system) and international resource mobilisation (bilateral and multilateral as well as grants). The extent to which these objectives can be realized on a sustainable basis and in an environmentally sound manner is dependent on the degree and economic efficiency with which critical factors of production are made available and combined with each other to produce the desired results that do not compromise the wellbeing of future generations.

Small-scale fisheries in developing countries play a vital role in contributing directly to food and nutritional security, poverty reduction, wealth creation, foreign exchange earnings and rural development. Such fisheries form part of the livelihood safety net and basic food security for many poor fisherfolk. Caridina niloticus by virtue of its high protein content is also an important ingredient in the animal feed industry. Omena harvesting is one of fishery activities dominated by youths and women hence an important source of employment and avenue for their participation in the economic development in beaches. According to the information gathered at the beach management unit it was reported that 4 - 6 youths are engaged in every fishing boat on a daily basis and 6 - 8 women are responsible for offloading the fish catch, washing and drying it. Post-harvest losses are however greatest challenge in this sector and such loss translates directly to reduced income and benefit at household level, as well significant reduction of the Omena fish availability to its consumers.

Lake Victoria is the most important source of fish in East Africa and the biggest source of freshwater fish on the African continent, and is critical in the livelihoods of fisherfolk and national development at large. The lake is also important in conservation terms because of its biodiversity. In 2017, Siaya County produced 31,945.4 Metric tons of fish valued at Kes 4.174 B. This was 12% increase in production compared to the previous year 2016 (28,392 Metric tons). Of the 12 food fish species produced from the lake, *omena* and Ochong'aa account for 40%, Nile perch 32% and tilapia 18%. Evidently, Omena and Ochong'aa are leading fish species with Kamariga, Osindo and Misori from West Uyoma beaches of Rarieda Sub County being the main landing sites for the two species.

Fish is an extremely perishable food commodity. It has been estimated that 10 percent by weight of world fish catch is lost by poor handling, processing, storage and distribution. However, losses in small-scale fish processing are said to be particularly high and figures as high as 40 percent are sometimes reported (FAO, 2010). The freshwater omena and or ochon'ga are processed by sun-drying on nets spread by the lake shore. On a sunny and windy day

therefore, the fish takes 2-3 days to dry (pers communication with key stakehlders). In cloudy and less windy days, the drying process may not be attained even after one week. Rainy days complicate as it increases the risks of fish being washed away by storm waters, rotting and contamination. High human population at the fish landing sites vis-a viz inadequate sanitary facilities also increases the risk of cross-contamination. The eating of the fish by birds and dogs during the drying occasions heavy reduction in volumes. Rancid oxidation and discoloration of fish also occurs in absence of adequate sunlight. Rancid products are reddish, dirty, and smelly with bitter taste reducing its appeal to customers and is sold for lower prices and or diverted to animal feed industry.

Sun-drying essentially require large open fields which cannot be sustained with the ever increasing human population at fish landing sites and associated settlements. In the absence of the sun and winds, freshly landed omena would rot and processor may at-best end up with discoloured products exhibiting off-odours. During such cloudy and rainy days, losses of up to 80% occur. Further, wastes are generated by human activities at the landing sites which compromise the safety of the dried fish product owing to poor hygiene and sanitary facilities at the landing sites. Consequently, omena fish harvested from L. Victoria suffers high post-harvest losses as a result of the rudimentary weather-dependent drying processes. Inadequate hygiene and sanitation facilities at the omena fish landing sites further compound this problem leading to huge physical and nutritional losses. Predation by birds, dogs and insects during drying also accounts for significant losses in volumes.

Post-harvest losses are equated to a loss of both valuable animal protein for consumers and loss of income for fisher folks, processors and traders. Any initiative geared towards reducing such losses contributes to the development goal of the fisheries sector. Considering that L. Victoria produces over 80% of the annual fish production nationally, its contribution to the Gross County Product of Siaya County and the GDP cannot be overstated. This presents an impetus to urgently institute measures that could facilitate transfer of innovative technologies to reduce post-harvest losses of this prime fish commodity. This has potential to contribute to food security as envisaged through enhanced fisherfolk income and food safety.

1.2. Purpose and Nature of the Project

The purpose for the proposed Kamariga Omena Drying Facility is to improve quality and quantity of fish/Omena production and it's by products for improved income, better market access and reduction of post-harvest losses. The Kamariga omena processing units of facility targets all the actors within the omena value chain (fishermen traders, boat owners net menders, transporters and other stakeholders). These are estimated to be about eight thousand two hundred (8,200) and among them about five thousand will be direct beneficiaries considering that 80% of the members of the community consume Omena at least 4 times a week due to its availability and affordability. The indirect beneficiaries include transporters and traders of the processed Omena whose shelf life, perishability and reliability will be enhanced.

1.4. Methodology

The methodologies used included a screening exercise to identify pertinent issues for coverage in line with the TOR and to complement the World Bank EMSF screening checklist findings

(Appendix I). The ESIA process included: Collecting information, evaluating the information and compiling the findings for use in project planning, implementation, decommissioning, Monitoring and evaluation. Primary data collection was done through; a general public meeting (baraza) held on 4th of March 2020 and 62 persons attended. This was followed by focused group discussions with 32 persons.

Key Informant Interviews were conducted for a total of 10 persons (lead agencies) and administered questionnaires (Sample provided in Appendix VIII), holding 4 separate focus group discussions (Youth, Women, Men and community leadership), (Appendix X) and two communities public barazas of (Appendix VI and IX). List of participants is attached Appendix VII. A total of 35 questionnaires were administered with a response rate of 98%. Secondary data sources was also undertaken through a review of various Key documents. Details in regard to each methodology is as described below; site visits were done in areas where major operations and work would take place during construction and operation of the project. The findings from all the data sources has been used to inform this study.

1.4.1. Environmental Screening and Scoping

This step was applied to determine whether an ESIA was required and what level of assessment was necessary. This was done in reference to requirements of the EMCA, 1999, and specifically the second schedule and World Bank Safeguard policies. Issues considered included the physical location, sensitive issues and nature of anticipated impacts. The Scoping process helped narrow down onto the most critical issues requiring attention during the assessment. Environmental issues were categorized into physical, natural/ecological and social, economic and cultural aspects whose analysis as given in Appendix I.

1.4.2. Desktop Study

This included review of existing documents in regard and review of proposed activities, project documents, designs policy and legislative framework (specified in section 1.4.1 above), as well as the environmental setting of the area among others. This was complemented with discussions with managers and design engineers and interviews with community.

1.4.3. Site Assessment

Field visits were carried out for physical observations of vegetation, water resources, physiography, geology and soil. At the visited sites, documentation on geology, soil characteristics and landscape were recorded. Photographs at selected sites were taken for inclusion in this report to further emphasis these observations. Field visits meant for physical inspections of the site characteristics and the environmental status of the surrounding areas to determine the anticipated impacts were conducted. It also included further interviews with the community and key stakeholders.

1.4.4. Public Consultation

The ESIA experts, in consultation with KSCAP, Siaya sought the views of persons who may be affected by the proposed project. The public consultations were preceded by the identification of stakeholders and project affected persons (PAPs- appendix VI- IX and XII) and plates 10-14. Public meetings were undertaken at the proposed site and the project area (Appendix on public baraza attendance- appendix VII-VIII). The general public baraza was attended by 62 persons while FDGs was attended by 32 persons. The record of minutes is provided in Appendix (VI, VIII and IX). Evidence of attendance (photos) on 20.01.2020, 4.03.2020 and 3.06.2020 is provided in Appendix plate (11-13 and 15-16)

1.4.5 Key Stakeholder Consultation

KII were carried out with the objective of improving the understanding of the procedures and key concerns in the ESIA process in general. The KII schedule is provided in Appendix (X). A total of government 10 officers (lead agencies) were interviewed on specific issues of concern to the project implementation (administrative, legislative, policy instruments). The following were consulted

- ✓ Kenya Wildlife Service (KWS)
- ✓ Kenya Forest Service (KFS)
- ✓ County Environment Officers
- ✓ Department of Agriculture
- ✓ National Museums of Kenya (NMS)
- ✓ Governor office/ Ward office
- ✓ WRA (Water Resources Authority)
- ✓ Water Department
- ✓ Fisheries Department
- ✓ Chief- Kagwa location

1.4. 6. Questionnaires

The aim of administering questionnaires was to capture community perception on the project's benefits, potential problems and possible solutions and whether they felt the project should be implemented or not. 98 % of the questionnaires were returned (sample in appendix VIII), all in support of the sub project.

1.5. Data Analysis

The ESIA expert used past experience and knowledge, scenario building, community input and expert opinion to analyse the data from the desk studies and field visits in order to determine the potential impacts of the proposed project, the severity of effects arising from these impacts and how any adverse impacts can be best mitigated and positive impacts enhanced. This analysis provides the framework for the recommendations on mitigation measures and provides the basis for the formulation of the environmental and social management plan as the actionable output from the ESIA process. The data was considered in terms of occupational health and safety with respect to the construction and operational phases of the proposed projects, as well as sustainability concerns such as global environmental protocols and impacts. In analysis of Environmental and social impacts the following were considered (provided in the checklist on appendix II)

- ✓ The current land use and impact of proposed project
- ✓ General land use of the adjacent sites
- ✓ Sensitive area and habitats or critical habitat
- ✓ Threatened plant and animal species
- ✓ Effects on Ambient/Natural Environment
- ✓ Drainage systems and storm water flow (including pollution indicators, impacts on water flow patterns and quality aspects, user interference and contamination)
- ✓ Topography- especially landscape and soil erosion
- ✓ Water quality aspects
- ✓ Soil conditions and potential contamination, landscape/aesthetics degradation
- ✓ Drainage patterns in relation to waste water/effluents, oil spillages
- ✓ Air quality in relation to atmospheric emissions and vehicles/machinery
- ✓ Noise and vibration
- ✓ Social issues
- ✓ project alternatives considerations such as scheduling, location, demand, technology, inputs and process alternatives

1.7. Organization of the ESIA Report

The report is organized into nine substantive chapters. Following this introductory chapter, Chapter 2 discusses the Project activities. Chapter 3 gives the environmental and social baseline, chapter 4 discusses the Policy, Legal, Institutional and Administrative Framework. Chapter 5 analyses public participation and its outcomes. Chapter 6 identifies and discusses the Potential environmental and social impacts. Chapter 7 discusses the project alternatives while chapter 8 provides the ESMP and ESMP monitoring plan. Chapter 9 provides the conclusions

and recommendation. This is followed by some of the literature sources consulted (References) and Annexes.

CHAPTER TWO

2.0. NATURE OF PROPOSED PROJECT AND PROPOSED ACTIVITIES

2.1. Location and Scale of Project

Kamariga Village, West Uyoma ward, Rarieda Sub County. Kamariga beach is a gazzeted beach along Lake Victoria under provisions of Fisheries Act. The proposed project site lies within Latitude 0° 14'11.5 S and to 34° 14'58.4''E, long. The project site is on land (0.14Ha) registered under Kamariga Beach Management Unit: UYOMA/KAGWA/1900 (Appendix V).



PLATE 1: LOCATION OF KAMARIGA OMENA DRYING FACILITY

It is approached from Makasembo secondary School through an all-weather road approximately 1.3 km off the tarmac. The site is approximately 300m before the current Kamariga BMU offices. Kamariga beach is centrally placed amongst the giant Omena producing beaches of Uyoma and Sakwa WardsThe major land use in the area is fisheries, settlement and Agriculture. The proposed site borders Lake Victoria to the west, market and settlement as well as farmlands on the Eastern side. The beach has a population of more than 1500 people that directly depend on the Lake Victoria Kenya waters through fishing activities. The proposed processing facility has the capacity to process 10 Tons of fish products. The proponent is Kamariga Beach Management Unit (BMU), one of the 85 beaches in Siaya County. The BMU is duly registered under Fisheries and Management Act 35 of 2016 with an office on the shores of the lake.

2.1.1: Project Construction works

The project consists of construction of a receiving and drying area as well as installation of one drier and compressor. The project will involve:

- ✓ Construction of drier facility
- Earth works: excavations, cut and fill
- Installation of 3 phase transformer and connection to the mains
- Electrical wiring
- Installation of a drier and associated electrical wiring
- Installation of a compressor
- ✓ Electrical and civil works including sewer and waste water reticulation

2.1.2 Construction Activities

The scope of the technical works for the Proposed development ranges from excavation and earth moving, compacting and filling, reinforcement, superstructure works, carpentry, masonry, plastering, electrical works and painting. The one building whose dimensions are 15 X 12 M (Appendix IV) will be portioned into various rooms to include; Machine room, Store, Receiving room, Dispatch area, Office and Board room. The constructions works will also include 2 toilets and effluent soak pit .The project inputs will include the following;-

Raw materials

Construction raw materials includes sand, cement, stones, crushed rock such as gravel/ballast, ceramic, steel metals, roofing materials including tiles, painting materials, among others. All these will be obtained. During operation phase Omena and Ochon'ga will be the main inputs which will require energy sources and water.

Machinery and Equipment

Trucks, concrete mixers tools and other relevant construction equipment will be utilized for the removal of debris, transportation of materials, and resulting construction debris. Most of the machinery will use petroleum products to provide energy. The drier and compressor are the main machinery and equipment to be installed and which shall generate noise and vibration during operational phase.

Labour

The labour force, most of which is local will be both skilled and non-skilled workers and will require services such as energy, water supply and sanitation facilities. The contractor wills as much as possible hire local labour as provided for in the ESMP and as part of conflict minimization effort.

Water

Water is used extensively in fish processing, for fish cleaning, to flush dirt from equipment and floors, and flume the dirt to floor drains and collection sumps. This may require installation of automatic water shut-off equipment The first step in reducing water consumption is to analyse water use patterns carefully, by installing water meters and regularly recording water consumption. Water consumption data should be collected during production hours, especially during periods of cleaning. Some data should also be collected outside normal working hours to identify leaks and other areas of unnecessary wastage. Automatic control of water use is preferable to relying on operators to manually turn water off. Installing automatic shut-off equipment, such as sensors, solenoid valves, timers and thermostats, could prevent wasteful practices. The proponent will also consider water reuse with approval of a food safety officer e.g. slightly contaminated waste waters may be recycled for use in the cleaning of the premises

without compromising product quality and hygiene. The piping for the recycled water will be marked to avoid confusion with fresh water lines. **Power (Energy)**

The electric power required for Omena processing will be accessed through connectivity to the Kenya Power Limited power line that is 150 m from the proposed site. Significant reductions in energy consumption will be achieved through use of energy-efficient equipment and heat recovery systems. Use of solar energy will also be used in the drying facility

Waste management

The volume of generated effluent is an important issue in fish processing. The amount of effluent is linked closely to water consumption; therefore efforts to reduce water consumption will also result in reduced effluent generation. There will be need construct designated waste disposal receptacle in site. Domestic solid waste to be stored in refuse bins temporarily before being taken away for proper disposal by NEMA licensed waste management firms. Excavated soil shall be disposed at a minimum distance of 20 meters from the project site The BMU and project management committee will ensure there recycling of recyclable wastes such as paper, metals, and plastics by engaging firms that undertake recycling of wastes.

The Project Management unit will construct ablution block, septic tanks and appropriate drainage systems for disposal of solid and liquid wastes. The Waste disposal sites must be fenced off and always be kept clean and clear at all times to avoid breeding of flies and parasites



PLATE 2: SURFACE RUNOFF SOIL TYPES AND SANITATION CHALLENGES IN THE PROPOSED SITE

Maintenance of operations

In a fish processing, plant work areas and equipment that are in contact with fish must be cleaned and sanitised regularly to maintain hygienic conditions. Cleaning requirements are normally stipulated by public health regulations. All production areas and equipment are cleaned daily, and the floors and machinery are also rinsed during production. A common cleaning routine at fish processing plants is to first hose down equipment and floors roughly, to enhance the effect of detergents. Detergents and sanitising agents are then applied, followed by washing and scrubbing. The detergents used are normally alkaline. There is a final rinse with clean water to remove all detergent and sanitising agents, detergents and disinfectants. The water consumption for cleaning can be very high accounting for 25–40% of the total water used at a fish processing plant. The organic load contained in cleaning wastewater could thus be high. In addition, hazardous substances such as sodium hydroxide and sodium hypochlorite are sometimes used in conjunction with cleaning.

2.1. 3 Methods of drying

Drying of the Omena will done using indirect steam drying, the product will be fed continuously into a rotary apparatus containing steam-heated elements (tubes, discs, coils etc.). A counter-current stream of air will be blown through the dryer to remove water vapour. Indirect steam drying is less energy efficient than direct-fired drying. Odourous gases released from the drying process, along with those from other parts of the process, are often released from the process. Specifications of the drying macine is indicated in Table 1 below:

TABLE 1: OMENA DRYING MACHINE SPECIFICATIONS

S/NO	Item	Use/Description	Specifications	Accessories
	Receiving Table	Physical sorting of received product before	Material: Stainless Steel 304-Food	With 1 drain
		weighing on reception and changing into	Grade	and 2 standard
		trays	Overall Length: 48" x 30" wide	sinks each
	Platform	Weighing the Received Product Before	Platform size: 1200 x1200 mm	Battery charger
	Weighing Scale	Drying and After Drying for Storage and	Function: Power On, Off, Tare,	
		or Dispatch	Funk {4}	Battery
			Power Supply: Mains / Battery	
			Power: 6V / 4AH	
			Capacity: 2,000kg	
			Voltage: 6V	
	Portable Bag	Closing Chute Bags on Packaging Dried	Maximum sewing speed: 1600 to	With Oil
	Stitcher	Omena	1800 Stitches per Minute	Lubrication
			Type of stitch: Single Thread Chain	System
			Stitch	
			Voltage: 220 V	
			Dimensions: 30cm W x 33cm D x	
			30cm H	
	Automatic	Washing and sorting the Omena before	Power Source: Electric	Distribution
	Omena Cleaner	drying	Power: 2.25 Kw, SS Pure Water	Cabinet:
	Grader	Size roller sorting machine	Pump 1.1 Kw with Japan Mitsubishi	Microcomputer
		Consists of 3 parts i.e. 1 st stage conveying,	inverter	Control, SS
		2 nd stage conveying and grading drum set	Capacity: 800 kg per hour	Cabinet.
		with 2 thick and thin rollers	Voltage: 220 V	Conveyor Belt
				Material: High

		Water supply part: SS fittings on the	Strength
		cover, SS pump and PPR fittings	Polyvinyl
		Dimensions: Fuselage Frame	Chloride Plastic
		Diameter 114 X 2.5mm 304 SS	Belt
		Pipe, Lifting Groove 1.5mm 304 SS	
		Plate, Grading Cylinder Diameter	
		125 X 130mm Sliding Plastic Steel	
		Drum, Conveyor Size 1746mm	
		Wide Plastic Mesh Belt, Baffle	
		Height 10mm, 2 Stage Hoist	
		Conveyor Belt Material: High	
		Strength Polyvinyl Chloride Plastic	
		Belt	
		Warranty: 1 year Guarantee	
		Rotating Part: SS Bearing 205,	
		Shaft Drive	
Fifo Batch	Application: Drying Omena	Heating Temperature: 40 °C to 300	Drying Trays
Omena Dryer		⁰ C	and Trolleys
		Material: Stainless Steel 304	Automatic
		Power Source: Electric	Temperature
		Power for Electric Heat: 60Kw	Control System
		Capacity: 500kg Per Batch	Forced
		Voltage: 220V	Ventilation
		Trays Quantity: 192 Pieces	Adjustable Air
		Trolley Quantity: 8	Distribution
		Dimensions: 4280 X 2200 X 2270	Plates
		mm	

		304 SS Drying Trays: LXWXH:	Noise
		640 X 460 X 45 Mm, Thickness:	Elimination
		0.5mm	Thermal Stable
		Drying Trolleys: 24 Trays	Axial Flow Fan
Water Treatment	Application: Effluent handling	Conventional Tanks: 2	
Unit		High Pressure Pumps: 2	
		2 Pressure gauges and 4 valves	
		1 Flow meter	
		PPR Piping and joints	
		Power: 2 KW	

2.1.3 Sanitary facilities

Toilet facilities will be provided during construction and operation phases. The toilets will adhere to the following recommendations:

- ✓ Toilet rooms must not have direct access from the processing area. An anteroom or a 90-degree turn into the toilet room is satisfactory.
- ✓ Toilet rooms shall be equipped with self-closing doors
- \checkmark There will be separate toilets for male and females.
- \checkmark b) A floor drain should be provided in each toilet room.
- ✓ Toilet rooms must be finished with durable and impervious materials and equipped with soap and towel dispensers.

2.1.4 Storage

A dry storage area is mandatory for fish products. Where applicable, separate dry storage facilities will be identified for: i) Clean-up chemicals and materials; ii) Food additives; iii) Packing materials; and iv) Equipment not in use. All window ledges in the storage area will be sloped inward to prevent ponding of water. Packaging materials used in direct contact with fish shall be of a food-grade contact packaging material. These materials must be stored in a clean, dry area. Plate (3-8) shows the current storage and environmental challenges at the current site.

2.1.5 General Approach to Site Management

- Concrete floors should be used in processing areas with a minimum floor slope of 2.5 cm (1 in) per 2.4 m (8 ft) is. Outside concrete pads (sloped to a drain) are required for all receiving, landing and storage areas
- Drains should be of a trough design. The trough drain should be a minimum of 15 cm (6 in) in width and depth, with a minimum slope of 2.5 cm (1 in) per 1.8 m (6 ft). The distance from the drain to any wall should not exceed 4.57 m (15 ft)
- The concrete lip of the drain should be equipped with an L-shaped angle iron embedded and anchored into the concrete to protect the concrete when the gratings are being removed during plant clean up
- Drains must be equipped with traps or other devices to preclude the entry of gases or vermin into the building through the drain
- Wastewater should be discharged into to an approved onsite disposal system
- Wall surfaces (floor to ceiling) in processing areas must be durable and impervious to water.
- Plywood and concrete surfaces must be sealed and then coated with a durable enamel or epoxy finish
- Walls must be light coloured and smooth so that dirt can easily be seen and easily removed
- All window ledges must be sloped inward to prevent ponding of water
- Ceiling surfaces must be smooth, impervious and washable with no exposed pipes, joints, or open support beams
- Safety shields should be installed on all light fixtures throughout the processing area to prevent shattered glass from falling into food products.
- Processing equipment must be constructed of approved materials such as stainless steel, aluminium or plastic
- Water must be of potable quality from an approved source (i.e. disinfected/treated water supply
- A dry storage area shall be provided

- All electrical conduits must be laid before plastering
- Drainage channels will be provided leading from run-off generation areas such as car parking areas, along the proposed building
- Sanitary facilities shall be provided to cater for staff during the construction and operation period
- Solid wastes shall be collected in a waste holding station from within the project site and regularly emptied at a designated area on the compound and later released to an appointed waste disposal firm that shall transport it away from the site to designated waste disposal areas in accordance with NEMA and Siaya county regulations on waste collection, handling and disposal.



PLATE 3 : LANDING SITE KAMARIGA

There is need to protect such sites from lake shore erosion, an environmental concern of great significance in the current proposal

CHAPTER THREE 3.0 ENVIRONMENTAL AND SOCIAL BASELINE INFORMATION

3.1 Environmental conditions

The section describes the geomorphological, climate and related features of the physical environment in the project area.

3.1.1. Topography

The project area can be classified as lacustrine plain characterised with peneplain slopes in relatively flat low-lying landscape with hills and minor carps that do not exceed 25%. This reduces the vulnerability to soil erosion. Though erosion hazards are moderate, disturbance of the soils during construction phase can increase soil erosion risks.



PLATE 4: LAKE SHORE EROSION AT LANDING SITES

Show erosion at landing sites. Shore protection will have to be undertaken at the landing site as provided in the ESMP.

3.1.2 Soils

The soils in the catchment site are predominantly humic gleysols and vertic gleysols that are poorly drained, deep to very deep, dark grey in colour with mottled clay and a thick acid humic top soil. The parent material are basic igneous rocks (basalts and nepheline phonolites). These type of soils expand during wet season and crack during dry season. Away from the specific

proposed construction site, silty clay loams very deep soils are common. As such soils favour water retention, deep excavation during construction and special consideration in construction of sanitary facilities as to mitigate faecal contamination of ground water will be required. Due to their poor workability during rainy season, the proponent will endeavour to undertake the construction during relatively dry season so as to mitigate on energy losses and emissions. Plate 5 provides evidence on some of the soils in the area.

3.1.3 Hydrology

The project site can be described as lacustrine plain with slopes of 0-5% in most places and on the shores of Lake Victoria, which drains from the Yala and Nzoia river systems. The site also has good potential of ground water attributed to Nyanzian rock aquifer system; however such potential in the sub project site is constrained by cost factors.

3.1.4 Climate

The county has an inland equatorial climate modified by the effects of altitude, relief and the influence of the large body of water of Lake Victoria, the subproject is located in an agroecological zone classified as marginal cotton zone (LM-4), an indicator of severe water constraints that severally constraints agricultural production. The area has two rain seasons. The long rains are in March to May and short rains in September to December. The mean annual rainfall, subproject area over the last 30 years is 890-1020 mm with 60% of the annual total falling during long rains. The first season rainfall lies between 350-480 mm while the second season is 125-135 mm in most years unreliable. Temperatures ranges from mean minimum temperature of 16.3° C and mean maximum temperature of 29.1° C. Humidity is relatively high with mean evaporation being between 1,800mm to 2,200mm. The relative humidity ranges between 73 % the morning and 52 % in the afternoon. According to the Kenya Soil Survey and Integrated Regional Development plan for the Lake Basin Development Authority, the sub Project area can be categorized as semi-humid, semi-dry Lower Midland zones (LM4). The climate can support suitable crop varieties such as maize, beans, sorghum and cassava and sweet potatoes (Table 2), with butternut and other horticultural crops such as tomatoes having great potential under irrigation. Leucaena, an agroforestry species has the potential in soil fertility management and resilience building for livestock farmers. Column 4 of Table 1 shows the agro-ecological zone of the sub Project area

Zone	Division /Subcounty	Suitable crops
LM1	Gem, Yala	Sugarcane, Maize, beans, finger millet, coffee, sweet potatoes and horticulture
LM2	Alego Usonga	Maize, beans, tobacco, finger millet, coffee, sweet potatoes and rain fed rice
LM3	Sakwa and Asembo	Maize, beans, finger millet, sorghum, cotton
LM4	Uyoma in Rarieda Sub- County and Yimbo in Bondo Sub-County	Sunflower, sorghum, cassava
LM5	Uyoma in Rarieda Sub- County and Yimbo in Bondo Sub-County	Sorghum, Millet

TABLE 2 AGRO	-ECOLOGICAL	ZONATION O	F SIAYA	COUNTY

Source: Jaetzold et al. 2007

3.2. Socio-Economic Characteristics

The Project area is inhabited mainly by the Luo, and has a rapid population growth that is putting pressure on environmental resources especially land, water, wildlife and forests. This has led to deforestation, cultivation in ecologically fragile areas like steep slopes, river banks and swamps, as well as sub-division of land into small uneconomical units of below 3Ha per household. High poverty level (62 %, poverty index) is one of the greatest challenges facing the proposed project area and the major underlying driver of environmental degradation. Environmental degradation, further fuels incidences of poverty and limits the capacity of the people to escape from the ravages of poverty.

The main socioeconomic activities include; subsistence crop farming, livestock rearing and artisanal fishing from Lake Victoria that is the major livelihood activity and source of income to many of the residents. The average land holding in the area is 2.0 Ha and 30% of the residents are wage earners against 15 % for the county. Women account for a larger proportion of fish mongers compared to men. They are involved in selling, drying and even frying of fish as a value addition enterprise.

Unemployment is relatively high and stands at 38% with the unemployed relying on earning from petty trade, farming, artisan gold mining and remittances from relatives. The youth make 25 % of the population with active labour being about 45% of the population of the area. The area is also made of people who are above 65 years (6%) than the county on average (5%). Fisheries and agriculture are the mainstay economic activity employing about 70 % of the population in the area. Table (2) provides welfare indicators in the area.

Parameter	2009	2018	*
Size (Sq.Km)	92.1	92.1	92.1
Density	302(388)	352	377
Ratio of males: Females	47: 53		
Number of Persons	27,819	32418	34699
Pit latrines		78(82)%	
HIV prevalence	35(38)%	20(24.8)%	
Average Farm size		3.0Ha	
Average Distance to water point		0.6Km	
Access to Water adequate clean		62% (65%)	
water			
Unemployment		35(40)%	
Water point per SQ.KM		I water point per	
		3.5 KM2	

TABLE 3: POPULATION AND KEY INDICATORS WELFARE IN PROPOSED PROJECT AREA UYOMA

Source: (Government of Kenya, 2018)

3.2.1 Land Use patterns

An agro-ecological zone describes agronomic conditions on basis of landform, soil types, rainfall, temperature and water availability which in turn influences the type vegetation, length of growing season and crop adaptability [FAO, 1996]. The ward spreads across agro-ecological LM4. The major land uses in the county are conservation, settlements, artisanal fisheries, and small scale agriculture. The Subproject area is predominantly rural with minor agro processing and small-scale fishing. Increasing ecological destruction and changes in land use are exerting tremendous pressure on ecosystem, placing millions of livelihoods within the sub project area at risk. However the carrying capacity of the land is on average less than the county mean due to the semi- arid climate. Irrigation has great potential in increasing agricultural productivity in the area.



PLATE 5: HOUSING SETTLEMENT AND SANITATION CHALLENGES NEAR KAMARIGA BEACH

. The bird scavengers are attracted to the site by open drying. This can be resolved by the processing facility

3.2.2. Livestock Production

Livestock-keeping is also an important sector among the people residing in the sub project area. Meat, milk and eggs provided by livestock serve as important sources of high-quality protein to complement diets that are based on starchy crops like maize, bananas, millet and cassava. Cattle are important in a few homes for traction and manure. The main type of cattle kept by farmers is of the Zebu type. The number of cattle households on average in the project area is 7 head of cattle and 15 chickens. The production level for milk remains low at 21itres per cow per day.

3.2.3 Wildlife

The varieties of wildlife found in the County include hippopotamus (Lake Victoria, Rivers Nzoia and Yala), crocodiles (Yala Swamp, parts of the Lake Victoria), Sitatunga (Yala Swamp) and monkeys and leopards. The County has several species of fish, but the most popular ones are Nile perch, Omena (Rastrineo bolaargentea), Fulu (Hatlo chromines) and Nile Tilapia. Nile perch is mainly exported to Nairobi and abroad. Other wildlife includes bush pig mainly in

Yala Swamp, Hyenas in Got Abiero, snakes e.g. pythons, cobras and various species of birds. There is a proposal for Lake Kanyaboli National Reserve. However the area is only home to birds and snakes whose status is not under threat. The flora in plate (9) is indicative of semi-arid climate.



PLATE 6: FLORA AND FAUNA IN THE PROPOSED PROJECT SITE

3.2.5. Industrial Activities

The sub Project area as is the rest of Siaya County has limited industrial activities mainly in agro processing, quarrying and artisanal mining and fishing. Further, the area does not have well developed service industry such as banking, insurance, transport and hospitality facilities. The industrial potential especially for fish processing, however, is not fully tapped. The implementation of this project will go a long way in tapping this potential with multiplier impacts on livelihoods in the area and Kenyan economy at large.

3.2.6. Tourism

The Project area has few if any tourism facilities. This contribute to creation of employment opportunities and source of market for agricultural produce. The main categories of tourism attraction include:

- ✓ Museum, Achieng'Oneko Mausoleum
- ✓ Lake Victoria Fish Cages in Lake Victoria.
- ✓ Rawalo Hills and Naya Hills
- ✓ Leisure based Tourism: Goye Beach, Madundu Beach and Luanda Kotieno

3.2.7 Education

The literacy levels in the area are more than 85%, which is relatively higher than the national average. The county has many learning institutions among them Jaramogi Oginga Odinga University, a public chartered university and many middle level colleges. About 25 % of the population are children in primary school with 12 % in secondary school. The County has 652 primary schools with a total enrolment of 248,336 pupils of which 124,381 are boys and 123,955 are girls. There are over 237 secondary schools with an enrolment of 78,468 students.

The enrolment consists of 40,463 boys and 38,005 girls. The West Uyoma ward has 21 schools, 19 of which primary schools and 2 are secondary schools.

3.2.8 Religious beliefs and cultural practices

The people living here are Christians affiliated to Protestants, Catholics and indigenous churches whose spread and distance varies. Strong attachment to African traditional religious practices, though declining is adhered to with tradition of widow Inheritance being common and driver of high HIV infection. Family leadership is patrilineal though women have some decision-making power that allows them to effectively participate in local political, economic and social discourses. However, land use and inheritance are largely controlled by men.

3.2.9. Housing and house types

The major housing type in the area is mud walled iron roofed houses. Table 3 provides a comparison of housing type in the area with the county aggregated statistics. As housing type reflect wealth status, the Project area could be said to be relatively rich compared to other areas of the county.

Housing type	% of Households in the Project site	County indicators
Earth floor	83	90
Cement Floor	17	12
Mud walled	85	82
Corrugated Iron Roof	78	61
Grass thatched	22	25
Brick /Stone wall	15	5%

TABLE 4: HOUSING TYPE IN THE PROJECT AREA

Source: ESIA team field data analysis, 2020

3.2. 10. Vulnerable and Marginalized Groups (VMGs)

The KCSAP PAD recognizes VMGs as the unemployed youth, elderly women and men, widows and orphans and people living with HIV/AIDS. Orphanhood in Siaya County is estimated at 15 per cent (KNBS, 2018). Percentage distribution of orphans between the ages of 0-14 years is about 12 as per the same survey. According to the KIHBS survey old age dependency ratio in Siaya is 13% of the population (KNBS, 2018). Vulnerability is driven largely by the HIV/Aids related mortalities. About 30% of the subproject residents are wage earners against 15% for the county. The higher than average proportion of wage earners in the sub Project area could be attributed to fisheries along the lake.

3.2.11. Land tenure

The land for the proposed Kamariga omega drying facility is private registered land under Kamariga BMU in Uyoma West of Siaya (appendix V) gives a copy of the land ownership certificate).

3.2.12. Communication

The means of communication in the subproject area are as listed in Table 4. Access to the means of communication is relatively high especially the mass media. This could be advantageous in advocacy initiatives on issues such as GBV. The mobile network coverage in the area is relatively high at 83 % compared to an average of 70% of the county but lower than the national connectivity rate of 85%. Network signal is good with all networks well represented.

TABLE 5 : ACCESS OWNERSHIP AND COMMUNICA	TION
--	------

Communication channel	% of Households with access/ownership
Radio	78 (75)
Tv	15 (13)
Mobile Phone	83 (90)
Computer	<1 %

Source: Field data collection, 2020

3.2.13. HIV/AIDS prevalence, Knowledge, Attitudes and Practices

The HIV prevalence in Siaya is above the national level, at 24.8 % and among the highest prevalence in Kenya. However, this has been going down from a high of 38% in 2010 and 27 % in 2015 (Government of Kenya, 2018). Drivers of HIV in the county can be attributed to several social, economic and cultural factors related to polygamy, circumcision, wife inheritance and poverty. The employment opportunities and influx of job seekers into the project site during construction of the proposed Kamariga Omena drying facility may escalate the spread of HIV/Aids in absence of Mitigation measures as proposed in the ESMP, such as sensitization of the workers and involved in the construction, promotion of the use of protective sex and voluntary testing.

3.2.14. Energy sources and their accessibility

Among the sources of energy and lighting in the subproject area are fuelwood (3%), tin lamp (10%), paraffin lantern (10%), solar (3%) and battery lamp/torch about 12%. About 92%. Of the main source of energy for cooking is fuelwood (firewood and charcoal) and The use of solar in the pumping of water may spur its adoption at household level and indirectly contribute to green growth development and climate mitigation. Clearly there is need for household energy source interventions with agroforestry providing an opportunity for meeting household fuel wood needs while providing for other co-benefits such as soil fertility improvement and carbon sequestration. This intervention should go hand in hand with technologies that mitigate indoor pollution risks from fuel wood. Table 5 provides the energy sources and their accessibility at household level in the project area.

Energy source Type	%Households
Paraffin	75 (70)
Electricity	18 (24)
Gas Lamps	10(6)
Fuel wood	92(84)
Charcoal	13 (15)

 TABLE 6 : ENERGY AND THEIR SOURCES IN SUBPROJECT SITE, CENTRAL SAKWA

Source: Field data analysis, 2020; Parenthesis refer to county aggregated levels and basis of comparison

3.2.15. Water Access and Sanitation

Streams, wells, boreholes, roof catchment, rivers, Lake Victoria, water holes, dams, ground catchments and piped supplies are the main sources of water for communities in Siaya. However, the Project area's groundwater potential is highly limited by lithology. The distribution of water sources most of which is surface water depends on the season and also weather patterns. Though the long-term objective of the Government is to enable household's access water within 500m of their settlement, the intervention measures the Ministry of water has put in place so far in terms of piped schemes, point water sources like boreholes, shallow

wells have not been met in the area. The average distance to a water point in the project area is 0.8 Km and it takes about 50 minutes which above the county average of 0.5 Km and time of 0.5 hours respectively. About 78% of the residents have pit latrines but there is no sewerage system in the area. This could imply risk of surface water contamination with faecal material.

3.3 Conflict and Grievance Resolution Mechanism (GRM)

The main grievances experienced with the area involve land succession and inheritance, natural resources, grabbing of public utility spaces and land boundary disputes, tenancy and labour. Domestic violence relating to sexual exploitation and abuse and gender-based violence are also common occurrences. The methods used to resolve such conflicts and which can be drawn upon by the BMU in the area include.

- ✓ Extended family members
- ✓ Religious institutions/ religious leaders
- ✓ Chief/Assistant chief
- ✓ Elders
- ✓ Courts

CHAPTER FOUR

4.0 ANALYSIS OF PROPOSED PROJECT ALTERNATIVES

During the course of formulating the proposed project, several project alternatives were considered to ensure that the best option of project development was adopted. This included a no project alternative which meant that the current status quo remains where debilitated infrastructure such as open air drying on the ground, inadequate stores, posing contamination and possible disease outbreaks. No project alternative would imply, the funds are not utilized and the beneficiaries would continue to be exposed to increased post-harvest losses and wastage due to poor drying conditions. Chances of contamination of *Omena* fish would be high due to inadequate preservation measures. The monetary losses incurred by the fisher-folks would continue to increase, making the beneficiaries vulnerable to food insecurity

4.1 Site alternatives

Two alternative sites were considered; the current site and alternative site around 500m from the current site. Though the current Kamariga Beach landing site for products is easily accessible thus reducing emissions in transport to alternative site, it is located in riparian land. It is also vulnerable to flooding, which has become frequent due to climate change. Construction at this site will also increase effluent load into the lake. The alternative site is vacant plot under Kamariga BMU. This site requires connection to the power line 150 m away. The site has ample space for future expansion if need be. The construction activity will not interfere with the water flow or any ecosystem in the neighbourhood in any way. However, the proponent will compensate for loss of vegetation through landscaping and community tree planting.

4.2. No project scenario

The no construction alternative would imply that the current post harvest practices with attendant food safety risks and high cumulative odourous emissions continue to undermine health and income for the fisherfolk. Such a decision will imply high risk of public health risks such as salmonella, low income and increased vulnerability to correlated shocks that amplify poverty in the community. The funds committed for the project implementation will not be utilized missing on the multiplier impacts in the economy. From the analysed project alternatives, it is evident that a drier will greatly empower the fisherfolk, an optimal course of action and justifiable from social, environmental and economic point of view.

4.3. Analysis of Alternative construction Materials and Technology

Technology choice has been dictated by power saving needs, scale, health and safety as well as the geomorphology. The proposed construction of omena drying facility will be done using modern, locally and internationally accepted materials to achieve public health, safety, security and environmental as well as aesthetic requirements. Equipment that saves energy will be given first priority without compromising on cost or availability factors. The auxiliary structures will be made using locally available source where appropriate. Cement, sand, ballast, metal bars and fittings that meet the Kenya Bureau of Standard requirements.

Alternative technologies available include the conventional brick and mortar style. Due to cost and durability, the brick and mortar style is most popular more so in Kenya. The scale and extent of the project is determined by design, the area allocated and funds available. Other various technologies available include: Hardened quality plastic pipes, timer construction, prefabricated space frame construction, steel frame and aluminum frame. The technology to be adopted will be the most economical and one sensitive to the environment. Heavy use of wood
during construction is discouraged because of destruction of forests. The exotic species would be preferred to indigenous species in the construction where need arises.

4.4. Scale and demand alternatives

The size of the drier will be dictated by current and anticipated landed Omena volumes (10 tons per day) in the medium term including trends in fishing dynamics which suggest that such volumes are unlikely to increase and necessitate the expansion of the intended capacity and/or cause operations at excess capacity, both of which are uneconomical. The proposed project is expected to greatly contribute in the enhancement of product quality and income to the community especially the fisher folk in the short, medium as well as the long term.

CHAPTER FIVE 5.0. POLICY, LEGAL AND ADMINISTRATIVE FRAMEWORK

The section analyses the institutional and governance framework and how it relates to the Project implementation. This is set out in three main subsections; local, World Bank safeguards and international framework.

5.1 General overview

Environment and Social Impact Assessment (ESIA) has received international recognition as vital tool for establishing, appraising and measuring the merits and demerits of proposed and ongoing projects, plans, policies and programs the world over (Lohani 1986). These techniques form key tools and instruments for ensuring sustainability of the projects when utilized during planning and management of the project activities and operations during decision-making. It forms a major component towards enhancing conscientious environmental management and conservation (World Bank, 1993; World Bank, 1999; UNEP, 1998). It is key component in project implementation, routine monitoring, review and evaluation to ensure sustainability and commitment to sustainable development in the project cycle/life-span (IFC, 1998).

Kenya has a number of strategies, governance frameworks, laws and guidelines on climate action such as, the Climate Change Response Strategy (G.o.K, 2010); Climate Smart Strategy (G.o.K, 2017); Agricultural Sector Transformation Strategy (G.o.K, 2019) and Green Economy Strategy under the vision 2030 (G.o.K, 2007). The Green Economy Strategy and Implementation Plan (GESIP), underpins Kenya's commitment to undertake a transition to a green economy (G.o.K, 2007). The Nationally Determined Contributions (NDCs) submitted by Kenya as part of the Paris Agreement. Global Green House Gas Mitigation largely focuses on policies for sustainable agriculture development and climate change action (G.o.K, 2015). The low carbon pathways towards food and nutrition security productivity and resilience in the agricultural sector are reiterated in the national adaptation plan (G.o.K, 2018).

Kenya has established and gazetted various regulatory legislations and provisions that necessitate certain projects and development plans/programs to undergo an ESIA (Tole, 1997). ESIA is recommended for all new projects with the potential to pose environmental impacts/risks at the onset. The Environmental Management and Coordination Act, Cap 387 of 1999 (revised 2015), direct that the proponent of a project undertake an ESIA/EA study and prepare a report thereof for presentation to the National Environmental Management Authority (NEMA). To facilitate this, regulations on ESIA and audits were established under the Kenya Gazette Supplement No. 56 of 13th June 2003 (The Environmental Impact Assessment and Audit Regulations, 2003). Besides, a number of other national policies and legal statutes, including World Bank ESS policies have been reviewed to enhance environmental sustainability in national development projects and conformity with funding requirements. Some of these policy and legal provisions are discussed in more details in the following sections.

5.2. National Policies

The sub project touches on a number of national policies during implementation as well as planning phases. This is discussed under:

5.2.1 Vision 2030

Vision 2030 is divided into three fundamental pillars: Economic, Social and Political pillars. The social pillar aims at realizing a just and cohesive society enjoying equitable social development in a clean and secure environment. In section 22, under Social Strategy, and paragraph 5.4, the strategy envisions Kenya becoming a nation that has a clean, secure and

sustainable. Environment by 2030. Value addition in the agricultural value chain is given prominence.

Relevance

The proposed omena processing sub-project is contributing towards achievement of this strategy as it focuses on value addition in a clean and food secure environment.

5.2.2 Sessional Paper No. 10 of 2012 on Kenya Vision 2030

The session paper covers broad categories of development issues including the waste management, pollution to reduce hazards and adjustment to climate change crises such as droughts. The policy recommends the need for enhanced re-use/recycling of residues including wastewater and increased public awareness raising and appreciation of clean environment as well as the participation of stakeholders in the management of wastes within their localities. Relevance

The proponent shall put in place measures in recycling of wastre residues and increases the level of public awareness on *waste management and water recycling* as need arises

5.2.3. National Policy on Water Resources Management and Development

The National Policy on Water Resources Management and Development (1999) enhances a systematic development of water facilities in all sectors for promotion of the country's socioeconomic progress. It calls for development of appropriate sanitation systems to protect people's health and water resources from institutional pollution. Industrial development projects, therefore, should be accompanied by corresponding waste management systems to handle the wastewater and other waste emanating there from.

Add relevance

The proponent proposed omena drying facility will put in place appropriate sanitation systems as per the Public health regulations. If possible waste management systems should be recyclable in nature so that wastes disposal can be minimised as much as possible

5.2.4 Kenya Youth Development Policy 2019

The policy seeks to provide an opportunity for improving the quality of life for the youth in Kenya through their empowerment and participation in economic and democratic processes as well as in community and civic affairs. It also advocates for creation of a supportive social, cultural, economic and political environment that will empower the youth to be active actors in national development.

Relevance

The proponent will ensure that the policy is implemented by creating opportunities for youth to earn decent and sustainable livelihoods; and to enhance effective civic participation and representation among the youth

✓ The proposed omena drying facility will create an opportunity for building capacity of the youth in green processes, technology and waste management by promoting solar energy and identifying innovations for waste management

5.2.5. National Gender and Development Policy, 2011

The National Gender and Development Policy provide a framework for advancement of gender equity and an approach that would lead to greater efficiency in resource allocation and utilization to ensure empowerment of women. The main objective of the Policy is to enable the mainstreaming of the needs and concerns of men and women in development processes.

✓ This policy will is relevant in ensuring that gender equity is adhered. Women are majority of the Omena Vendors and this project will improve their livelihoods

5.2.6 National Policy for Prevention and Response to Gender Based Violence 2014

The policy seek to ensure; a coordinated approach in addressing GBV and effective programming; enhanced enforcement of laws and policies towards GBV prevention and

response; increase in access to quality and comprehensive support services across sectors; and improve sustainability of GBV prevention and response interventions.

GBV is based on socially ascribed (gender) differences between males and females. Gender can be seen as the allocation of roles, attitudes and values that are deemed by the community to be appropriate for each sex. These roles define power relations between men and women regarding who makes decisions and who owns resources. They are learned and reinforced through interactions in the home and community.

Relevance

The proponent will ensure that there is no aspect of GBV based on the roles of the weak and vulnerable members of the community who will benefit from the project by undertaking the following measures

- ✓ Put in place multi-sectoral Standard operating Procedures for GBV Prevention and response, GBV workplace policies, sexual harassment policies, men to men programmes and male involvement programmes.
- ✓ Put in place standards and mechanisms for debriefing/ counseling service providers dealing with GBV across sectors
- ✓ Protect vulnerable persons by implementing a witness protection programme for GBV victims and survivors.
- ✓ Establish an elaborate communication strategy incorporating all actors including the public, service providers, government agencies and non-state actors so as to effectively respond to GBV.

5.2.7 Beach Management Units (BMU) Regulations, 2017

These regulations created Beach Management Units (BMUs) comprising of stakeholders within fishing communities with mandates of conservation, protection, monitoring and control of fishery resources and the environment, and fisheries planning and development in collaboration with the government. The participation of fisher-folks is in line with the general principles of Code of Conduct for Responsible Fisheries (FAO, 1995). BMUs act as co-management institutions and provide for fisheries management at grassroots level *Relevance*

Relevance

The Kamariga Beach Management Units will take lead during construction, operation of the proposed project.

5.2.8 National Climate Change Action Plan

The National Climate Change Action Plan (2018-2022) identifies priority adaptation and mitigation actions for transforming to a low carbon climate resilient development pathway. The priority adaptation actions for agriculture in the plan include Coordination and mainstreaming of climate change into agricultural extension; establishment and maintenance of climate change related information for agriculture; and upscaling specific adaptation and mitigation actions. *Relevance*

Development of the proposed omena drying facility will ensure that Green House Gas emissions are within allowable limits with minimal consequences to the larger environment. The solar powered drying facility will result in lower emission levels from the drying facility

5.2.9 Kenya climate Smart Agriculture Strategy 2017-2026

The Kenya climate Smart Agriculture Strategy (KCSA) is aimed at adapting climate change, building resilience of agricultural systems while minimizing emissions for enhanced food and nutritional security and improved livelihoods. The strategy objectives are to enhance adaptive capacity and resilience of farmers, pastoralists and fisher-folks to the adverse impacts of climate change, develop mechanisms that minimize greenhouse gas emissions from agricultural production systems and address cross-cutting issues that adversely impact Climate Smart Agriculture

Relevance

✓ The proponent will put in place energy efficient technologies and innovations that will entail reduction of the rate of emissions associated with preservation and transportation along the value chains that either use fuel efficiently or green energy

5.2.10 Agricultural Sector Transformation and Growth Strategy 2019-2029

The importance of agriculture has been emphasized in Kenya through Vision 2030 and the Medium Term Plan III and most recently the President's Big Four priority agenda for 2017-2022, which emphasizes the importance of 100% food and nutrition security for all Kenya. To transform Kenya's agricultural sector and make it a regional powerhouse, the Government has formulated the Agricultural Sector Transformation and Growth Strategy (ASTGS). The Strategy is based on the belief that food security requires a vibrant, commercial and modern agricultural sector that supports Kenya's economic development sustainably and its commitments to regional and global growth

✓ Among the key flagships is support to fisher-folks. The strategy aims at increasing incomes of the small scale fisher folks. This will create an enabling environment for the youth in promotion of Small Medium Enterprises by enhancing value addition and minimizing post-harvest losses

5.3. Legal framework

This study was conducted within an environmental and policy framework as outlined hereunder and in particular, the Environmental Management and Coordination Act, 1999 which has explicit provision for Environmental Impact Assessment (EIA) for projects likely to have adverse environmental impacts. The project has to comply with the provisions of the Environmental Management and Coordination Act, 1999 (Revised 2015) and its subsidiary legislation, as well as sectoral laws that protect the environment from development activities

5.3.1 Constitution of Kenya 2010

The constitution is the supreme law of the land of Kenya. The Constitution requires the public to be consulted and the study has complied by consulting the public. It has also developed measures to mitigate foreseen impacts which will ensure sound management of the environment. Article 42 states "every person has the right to a clean and healthy environment, which includes the right a) to have the environment protected for the benefit of present and future generations through legislative and other measures, particularly those contemplated in Article 69, and b) to have obligations relating to the environment fulfilled under Article 70". There a number of provisions support this investment however key among others is Part 3 article 56. The State shall put in place affirmative action programs designed to ensure that minorities and marginalized groups— (b) are provided special opportunities in economic fields; (c) are provided special opportunities for access to employment; (d) develop their cultural values, languages and practices; and (e) have reasonable access to water, health services and infrastructure.

Relevance

✓ The Constitution is vital in identifying the need for this development initiative, since it endeavours to improve the general wellbeing of the people both environmentally and socio-economically and it will govern the means to ensuring the method in which the proposed omena fish drying project is carried out, by providing an ESIA which is provided in this report.

5.3.2 The Environment Management and Coordination Act, 1999 (Revised 2015)

The Act also deals with pollution prevention and waste management which is relevant to this project, as it will generate different categories of waste during construction and operation. Part II of the Environment Management and Coordination Act, 1999 states that every person in Kenya is entitled to a clean and healthy environment and has the duty to safeguard and enhance the environment. In order to ensure this is achieved, part VI of the same Act directs that any proponent of a new project should undertake EIA/EA study and prepare an appropriate report for submission to the National Environmental Management Authority (NEMA), who in turn may issue a license as appropriate.

Relevance

This is one of the activities that require an ESIA and the project has conducted an ESIA study and is the subject of this report which also conforms to World Bank environmental social safeguards policies and guidelines

- ✓ In carrying out the ESIA study and preparing this report the requirements of this regulations and those of the World Bank Environmental and Social Safeguards have been integrated and followed throughout the process. The proponent did the screening and scoping then as advised by the NEMA office commissioned this ESIA study
- ✓ The proponent shall observe the guidelines as set out in the environmental management plan laid out in the ESIA report as well as the recommendation provided for mitigation of adverse impacts arising from the project activities

5.3.3: The Environment (Impact Assessment and Audit) Regulations, 2003

Environmental Impact Assessment (E SIA) is a critical examination of the effects of a project on the environment. The goal of an ESIA is to ensure that decisions on proposed projects and activities are environmentally sustainable. An ESIA is conducted in order to identify impacts of a project on the environment, predict likely changes on the environment as a result of the development, evaluate the impacts of the various alternatives on the project and propose mitigation measures for the significant negative impacts of the project on the environment. The EMCA, 1999 requires that during the ESIA process a proponent shall in consultation with the Authority seek views of persons who may be affected by the project or activity through posters, newspaper, radio and hold at least three public meetings with the affected parties and communities. The Project proponent pays for the entire ESIA process. The regulations provide for the detailed procedure of carrying the ESIA and audit process in Kenya. They also provide explicitly for public consultation and mechanisms for doing it. The regulations also indicate "Issues to be Considered in Environmental Impact Assessment' in the second schedule of the regulations and "General Guidelines for Carrying out an Environmental Impact Assessment Study" in the third schedule to the regulations. In conducting this exercise, the stakeholders were consulted and their views have been integrated in this report. Implementation of the incorporated ESMP and monitoring of the same is key to actualizing this regulation.

Environmental Audit (EA) is the systematic documentation, periodic and objective evaluation of activities and processes of an ongoing project. The goal of EA is to establish if proponents are complying with environmental requirements and enforcing legislation. The purpose of EA is to determine the extent to which the activities and programs conform to the approved environmental management plan. A comprehensive EA ensures a safe and healthy environment at all stages of project operations and decommissioning. Annual environmental audits will be undertaken during this sub-project operation and maintenance phase.

✓ In carrying out the ESIA study and preparing this report the requirements of this regulations and those of the World Bank Environmental and Social Safeguards have been integrated and followed throughout the process. The proponent did the screening and scoping then as advised by the NEMA office commissioned this ESIA study

✓ The proponent shall observe the guidelines as set out in the environmental management plan laid out in the ESIA report as well as the recommendation provided for mitigation, minimization, and avoidance of adverse impacts arising from the project activities

5.3.4. Environment management and coordination (waste management) regulations 2006 These are described in legal notice No. 121 of the Kenya Gazette supplement No. 69 of September 2006. These regulations apply to all categories of waste as provided in the regulations. These include:

- ✓ Industrial wastes;
- ✓ Hazardous and toxic wastes;
- ✓ Pesticides and toxic substances;
- ✓ Biomedical wastes;
- ✓ Radio-active substances

These regulations outline requirements for handling, storing, transporting, and treatment/ disposal of all wastes categories as provided therein. Wastes contaminated with petroleum products are considered hazardous. Treatment of toxic or hazardous wastes should be done using the classes on incinerators presented in the third schedule of these regulations. These regulations are relevant to handling of the various wastes during construction phase. The ewaste guidelines provide a framework for identification, collection, sorting, recycling and disposing of electrical and electronic waste (e-waste). The project proponent and agents as a responsible citizen have obligation to include these guidelines within her operations for a cleaner and sustainable environment. Through the ESIA, the ESMP has provided measures for managing waste generated through the proposed project.

✓ Measures to be undertaken for proper waste disposal include clearance of non-reusable and recyclable waste and disposing off in designated disposal site.

5.3.5. Environment management and coordination (water Quality) Regulations 2006

These are described in legal notice No. 120 of the Kenya gazette supplement No. 68 of September 2006. These regulations apply to drinking water, water used for agricultural purposes, water used for recreational purposes, water used for fisheries and wildlife and water used for any other purposes. This includes the following:

Protection of water sources for domestic use, industrial use, effluent discharge and Water for agricultural use. These regulations outline:

- \checkmark Quality standards for sources of domestic water;
- ✓ Quality monitoring for sources of domestic water;
- ✓ Standards for effluent discharge into the environment;
- ✓ Monitoring guide for discharge into the environment;
- ✓ Standards for effluent discharge into public sewers;
- ✓ Monitoring for discharge of treated effluent into the environment

Water for the project site will be sourced from the existing service providers. In monitoring, water quality standards would be the basis for future audits. The contractor will adhere to the provisions of the ESMP such as servicing of vehicles and machinery in licensed dealers and use of oil sumps in the service area. The ESMP has outlined the water quality control measures in the ESIA including proper waste disposal measures for oil spills and other wastes, proposed fencing of water pan area, and provision of livestock troughs and communal water drawing point.

- ✓ The ESMP has outlined the water quality control measures such as standards for discharge of effluents into the environment
- ✓ The proponent will implement recommended guidelines on monitoring for discharge of treated effluent into the environment

5.3.6 Environmental Management and Coordination (Conservation of Biodiversity, Access to Genetic Resources and Benefit Sharing) Regulations 2006

The regulations are supposed to ensure Conservation of Biodiversity in the country because, Kenya has a large diversity of ecological zones and habitats including lowland and mountain forests, wooded and open grasslands, semi-arid scrubland, dry woodlands, and inland aquatic, and coastal and marine ecosystems. In addition, a total of 467 lake and wetland habitats are estimated to cover 2.5% of the territory. In order to preserve the country's wildlife, about 8% of Kenya's land area is currently under protection. One requires a NEMA approval if areas of rich biodiversity are going to be affected by a development project. The Conservation of Biodiversity Act Sections 5-9 provides for the protection of endangered species, creation of an inventory, and monitoring of their status, protection of environmentally significant areas, provision of access permits, material transfer agreements and benefit sharing. These regulations layout pollution control measures with a view of avoiding areas of environmental significance and protection of endangered species.

- ✓ This subproject is small scale and in an area without tress therefore the biodiversity is not disrupted and no habitat or any threatened species of flora or fauna is threatened
- ✓ The Proponent and contractor will ensure that great care is exercised in the protection of vegetation during construction.
- ✓ The contractor will abide by C-ESMP which provides for construction of gabions to arrest lake shore erosion

5.3.7. Environmental Management and Coordination (Wetlands, River Banks, Lake Shores and Sea Shore Management) Regulation, 2009.

This Act applies to all wetlands in Kenya whether occurring in private or public land. It contains provisions for the utilization of wetland resources in a sustainable manner compatible with the continued presence of wetlands and their hydrological, ecological, social and economic functions and services. The project is will only capture a fraction of runoff in the catchment.

✓ The Proponent shall comply with the provisions of the Act in protecting wetlands, preventing and controlling pollution and siltation of rivers during construction and operation phase by having an effluent handling plan and license of the same from NEMA

5.3.8. The Fisheries Management and Development Act No.35 of 2016

This is an Act of Parliament that provides for the conservation, management and development of fisheries and other aquatic resources in Kenya to enhance livelihoods of fisheries dependent communities. Relevant sections of this act to the subproject include Part 1 which provides for the establishment of Kenya fisheries services and it's powers in licensing fish processing establishments. Part V of the Act specifies fisheries conservation and management development measures including fisheries impact assessments, pollution control, fishing gear, fisheries management plans and the establishment of Beach Management Units (BMU).Part VI specifies fish production and marketing while Part VII provides for fish safety and quality standards that meets applicable food quality , health and sanitation standards. Part X species the licensing procedures and conditions in fishing and the establishment of fish processing establishments. The Act also provides for structured community participation (Beach Management Units) in fisheries management and setting of standards while and inclusion of Vulnerable and marginalised groups (VMGs) such as women.

✓ The proponent, Kamariga BMU is a registered entity under this act and is in compliance of specifications on fish landing, limitation of the gear type including mesh size used by its members, observation of closed seasons, fisheries management plan and reporting. It is also compliant on participation of VMGs through inclusion of women and people living with disabilities in its management

- ✓ The Processing plant will observe HACCP food system as precautionary measure to management and development of the processing unit, fish quality, sanitation and health standards
- ✓ The ESMP from the ESIA has incorporated specific measures on lake shore protection and ecosystem approach through catchment conservation measures, IPMP and sanitation
- ✓ The processing facility will improve livelihoods of the community in many ways i.e. employment, increased earnings and reduced public health risks thus meeting the vision and mission of the Act.
- ✓ The Proponent has applied for establishment of fish processing unit in compliance with the Act
- ✓ The BMU will come up with and enforce policies on zero tolerance for sexual exploitation in its operation while ensuring women and youth fully participate and gain from the project

5.3.9. The Public Health Act (Cap. 242)

The primary purpose of this Act is to secure and maintain public health. Some of its provisions relevant to this project include prohibition of nuisance activities such as spillage or noise or other condition deemed to be injurious or dangerous to human health. According to Part IX Section 115 no person will be allowed to cause nuisance or condition liable to be injurious or dangerous to human health. Section 116 requires local Authorities to take all lawful, necessary and reasonably practicable measures to maintain their jurisdiction clean and sanitary to prevent occurrence of nuisance on conditional liable for injurious or dangerous to human health. *Relevance*

- ✓ Since the subproject will be implemented at a time when the whole World and the country is experiencing Covid-19 pandemic. Necessary arrangements and preventive measures will be made to prevent contractor, workers, county government team and other players from being infected with the virus guided by the Public Health Act (Covid-19 Restrictions of movement of persons and related measures)
- ✓ Other Mitigation measures include compliance on social distancing and use of sanitizers during all phases of the project
- ✓ HIV/AIDS has been captured as a risk in ESMP and the contractor required incorporate HIV awareness initiatives during construction of the Project

5.3.10. The county Government Act No. 17 of 2012

The Act replaces the Local Government Act (Cap. 265) and gives effect to chapter 11 of the constitution, spells County government powers, functions and responsibilities and range of services under the purview of County governments. The relevant sections of the Act which impact on the project are provisions which touch on:

- ✓ County planning
- ✓ Public communication and information access for effective citizen participation
- ✓ The Integration of minorities and marginalized groups in development planning (VMGs)
- ✓ Dispute resolution concerning planning processes in particular the CIDP, sectoral and spatial plans

In particular Section 115 of the act provides for Strategic Environmental Assessments and Environmental Impact Assessments (EIAs) in Project planning as decision making tools. Among the services devolved to county governments include waste management such as sewerage systems and solid waste handling infrastructure. County authorities thus have responsibilities to establish and maintain sanitary services for the removal and destruction of, or otherwise deal with all kinds of refuse and effluent and where such service is established, compel its use by persons to whom the service is available. To effect the Act, County governments are empowered to make by-laws in respect of all such matters as are necessary or desirable for the maintenance of health, safety and wellbeing of the inhabitants of its area as provided under the Act.

- ✓ Undertaking this ESIA process comply with the outlined principle of citizen participation and aligns well with the CIDP, sectoral and spatial plans as the current site evidenced by registration of the BMU
- ✓ The Grievance Resolution mechanism has been ensured through awareness and sensitization meetings with the county department for public participation taking the lead
- ✓ The ESIA process captures Conflict resolution GRM protocols which is in line with county government Act on public participation and conflict resolution as well resonates and complements county government service provision obligations

5.3.12. The Penal Code (Cap. 63)

Section 191 of the Penal Code states that, any person or institution that voluntarily corrupts or spoils water from public springs or reservoirs', rendering them less fit for its use is guilty of an offence. Section 192 of the same act says a person who makes or vitiates the atmosphere in any place to make it noxious to health of persons/institution in dwellings or business premises in the neighbourhood or those passing along public way, commit an offence.

- ✓ The proponent has taken all the necessary measures to ensure that the provisions of this Act are complied with through this ESIA/ESMP
- ✓ The proponent has taken precautionary steps towards preventing or reducing adverse effects that might arise from the activities
- ✓ The ESS safeguards codifies and complements mitigation measures on offences such as GBV, child abuse, SEA, OSHA that are to be accepted and enforced by the contractor during construction and Kamariga BMU during all Project phases

5.3.13. Occupational Health and Safety Act 2007

This legislation provides for protection of workers during construction and operation phases. It is tailored at implementation of the EHS plan in compliance with the relevant sections of this Act. The Act is relevant both during construction and operation phases of the project due to the fact that the project will involve workers at all stages. Various health hazards are likely to emanate from the proposed project's activities such as workplace accidents. Health issues will therefore be integrated into the project to ensure safety of workers. The relevant subsections relevant to the project are 13, 14, 17, 18, 21a and b, 22c and d, 25 and 51.

- ✓ The work site will be registered with Directorate of Occupational Safety and Health (DOSH) and the contractor will be required to ensure all necessary records on workers are kept during construction phase by providing PPEs, registration of workers, train workers on emergency preparedness and response while ensuring all SOPs on Covid 19 containment are adhered to
- ✓ Occupational health and safety audits will be carried out periodically to ensure compliance with this Act particularly in the construction and operation phase

5.3.14. The Land Planning Act (cap 303)

This Act is the overall planning law for land for both agricultural and constructed environments. Under this Act, all developments or changes to land use must be approved by a planning authority. Section 9 of the subsidiary legislation (the development and use of land legislation 1961) stipulates that before any plans are submitted to the Minister for approval steps should be taken as may be necessary to acquaint the owners of any land affected such plans particulars of comments and objects made by land owners should also be submitted. *Relevance*

✓ The Act will reduce conflicts among other land users in the area, such as residential and commercial land uses. This is because affected land plans shall be submitted to the Minister for approval and the steps taken will be acquainted to the owners of lands affected hence the land owners will be involved in the process

5.3.15. Physical Planning Act, 1999

It provides for the preparation of a physical development plan for the purpose of improving the land and providing for the proper physical development of such land, and securing suitable provision for transportation, public purposes, utilities and services, commercial, industrial, residential and recreational areas, including parks, open spaces and reserves and also the making of suitable provision for the use of land for building or any other purposes. Section 29 of the Act empowers local authorities to control all development activities so as to ensure conformity to approved planning standards. Section 30 states that any person who carry out development without permission will be required to restore the land to its original conditions. *Relevance*

- ✓ The Act also provides an environmental impact assessment for a project which is likely to have injurious impact on the environment, and this ESIA meets that requirement. Such an EIA must be approved by the National Environment Management Authority (NEMA).
- ✓ Undertaking this ESIA project report was sanctioned by Kamariga Beach Management Unit is in tandem with the provisions and is in line with sectoral and spatial planning and Siaya County CIDP.

5.3.16. The Wildlife Conservation and Management Act, Cap 376

The Act provides for the protection, conservation and management of wildlife in Kenya. The provisions of this Act should be applied in the management of the project especially where it passes through protected wildlife habitats, migratory areas or dispersal corridors. The Act establishes the Kenya Wildlife Service (KWS) and provides for the establishment of national parks and national reserves and defines how they are to be managed. As per the Act, the overall mandate of KWS is to conserve and manage wildlife in Kenya. Its key responsibilities are:

- ✓ Sole jurisdiction over National Parks; supervisory role in the management of National Reserves, Local and Private Sanctuaries;
- ✓ License, control and supervision of all wildlife conservation and management activities outside the protected areas; conservation education and training; and Wildlife Research. *Relevance*

The project area is likely to be within the migratory area or disposal corridors. The migratory areas may be having migratory birds from parts of the world and hence there is need to sensitize communities on endangered migratory birds and how to accommodate them

5.3.17. The Land Registration Act, 2012

The Land Registration Act is place to revise, consolidate and rationalize the registration of titles to land, to give effect to the principles and objects of devolved government in land registration, and for connected purposes. This Act applies to Subject to section 4, this Act shall apply to:

- ✓ Registration of interests in all public land as declared by Article 62 of the Constitution
- ✓ Registration of interests in all private land as declared by Article 64 of the Constitution
- ✓ Registration and recording of community interests in land. Section 24 states that the registration of a person as the proprietor of land shall vest in that person the absolute ownership of that land together with all rights and privileges belonging or appurtenant thereto;

The registration of a person as the proprietor of a lease shall vest in that person the leasehold interest described in the lease, together with all implied and expressed rights and privileges

belonging or appurtenant thereto and subject to all implied or expressed agreements, liabilities or incidents of the lease. The Act also provides for the compulsory or otherwise acquisition of land from private ownership for the benefit of the general public. For the acquisition to take place, the minister responsible must issue a gazette notice. The Act also provides for full compensation to the affected parties. The sub-project is being undertaken on registered land under the BMU.

Relevance

✓ The land is registered as private land under Kamariga BMU and there are no encumbrances on the right to access, use and control as specified and guaranteed under the act

5.3.18. Labour Laws of Kenya including Employment Act 2007

This is the revised employment act in Kenya, repealing the former Employment Act Cap 226. It deals with new employment conditions of employment and the rights of workers. All workers, including those employed during the construction phase, will be employed under this Act which includes provision with respect to minimum wage, working conditions and time, and also in the resolution of disputes. This provision will guide the contractor in engagement and payment of the workers during implementation. The contractor will be given a copy of this report (C-ESMP) for reference too.

Relevance

- ✓ The Proponent through the contractor will make sure that fairness and gender equity are followed during the recruitment of the labour force to be used during the construction phase. Preference will be given to the local community for both skilled and unskilled labour.
- ✓ The proponent will ensure the engaged contractor is qualified, register the site as place as provided under DOSH. The contractor and proponent will also provide requisite PPEs to the workers and the latter will always be required to wear these.

5.3.19. The Sexual Offences Act, 2006

This Act protects people and employees from any unwanted sexual attention or advances by staff members. This act ensures the safety of women, children and men from any sexual offences which include: rape, defilement, indecent acts. This law will govern the code of conduct of the Contractor's staff and provide repercussions of any wrong doing *Relevance*.

✓ The ESMP provides for the implementation of a SGBV action plan with an Accountability and Response Framework as part of the Construction-ESMP (C-ESMP) and administration of the whole project cycle

5.3.20. The Standards Act Cap 496

The Act is meant to promote the standardization of the specification of commodities, and to provide for the standardization of commodities and codes of practice; to establish a Kenya Bureau of Standards, to define its functions and provide for its management and control. Code of practice is interpreted in the Act as a set of rules relating to the methods to be applied or the procedure to be adopted in connection with the construction, installation, testing, sampling, operation or use of any article, apparatus, instrument, device or process. The Act contains various specifications touching on electrical products.

Relevance

✓ The Proponent shall ensure that commodities and codes of practice utilized in the project adhere to the provisions of this Act especially in relation to siting of sanitary facilities, solar system and structural standards in the enforcement of construction standards.

5.3.21. The Energy Act, 2019

The Energy Act, 2019 was enacted in response to calls to consolidate the laws relating to energy; promote renewable energy; promote exploration, recovery, and commercial utilization of geothermal energy; regulate midstream and downstream petroleum and coal activities, among others.

Relevance

✓ The Proponent will incorporate solar energy solutions by installing solar panels to reduce energy costs. This is in line with the promotion of renewable strategies and reduction of emission from non-green energy resources as indicated in the Energy Act and in Climate Change Act

5.3.22. National Construction Authority (2011)

The National Construction Authority Act, Number 41 of 2011 streamlines, overhauls and regulates the construction industry in Kenya. The Act contains provisions on the quality and safety standards of any construction work.

Relevance

- ✓ The Proponent will ensure the engagement of a registered and qualified contractor as required under the water act for the construction of the project
- ✓ The proponent will ensure that regular site meetings by all the relevant stakeholders are held to oversight the work
- ✓ The work site will be registered with Directorate of Occupational Safety and Health (DOSH)

5.4. World Bank Environmental and Social Safeguards

The following World Bank environmental safeguards (Operational Policy (OP) /Bank Procedure (BP)) will guide the proposed sub-project.

5.4.1 OP/BP 4.01 (Environmental Assessment)

The World Bank considers environmental impact assessment (ESIA) as one among a range of instruments for environmental assessment. These are procedures that ensure the proposed development is sustainable and environmentally sound. Although its operational policies and requirements vary in certain respects, the World Bank follows a relatively standard procedure for the preparation and approval of an environmental assessment study, which:

- ✓ Identifies and assesses potential risks and benefits based on proposed activities, relevant site features, consideration of natural/human environment, social and transboundary issues
- ✓ Compares environmental pros and cons of feasible alternatives
- Recommends measures to eliminate, offset, or reduce adverse environmental impacts to acceptable levels (sitting, design, technology offsets)
- ✓ Proposes monitoring indicators to implement mitigation measures
- ✓ Describes an institutional framework for environmental management and proposes relevant capacity building needs

The objective of the World Bank's environmental and social safeguard policies is to prevent and mitigate undue harm to people and their environment in the development process. These policies provide guidelines for the bank and borrowers in the identification, preparation, and implementation of programs and projects. Safeguard policies have often provided a platform for the participation of stakeholders in project design, and have been an important instrument for building ownership among local populations. The World Bank's environmental assessment policy and recommended processes are described in Operational Policy (OP)/Bank Procedure (BP) 4.01: Environmental Assessment. Its purpose is to improve decision making, to ensure that project options under consideration are sound and sustainable, and that potentially affected people have been properly consulted. ✓ The guidelines have helped to assess the potential environmental risks and impacts of the sub project in its area of influence; examine sub-project alternatives; identify ways of improving the sub-project selection, siting, planning, design, and implementation by preventing, minimizing, mitigating, or compensating for adverse environmental impacts and enhancing positive impacts. The assessment has taken into account: the natural environment (air, water, and land); human health and safety) social aspects (involuntary resettlement, indigenous peoples and physical cultural resources). The project has consulted the public on the development of the project and the ESIA as required under this Policy.

Environmental Assessment is one of the 10 environmental, social, and legal Safeguard Policies of the World Bank. Other safeguard policies of relevance to this study include:

- ✓ Bank Safeguard Policy 4.04 Natural Habitats;
- ✓ OP/BP 4.01 Environmental Assessment
- ✓ OP/BP 4.09 Pests Management
- ✓ OP/BP 4.11 Physical Cultural Resources

5.4.2. OP/BP 4.04 Policy on Natural Habitats

The policy is designed to promote environmentally sustainable development by supporting the protection, conservation, maintenance and rehabilitation of natural habitats and their functions. The policy seeks to ensure that World Bank-supported infrastructure and other development sub-projects take into account the conservation of biodiversity, as well as the numerous environmental services and products, which natural habitats provide to human society. The policy strictly limits the circumstances under which any Bank-supported sub-project can damage natural habitats (land and water area where most of the native plant and animal species are still present).

A careful evaluation of the project context was carried out in the baseline studies to determine if the project setting has any significance that may disrupted due to the implementation of the sub project. This safeguard policy requires a precautionary approach to natural resources management and requires the conservation of critical environments during project development. In order to ensure conservation and project sustainability, this policy requires that:

✓ Project alternatives are sought when working in fragile environments. Key stakeholders e.g. WRA were consulted during the project design, implementation, monitoring and evaluation of mitigation.

The proponent through this ESIA and ESMP has taken advance measures for protecting, preserving and conserving the environment in the project setting from predicted and emergent adverse impacts. Specifically the project is not being implemented in a natural habitat or an area of ecological significance

5.4.3. Pest Management (Operational Policy, OP/BP 4.09)

The policy is meant to minimize and manage the environmental and health risks associated with pesticides use and promote and support safe, effective, and environmentally sound pest management. Though the policy as no procurement of pesticides or pesticide application equipment is envisaged for the subproject per see, the envisaged horticultural and agricultural improvement activities tied to the wider project implementation objectives may involve pesticide use and subsequent increase in health and environmental risk.

✓ The use of pesticides and agrochemicals will be on very small-scale level and an integrated pest management plan will not be required.

5.4.4 OP/BP 4.11 (Physical Cultural Resources)

The objective of this policy is to avoid or mitigate adverse impacts on physical cultural resources from development sub-Project, Identify the likely physical cultural resources issues, if any, to be taken into account by the EA or If the sub-project is likely to have adverse impacts

on physical cultural resources, identify appropriate measures for avoiding or mitigating these impacts as part of the EA process. These measures may range from full site protection to selective mitigation, including salvage and documentation, in cases where a portion or all of the physical cultural resources may be lost. The proponent during the environmental and social screening exercise investigated from the community of the possibility of a history of any physical or cultural significance of the proposed project site. It was found out from the community that the land for the sub project site has not and is not known to have any physical or cultural object/resource that the proposed development may interfere with. This ESIA report has established that (OP/BP 4.11 -Physical Cultural Resources) will not be triggered through the implementation of the proposed sub project. This act resonates well with OP 4.11 on the protection of physical cultural resource under World Bank safeguards operation policy. International best policy for accidental discovery of heritage resources and burial sites will be adhered to through well-established documentation and line of communication protocols, securing the site and ceasing operations where chance finds occur as to avoid any further damage in case of such chance finds (see Appendix XII on Chance find procedures)

5.5. Relevant International Conventions and Treaties

Kenya has ratified various international conventions on environment that are applicable to this study. The Ministry of Foreign Affairs deals with international treaties at the primary stages of negotiation. The ministry offers advisory guide to the government on the need to ratify such a treaty if considered to be of national interest. Implementation portfolio then moves to the line ministry, relevant departments and co-operating agencies. If some international issues arise, various international agreements listed above or that exist will be applied for this project. Conventions are agreements that are legally binding on states that have become parties to them Kenya is signatory to several international conventions and treaties that would need to be adhered to in implementing this project and are geared towards environmental protection and conservation. Some of these include;

- ✓ ILO Conventions ratified by Government of Kenya
- ✓ Safety and Health in Construction Recommendation, 1988
- ✓ Convention on Wetlands or the Ramsar Convention
- ✓ Convention on Biodiversity
- ✓ The Convention on International Trade in Endangered Species (CITES)
- ✓ Convention on the Conservation of Migratory Species
- ✓ United Nations Framework Convention on Climate Change
- ✓ Important Bird Areas

5.5.1. International Convention on Biological Diversity (1992)

The convention promotes the protection of ecosystems and natural habitats, respects the traditional lifestyles of indigenous communities, and promotes the sustainable use of resources. **5.5.2. Ramsar Convention 1971**

The convection governs wetlands of international importance. The convention was entered into force in Kenya in 1990 and it governs Lake Nakuru, Lake Baringo, and Lake Natron, which is a shared ecosystem between Kenya and Tanzania. Kenya is therefore committed to avoid degradation of wetlands under its jurisdiction. Kenya has also ratified the Agreement of the Conservation of Eurasian Migratory Water Birds (2001) and the African Convention on the Conservation of Nature and Natural Resources (1968), the Convention on International Trade in Endangered Species of Wildlife Fauna and Flora (CITES) 1973 which prohibits trade in species such as Dugongs and also in Ivory.

Relevance

✓ The proponent will need to ensure that these important conventions are not violated during construction, operation or decommissioning of the proposed projects.

5.5.3. The United Nations Framework Convention on Climate Change (UNFCCC)

This is an international environmental treaty produced at the United Nations Conference on Environment and Development (UNCED), informally known as the Earth Summit, held in Rio de Janeiro from 3rd to 14th June, 1992. The objective of the treaty is to stabilize greenhouse gas concentrations in the atmosphere at a level that would prevent dangerous anthropogenic interference with the climate system. The treaty itself sets no mandatory limits on greenhouse gas emissions for individual countries and contains no enforcement mechanisms. In that sense, the treaty is considered legally non-binding. Instead, the treaty provides for updates (called "protocols") that would set mandatory emission limits. The principal update is the Kyoto Protocol, which has become much better known than the UNFCCC itself.

Relevance

✓ The proponent will adhere to energy and water saving technologies such as recycling to reduce the environmental footprints.

5.6. Institutional framework

The project beneficiary is the Kamariga BMU of West Uyoma ward and is responsible for day to day administration of the project during the operation phase. KCSAP, Siaya has the overall responsibility of ensuring successful project implementation, especially the procurement and mainstreaming of ESS safeguards. The actual construction of the project will be undertaken by a contractor identified through a competitive bidding process. Monitoring the environmental impacts of the project and compliance to the Environmental Management Plan will be the responsibility of the National Environment Management Authority (NEMA) through the County Director of Environment, and the Subcounty Environment Committee. However, the CPCU, ESIA expert, Health and safety officer and the CESCO will undertake routine monitoring to ensure compliance and implementation of the ESMP.

CHAPTER SIX 6.0. PUBLIC PARTICIPARTION AND STAKEHOLDER CONSULTATION

6.1 Overview

A public participation and stakeholders consultations were undertaken in line with the Environment Management and Co-ordination Act 1999, and the County Government Act, 2012 as well as the World Bank ESS guidelines and policies. Plate (8-11 and 12-13 in appendix XII) provides evidence of the ESIA process compliance with the provision.



PLATE 7: A FOCUS GROUP DISCUSSION AT BMU OFFICE, THIS WAS DONE BEFORE COVID 19 OUTBREAK.



PLATE 8: COMMUNITY ENGAGEMENT AT THE PROPOSED PROJECT SITE.

Compliance with the Ministry of Health guidelines on Covid 19 Containment

The objectives of public consultations for this Environmental Audit were to:

- ✓ To seek and examine views on health, safety, social and environmental issues from the potentially affected community;
- ✓ To lay the foundation for future negotiations on any issues that may arise so as to build consensus and reach a mutually acceptable resolution of issues.
- ✓ Provide the establishment's neighbours/community with a forum to air any issues or concerns they may have with the establishment's operations on Health Safety, social and Environment (HSSE)
- ✓ To facilitate the integration of plausible EHS management practices into the Environmental and Social Management Plan(ESMP) as recommended by neighbours/community

6.2. Methodology used in the Consultative Public Participation

The environmental and social assessment public participation exercise was conducted in January 2020 by the expert using 3 approaches (i) Focus group discussions and Key informant interviews , (ii) Field surveys and observational checklists and (iii) Public meetings. In general, the following steps were followed in carrying out the entire CPP process: -

- ✓ Identification of institutions and individuals interested in the process and compiling a database of the interested and affected parties (KII schedule in Appendix IX and Community barazas in Appendix (VI-VIII and XII)
- ✓ Administration of questionnaires to different target groups and local community members in the administrative area for the proposed project site (Appendix VII)
- ✓ Public / Technical Meetings at various levels and with different target groups

6.2.1 Stakeholders Public Meetings

Consultative experts' meetings were continuously held during the field exercise to consolidate the issues affecting the project as well as capturing issues raised by the project affected persons. Two comprehensive public meeting as well as technical meetings were held on the dates of 20 January,2020, and 4th March, 2020 with the local residents, chiefs, Village elders, and other local administrative leaders in attendance in the project area. (Minutes of meeting and list of participants Appendix VI and VII respectively). The Issues arising in the meeting are captured in as reported in the following sections;

6.2.2. Potential Positive environmental and social Impacts

- ✓ Increased employment opportunities
- ✓ Reduced exploitation from middlemen
- \checkmark Conservation of the catchment through tree panting
- ✓ Resilience to weather related price swings
- ✓ Improved food security and nutrition security
- ✓ Reduced work load on women in drying the fish

6.2.3. Potential Negative environmental Impacts

Overall, no adverse community objection to the project implementation made during the consultative process by the community and other stakeholders such as the County Government of Siaya and Lead agencies such as Fisheries department. However, lake show erosion, management of effluents and emissions were perceived to be critical. The design will take into account waste management i.e. the technology in sanitary facilities and effluent management as well as food safety standards.

6.2.4. Social Concerns

Various issues, most which fall under the social category, were raised particularly during site meetings and FGDs (See appendix VIII and IX). The issues are highlighted under and their resolution is captured in Table (6).

- ✓ Sexual exploitation
- \checkmark Licensing of the processing unit as provided for under the fisheries act
- \checkmark Funding of income generating activities to motivate the PMC and the youth

✓ Management hiring of labour to minimize conflicts between the contractor and the community

6.4 Analysis of public consultation and participation

The views of these stakeholders were considered and their names and their contacts were taken for future references as required by NEMA. (See Appendix VII) for the list of participants in the Public Participation and Consultation. The local communities and major stakeholders independently gave their views, opinions, and suggestions in their best interest, bringing out the factors that affected the circumstances, influences, and conditions under which their organizations exist.

From the field work and the public meetings, it was apparent that the majority of the stakeholders were aware of the proposed project and unanimously supported its implementation. The consultant and proponent also responded to the queries that the public sought to know about the project. All the environmental issues raised can be adequately mitigated. Other issues surrounding the project were successfully settled during the public meetings since representatives, of the proponent were in attendance and responded to the issues which were unclear to the public. The comments are captured in appended sample questionnaires (Appendix VII and KII in appendix IX). The community showed unanimous approval for the project. Employment creation, adaptation to and mitigation of extreme climate change episodes, poverty alleviation and environmental conservation were the main reason for the support. From the field work and the public meetings, it was apparent that the majority of the stakeholders were aware of the project. The consultant and proponent also responded to the queries that the public sought to know about the project which is summarized in Table (7).

NB The sections above have not reported on the issues raised by men, women or VMGs who participated in the consultations in regard to which among them will be impacted by the subproject investments. Despite the photos showing their participation and the common knowledge that omena value addition is a female dominated enterprise there is no mention of their concerns in the section.



PLATE 9: COMMUNITY MEMBER GROUP PHOTO AFTER CONSULTATION

TABLE 7: SUMMARY OF MAJOR ISSUES DURING PUBLIC PARTICIPATION

Venue	No. of participants	Major issue/concern	Response
Kamariga BMU grounds	10 Women 18 Men people during FGDs and 48 men and 20women during general baraza. 18Men and 6 Women persons during ESMP feedback	Management of effluent	Need for construction of bathrooms, sceptic tank, construction and good mainaitainace of drainage facilitis
		Management hiring of labour to minimize conflicts between the contractor and the community	Training of PMC and sensitization on GMR mechanism for accountability
		Sexual exploitation of women	BMU to come with policy and code of that stresses zero tolerance to sexual exploitation and abuse among its members and work place
		Women empowerment and gender mainstreaming	The beach Management Unit as the proponent to ensure gender issues are mainstreamed during operation phase i.e Men wwomen and the youth are well considered. The VMG in the community have to be given priority
		Criteria on employment during construction and operation phase Lake shore erosion	The contractor to engage with the community on hiring processes Contractor to build gabions in the fish landing site
		Increased waste generation and post-harvest losses	Acquire services of licensed waste handler to dispose wastes which are likely to increase in high volumes

CHAPTER SEVEN

7.0 ANTICIPATED IMPACTS AND MITIGATION MEASURES

7.1. Introduction

The impacts that are expected to arise from the proposed project have been identified and discussed in regard to all phases of the proposed project cycle; construction, operational and decommissioning. The negative impacts will be minimal and will only last for the time of construction phase, with a few lasting for a long time during the operation phase.

7.2 Impacts during the construction phase

7.2.1 Anticipated positive impacts

7.2.1 Creation of employment opportunities

Income generation opportunities will include; employment during construction phase for both skilled and un-skilled labour. Indirect employment will be created where suppliers of foodstuffs and other goods and products will gain income by supplying their services and products to the construction site.

The community members will provide labour for site clearing, construction works, loading and offloading of construction materials provision of security at the site among others.

Enhancement measures

- The Proponent will as part of the contracting recommend to the contractor(s) to consider the area resident for most of the job opportunities at the site including masons, casual labour etc.
- The Proponent will ensure that the contractor(s) do not discriminate some of the community members or other persons based on gender, skin colour, and relationship with the members of the Kamariga omena drying facility, and offering job opportunities at the site.

7.2.2 Demand for locally available construction materials

It is estimated that part of the project cost will be used in the procurement of materials. This includes building stones/bricks, sand, steel and cement among others. The supply of these materials translates into boosting both the local and national economies. The multiplier' effect of this project also translates into increased revenue to the county and national governments in terms of tax and other service charges.

Enhancement measures

• The Proponent must ensure that as part of the contracting agreements the contractors are advised to procure most of the construction materials from the suppliers within the local community.

7.2.3 Improved infrastructure through development of foot paths

The proposed project will result in the development of drainage structures and foot bath crossings which has been provided in the Bill of Quantities.

7.3 Anticipated Negative Environmental impacts

The anticipated negative impacts during the construction phase include the following:

7.3.1 Hydrological impacts

Earth movement, disposal of vegetation and other cleared materials and the inadequate disposal of liquid and solid waste, including the human waste from the workers, may also cause physical and chemical alteration of surface and ground water quality. Since storm water discharge

constitutes a major portion of the total volume of the receiving water, adverse public health effects are likely in absence of water testing and treatment interventions. This may occur during construction phase as well during operation phase.

Proposed mitigation measures are:

- \checkmark The contractor to install roof harvesting to reduce runoff
- ✓ The contractor and proponent to allow for natural seepage of water by replanting disturbed areas with grass and indigenous trees
- ✓ The proponent to recycle rinse water and wastewater for some specific noncritical applications where feasible as long as hygiene considerations are observed
- ✓ Improve the process lay out to facilitate cleaning and eliminate wet transport of wastes, thereby minimizing water consumption

7.3.2. Landscape disturbance, erosion and Vegetation loss

Surface alterations made as part of the project works could destabilize the soil and lead to soil erosion. Land clearing in the project area and excavation may generate large amounts of fill and rubble, which will need to be transferred to appropriate disposal sites. Compaction is also expected to be limited to the construction area. Increased disturbances of the soil and movement will increase the risk of erosion. Civil works, excavations, or an inadequate planning of cuts and fills, can affect the water table significantly.

Mitigation

- ✓ The proponent will adopt selective de-vegetation that aims at clearing only within the project site where necessary.
- ✓ The proposed Project component on NRM and SLM to support community establishment and planting of indigenous trees in the catchment areas around the storage facility compound
- ✓ Excavations of the site will be confined only within the sections upon which construction is taking place
- ✓ Excavated earth will be held away from drainage channels
- ✓ *The proponent shall undertake landscaping of the proposed facility*

7.3.3 Surface Run-off Management

Surface run-off will mostly be from open surfaces such as parking bays and roof-tops. Some pollutants are expected from such areas. These include: Waste/spilt oils and grease from parkyards and which are undesirable in sewerage systems and the public drainage network. Some waste oils and lubricants may contain hazardous substances such as poly-chlorinated biphenyls (PCBs). Solid wastes including papers, plastics and broken pipes among other refuse may also be thrown in the compound and drains.

Mitigation measures

- ✓ Stone pitch the inlet ways
- ✓ Sensitization and afforestation of the landscape/catchment
- ✓ Adhere to structural standards in water storage

7.3.4. Noise and Vibration Generation

Different construction activities (earth movement, transport of construction equipment, heavy machinery use, mining, etc.) would temporarily increase noise. Continuous exposure to noise levels above 85 dB may cause hearing problems leading to occupational deafness. Noise and vibration produced during construction may have some temporary negative impacts to the immediate residents. These may include impairing verbal communication, temporary hearing problems/temporary threshold shift (TTS), noise annoyance or even interference of the normal

behaviour of domestic. Anticipated source of noise pollution is the excavating machine and during mobilization of the construction materials to the site.

Mitigation measures

- \checkmark People participating in the construction should be provided with Personal Protective Equipment (PPE) such as ear muffles for ear protection
- ✓ Sound-attenuated equipment should be used as much as possible
- ✓ No unnecessary hooting by contractor and resident vehicles
- ✓ Noise levels should be kept within acceptable limits preferably as stipulated within the Environmental Management and Coordination (Noise and Excessive Vibration Pollution) Control Regulations 2009 :
- ✓ Limit pickup trucks and other small equipments to an idling time, observe a common sense approach to vehicle use and encourage workers to shut off vehicle engines whenever possible during construction phase

7.3.5 Air pollution and Aerial Emissions

Different construction activities (earth movement, transport of construction equipment, heavy machinery use, mining, etc.) would temporarily increase dust emissions. The impact of dust is however expected to be minimal.

Increased use of fossil fuels during construction phase of the project is anticipated and the emissions may contain potential pollutants like NOx, Sox, Cox2/CO and other hydrocarbons, depending on the type of fuel used by the vehicles. This may also have an effect on level of greenhouse gas emissions. Excavation and the movement of vehicles carrying material such as fuel and other required construction materials and equipment during construction will result to generation of dust in the air. During operation phase, the main emission will be odours fish products.

Mitigation measures

- \checkmark Ensure that maintenance on all machinery is done regularly to avoid the emission of noxious gases.
- ✓ Drivers and machine operator to avoid unnecessary running of motor vehicle engines and machinery when not in use
- ✓ Harnessing of solar energy during construction and operation phase to mitigate greenhouse gas emission
- \checkmark

7.3.6 Dust generation

Excavation and the movement of vehicles carrying material such as fuel and other required construction materials and equipment during construction will result to generation of dust in the air. This will take a short duration and mitigation measures are easily implemented.

Proposed mitigation measures are

- \checkmark Provision of dusk masks to workers by the contractor
- ✓ Contractor to instruct drivers to avoid high speed near settlements in the project area
- \checkmark Contractor to ensure water and mist sprays are undertaken during dry periods
- ✓ Contractor to ensure trucks removing waste and delivering materials such as cement are covered

7.3.7 Effluents and pollution of water sources

This may occur during construction phase as well during operation phase. Recommended mitigation measures are:

✓ Sweep up solid materials for use as a by-product, instead of washing them down the drain

- ✓ Fit drains with screens and/or traps to prevent solid materials from entering the effluent system.
- ✓ Use dry cleaning techniques where possible, by scraping equipment before cleaning, pre-cleaning with air guns and cleaning floor spills with squeegees
- ✓ the contractor should provide oil sumps at the construction yard and/or service and fuel at registered oil dealers yards necessary measures to prevent oil and grease spills and soil erosion which may pollute the water
- ✓ Install bio-digester for latrine facilities
- ✓ Regular testing and chlorination of water at the community water point
- ✓ Community sensitization and awareness on sanitation

7.3.8. Aerial emissions

Fish processing is associated with emission of foul odours. The following mitigation measures are recommended;

- ✓ Avoid processing batches of raw material that are of considerably lower than average quality; this will reduce the odour components
- Reduce the stock of raw materials, waste, and by-products and store this stock for short periods of time only in closed, well-ventilated place
- ✓ Keep all working and storage areas clean and remove waste products immediately from the production line
- ✓ Cover all transfer systems, wastewater canals, and wastewater treatment facilities to reduce the escape of foul odours
- ✓ Contractor to erect construction signage and secure the construction site to minimize risks of accidents and prevent injuries to the public
- ✓ The contractor and proponent to provide First Aid box and a trained person to handle site emergencies and incidences, while displaying emergency number for ambulances/provision of emergency vehicles

7.3.9 Oil spills/Fuels and Lubricants

Oils and grease spillage on the ground may cause contamination to the soil and groundwater. Proposed mitigation and management measures are:

- ✓ Proper maintenance of vehicles and other equipment (using petroleum products) to avoid fuels and lubricants spills at the project site.
- ✓ The contractor should properly handle, storage, and disposal off oils and greases and their wastes during construction by ensuring that servicing is strictly done at designated servicing yard or external petroleum stations

7.3.10 Waste generation

The project will entail a small scale of masonry work, excavations. This will generate wastes that need to be managed. Most of the waste will be generated during the construction waste. This includes papers used in packaging cement and soil this can pose the risk of the site being a breeding for pests, pollution of the physical environment and attraction for scavengers.

Mitigation measures

- ✓ BMU and contractor to provide solid waste collection facilities and encourage waste segregation through sensitizing workers and community on waste management practices
- ✓ Contractor to practice waste separation to enable easy recycling of re-usable waste materials
- ✓ Contractor to provide temporal waste disposal receptacle on site
- ✓ Contractor to liaise with licensed waste collector to routinely collect and dispose the waste

- ✓ Contractor to dump unused excavated materials and debris in designated places
- ✓ Where feasible, reprocess omena waste into commercial by-products such as chicken feed and organic manure
- ✓ wastes should be recovered and taken to the by-product facility in time to prevent product deterioration
- ✓ Contractor to install drip pans or trays to collect drips and spills

7.3.11 Limitations of access and interruption of services

The construction will neither interrupt traffic on the present road nor limit access to populated areas and workplaces. Construction activities will not interrupt or reroute traffic, and cause interruptions in water, light and telephone service. However, it may interrupt access to water by the community during the construction phase.

 \checkmark Allow access to services and resources like water as need arises

7.4 Anticipated negative Social and health impacts during construction phase

7.4.1. Labour Influx Effects

This impact is triggered during Project Construction Phase due to the Project attracting various categories of workers from local, national and international markets. The construction of the project will attract jobseekers and hawkers with possibility of thieves intruding into the area. This therefore leads to concentration of people in one area drawn from diverse social and cultural backgrounds often resulting to a number of issues as listed below;

- \checkmark Strain on various resources especially water resources
- ✓ Grievances from local community members over job opportunities
- ✓ Sexual Exploitation and Abuse
- ✓ Unwanted Pregnancies

Mitigation measures

- ✓ The contractor awarded the Project will develop a labour Management Plan (LMP) in consultation with local leaders
- ✓ The contractor will ensure effective community engagement and strong grievance mechanisms on matters related to labour, with a discrete mechanism for safely and confidentially reporting issues of SEA and GBV at the community level triggered by the Project
- ✓ Effective contractual obligations for the contractor to adhere to the mitigation of risks against labour influx, the contractor should engage a local community liaison person who is also trained in PSEA
- ✓ The contractor will ensure proper records of labour force on site while avoiding child and forced labour
- ✓ The contractor will ensure compliance to provisions of Workplace Injuries and Benefits Act (WIBA) 2007 by providing insurance cover for workplace injuries and aligning with the employment Act labour requirements
- ✓ The contractor will develop and implement a children Protection Strategy to ensure that no child under the legal age of 18 years in employed to the Project
- ✓ The contactor should institute a security plan e.g. through a register for all visitors and workers
- ✓ The contractor will Adopt and adapt Nyumba Kumi strategies

7.4.2 Spread of COVID-19 amongst community members during consultations and construction

During project execution (civil works), large numbers of workers will be required to assemble together in consultation engagements, meetings, toolbox talks and even at work sites; varied number of workforce including suppliers of material and services are also expected to come in

from various places in the country which may be COVID-19 hot spots; and interaction of workers with the project host community will happen as workers find accommodation close to work sites, and/or return to their homes after works. There is also the risk that the project may experience large numbers of its workforce becoming ill and will need to consider how they will receive treatment, and whether this will impact on local healthcare services including the project host community.

Mitigation measures

- ✓ The Contractors will develop SOPs for managing the spread of Covid-19 during project execution and submit them for the approval of the Supervision Engineer and the Client before mobilization. The SOPs shall be in line with the World Bank guidance on COVID-19, Ministry of Health Directives and site-specific project conditions;
- ✓ Mandatory provision and use of appropriate Personal Protective Equipment (PPE) shall be required for all project personnel including workers and visitors;
- ✓ Avoid concentrating of more than 15 persons or workers at one location. Where more than one person is gathered, maintain social distancing of at least 2 meters
- ✓ All workers and visitors accessing worksites every day or attending meetings shall be subjected to rapid Covid-19 screening which may include temperature check and other vital signs;
- ✓ The project shall put in place means to support rapid testing of suspected workers for covid-19;
- ✓ Install handwashing facilities with adequate running water and soap, or sanitizing facilities at entrance to work sites including consultation venues and meetings and ensure they are used;
- ✓ Ensure routine sanitization of shared social facilities and other communal places routinely including wiping of work stations, door knobs, hand rails

7.4.3 Gender based violence and sexual harassment (GBV/SH)

This impact is triggered during project construction phase when the contractor(s) fail to comply with the following provisions:

- a) Gender Inclusivity requirements in hiring of workers and entire project management as required by Gender Policy 2011 and 2/3 gender rule; and
- b) Failure to protect human risk areas associated with, disadvantaged groups, interfering with participation rights, and interfering with labour rights.

The proposed Mitigation Measures of Human Rights and Gender Requirements are:

- \checkmark Ensure clear human resources policy against sexual harassment that is aligned with national law.
- ✓ Integrate provisions related to sexual harassment in the employee Code of Conduct.
- ✓ Ensure appointed human resources personnel to manage reports of sexual harassment according to policy.
- ✓ The contractor(s) shall require employees, sub-contractors, sub-consultants, and any personnel thereof engaged in construction works to individually sign and comply with a Code of Conduct with specific provisions on protection from sexual exploitation and abuse.
- ✓ The contractor(s) will implement provisions that ensure that GBV at the community level is not triggered by the project, including:

- \checkmark Effective and on-going community engagement and consultation, particularly with women and girls.
- ✓ Review of specific project components that are known to heighten GBV risk at the community level, e.g. compensation schemes; employment schemes for women; etc.
- ✓ The contractor shall develop specific plan for mitigating these known risks, e.g. sensitization around gender-equitable approaches to compensation and employment.
- ✓ The contractor will ensure adequate referral mechanisms are in place if a case of GBV at the community level is reported related to project implementation.

7.4.4 Child Abuse

Children within project areas will be exposed to risks associated with interaction between them and Project Workers. This includes child labour and sexual abuse which coherently leads to teenage pregnancies and exposure to communicable diseases such as HIV/AIDS. The contractor will undertake the following mitigation measures.

. Mitigation measures

- ✓ The contractor will develop and implement a Children Protection Strategy that will ensures minors are protected against negative impacts associated by the Project including on SEA...
- ✓ Children under the age of 18 years should not be hired on site as provided by Child Rights Act (Amendment Bill) 2014.
- ✓ Not invite unaccompanied children to workers home, unless they are at immediate risk of injury or in physical danger.
- ✓ Throughout the project phase the proponent shall refrain from hiring children for domestic or other labour, which is inappropriate given their age, or developmental stage, which interferes with their time available for education and recreational activities, or which places them at significant risk of injury.
- ✓ Comply with all relevant local legislation, including labour laws in relation to child labour specifically provisions of Kenya's Employment Act Cap 226 of 2007 Part VII on protection of children against exploitation

7.4.5 Sexual Exploitation and Abuse (SEA)

This impact refers to sexual exploitation and abuse committed by Project staff against communities and represents a risk at all stages of the Project, especially when employees and community members are not clear about prohibitions against SEA in the Project.

Mitigation Measures to Risk of SEA

- ✓ Develop and implement an SEA action plan with an Accountability and Response Framework as part of the C-ESMP. The SEA action plan will follow guidance on the World Bank's Good Practice Note for Addressing Gender-based Violence in Investment Project Financing involving Major Civil Works (Sept 2018). The SEA action plan will include how the project will ensure necessary steps are in place for:
- ✓ Prevention of SEA: including COCs and ongoing sensitization of staff on responsibilities related to the COC and consequences of non-compliance; project-level IEC materials;
- ✓ Response to SEA: including survivor-centered coordinated multi-sectoral referral and assistance to complainants according to standard operating procedures; staff reporting mechanisms; written procedures related to case oversight, investigation and disciplinary procedures at the project level, including confidential data management;
- ✓ Engagement with the community: including development of confidential communitybased complaints mechanisms discrete from the standard GRM; mainstreaming of PSEA awareness-raising in all community engagement activities; community-level IEC materials; regular community outreach to women and girls about social risks and their PSEA-related rights;

✓ Management and Coordination: including integration of SEA in job descriptions, employments contracts, performance appraisal systems, etc.; development of contract policies related to SEA, including whistle-blower protection and investigation and disciplinary procedures; training for all project management; management of coordination mechanism for case oversight, investigations and disciplinary procedures; supervision of dedicated PSEA focal points in the project and trained community liaison officers.

7.4.6 Risk of Increased incidences of HIV/AIDS and STIs

The influx of people may bring communicable diseases to the project area, including sexually transmitted infections (STIs), or the incoming workers may be exposed to diseases to which they have low resistance. This can result in an additional burden on local health facilities and resources. Local health and rescue facilities may also be overwhelmed and/or ill-equipped to address the industrial accidents that can occur in a large construction site.

Proposed mitigation measure for this are:

- ✓ Contractor(s) to sensitize workers and community members on HIV/AIDS awareness and other communicable diseases to be instituted and implemented as part of the contractor's Health and Safety Management Plan to be enforced by the Supervising Engineer. This will involve periodic HIV/AIDS and other communicable diseases Awareness Workshops for Contractor's Staff.
- ✓ Controlled access to contractor's workforce camps by outsiders.
- ✓ Contractor(s) to provide standard quality condoms at the construction site during the construction period.

7.4.7. Increased Occupational health and safety (OHS) risks

Construction work carries with its certain risks for worker health and safety. Risk of accidents is high for those who handle heavy machinery. The envisaged work does not require any form of explosive or blasting thus negative impacts are not foreseen. During construction phase, the risk of accidents will be from sources such as falling objects and machinery and inhalation of dust and electrocution. Occupational health and safety issues that occur during the operational phase in fish processing projects primarily include the following:

- ✓ Physical hazards
- ✓ Lifting, carrying, and repetitive work injuries
- \checkmark Exposure to chemicals
- \checkmark Exposure to heat
- ✓ Confined space
- ✓ Exposure to noise and vibrations

Mitigation measures

Occupational health and safety risks are likely to increase during construction and operation phases. The proposed mitigation measures against OHS risks include; The proponent will adhere to OSHA regulations and other related work place regulations and good practices. Specifically

- ✓ The contractor and proponent will provide PPEs such as helmets and gunboots to workers and ensure that they are using them at all time when engaged at the site
- \checkmark The Contractor to place labels and warning signs in areas posing risk of injury or accidents
- ✓ The contractor to have Incident and Accident Registers on site for recording of injuries or any OHS incidence
- ✓ Contractor and proponent to hire a qualified health and safety officer to oversee OSH issues
- \checkmark the contractor and proponent should ensure there is a temporal toilet/pit latrine

- ✓ Contractor to erect construction signage and secure the construction site to minimize risks of accidents and prevent injuries to the public
- ✓ The contractor and proponent to provide First Aid box and a trained person to handle site emergencies and incidences, while displaying emergency number for ambulances/provision of emergency vehicles

7.4.8. Risk of Accidents

During Construction phase, increased traffic flow into and through the site will occur. This increases the risk of accidents unless the traffic is properly controlled. Erection of proper signage and appropriate warning at least 100m from the hazard will mitigate against the chances of accidents. Hauling of equipment (plant and machinery) and other materials and supply to the project site may pose a potential risk of accidents to animals and even people, especially children.

Mitigation measures

During Construction phase, increased traffic flow into and through the site will occur. This increases the risk of accidents unless the traffic is properly controlled.

Proposed mitigation measures

- ✓ Drivers to be instructed not to speed especially near settlements when supplying materials to the site to prevent accidents especially to animals and children
- \checkmark Provision of PPEs to all workers by the contractor and proponent
- ✓ Installation of warning signage at the construction site and identified
- ✓ Contractor to implement a fall protection program that includes training in climbing techniques and use of fall protection measures
- ✓ Use of helmets and other protective devices

7.4.9 Risks of increased spread of COVID-19 at work sites

During project execution (civil works), large numbers of workers will be required to assemble together in meetings, toolbox talks and even at work sites; varied number of workforce including suppliers of material and services are also expected to come in from various places in the country which may be COVID-19 hot spots; and interaction of workers with the project host community will happen as workers find accommodation close to work sites, and/or return to their homes after works. The potential for the spread of any infectious disease like COVID-19 by projects is high. There is also the risk that the project may experience large numbers of its workforce becoming ill and will need to consider how they will receive treatment, and whether this will impact on local healthcare services including the project host community. The presence of international workers, especially if they come from countries with high infection rates, may also cause social tension between the foreign workers and the local populations.

The proposed Mitigation Measures against spread of COVID-19 amongst workers are:

- a) The contractor(s) shall put in place measures to prevent and manage the spread of the COVID-19.
- b) The contractor(s) will develop a SOPs for managing the spread of COVID-19 during project execution and submit them for the approval of the Supervision Engineer and the Client before mobilization. The SOPs shall be in line with the World Bank guidance on COVID-19, Ministry of Health Directives and site-specific project conditions.

- c) Mandatory provision and use of appropriate Personal Protective Equipment (PPE) shall be required for all project personnel.
- d) The project shall put in place means to support rapid testing of suspected workers for COVID-19.
- e) Avoid concentrating of more than 15 persons or workers at one location. Where more than one person are gathered, maintain social distancing at least 2 meters. All workers and visitors accessing worksites every day or attending meetings shall be subjected to rapid COVID-19 screening which may include temperature check and other vital signs.
- f) Install handwashing facilities with adequate running water and soap, or sanitizing facilities at entrance to work sites including consultation venues and meetings and ensure they are used.
- g) Ensure routine sanitization of shared social facilities and other communal places routinely including wiping of workstations, door knobs.

7.5 Impacts during the operation phase

7.5.1 Anticipated positive impacts during operation phase

7.5.2 Reduced post-harvest losses

With the Omena processing facility, the BMU the potential to plan the marketing and even forward contracts and market hedging hence improved earnings that it currently dictated by middlemen. The reduced losses that can reach up to 50 % of the Omena and to scavengers or through rotting and contamination hygienic processing that will be curtailed by hygienic processing will highly lead to increased earning to the fisherfolk.

7.5.3 Resilience building in the community

Hygienic processing reduces market and price risks especially the selling of Omena products at throw way prices to middlemen during wet/ cold seasons. The stabilized prices means increased income and resilience of the community to weather and market related risks. The subproject intervention will contribute to creation of decent jobs, increase value of the omen and support many livelihoods hence resilience of the community.

7.5.4 Improved food and nutritional security

Sanitation goes hand in hand with food security. The high standards in handling of the fish products will increase access to the nutritious yet affordable fish products at the community level and in the Kenyan market at large thus enhancing food security outcomes.

7.5.5. Livelihood diversification

The proposed project will enhance livelihood diversification through sustainable land management and natural resource management interventions, projects and farm level conservation works. Diversification will contribute towards employment creation, income generation and food security in the project area.

7.5.6 Biodiversity conservation and enhancement

Landscaping and planting of trees and ornamental plants at the processing plant will enhance the aesthetics and spur the community to adopt the same. Tree planting will enhance biodiversity of the area. The group nursery will provide seedlings to the community and enhance environmental conservation

7.5.7 Improved beach hygiene and public health

The proposed construction of sanitary facilities (bio-digester technology) in the project and

fencing of the Omena drying facility will mean less contamination and pollution of the water and the environment. All this will result in improved sanitation and risk of water borne diseases. The control of scavenging birds (plate 11) will also reduce the risk of spreading zoonotic diseases to humans and poultry in the area.



PLATE 11: BIRDS FEEDING ON OMENA AND OCHONGO DURING DRYING: This increases contamination, loss of quality and economic losses to the Fisherfolk. The processing facility has the potential to reverse this.

7.5.8 Enhanced food safety

Open drying increases the chances of cross contamination especially with outbreak of cholera, and salmonella, a common occurrence along the beaches and fish products. With the improved standards in handling and processing the fish products .i.e. adherence to HACCP protocols will greatly reduce the likelihood of contamination of the products thus eliminating spread and risk of such water borne diseases.

7.6 Anticipated negative environmental impacts during operation phase

7.6.1. Health risks and Environmental contamination from pesticides

There will be minimal use of fumigants or agrochemicals. The fumigation of the factory during processing and use of anti- termites in the foundation during construction is also envisaged.

Mitigation measures

- Proponent shall promote use of Integrated Pest Management in an event that there will be pests
- ✓ If absolutely necessary then application of bio-pesticides will be considered.

7.6.2. Increased water usage

Fish processing uses a lot of water in washing the raw products and cleaning the processing premises. The use of detergents in this processes increases environmental pollution risks hence it is critical in mitigation of the adverse environmental impacts. Water for the processing facility will supplied through the appropriate service provider.

Mitigation measures

- \checkmark Use high pressure rather than high volume for cleaning
- ✓ Install meters to monitor water usage
- ✓ Pre-soak floors and equipment to loosen dirt before final cleaning
- \checkmark Reuse relatively clean waste water for other applications
- ✓ Consider use of compressed air instead of water where possible
- ✓ Use pressurised spray nozzles to increase effectiveness in cleaning of surfaces and reduce water consumption
- ✓ the wastewater from the final rinse can be collected and used for the initial rinse on the following day
- ✓ determining the required amount or concentration for effective cleaning scraping followed by an initial rinse to reduce the consumption of detergents
- \checkmark Put in place rain water harvesting structures by constructing roof catchment

7.6.3 Increased energy use

The Omena drying processing will require electricity to operate the drier, provide lighting, run air compressors and production of steam and hot water for cleaning and sanitising.

- \checkmark Use of solar to generate energy will be given priority
- Proponent to implement switch-off programs and install sensors to turn off or power down lights and equipment when not in use
- ✓ During operation phase BMU to ensure regular maintenance as to optimise energy efficiency of equipment and schedule maintenance activities on regular basis to avoid inefficiencies and breakdowns

7.6.4 Fire

Fire damage is unpredictable given that the proposed project will not involve a lot of flammable materials. However, if appropriate measures are not put in place, a fire outbreak can accidentally occur and cause damage to property and even lead to death.

Mitigation measures

- ✓ Declare places with flammable construction materials as "NO SMOKING ZONES" and display conspicuous notices of the same.
- ✓ Train workers and the management on emergency (fire) preparedness and management.
- ✓ Provide fire extinguishers at the site e.g. at the fish hatchery, at the fish feeds formulation unit, at the office etc.

7.6.5 Solid wastes

Increase in waste generation will come from packaging materials and operations such as when separating omen from chaff.

Mitigation measures

- ✓ Provide bins for separate collection of wastes into appropriate sorts such as recyclable and non-recyclable, general wastes and infectious wastes e.g. used masks and label the collection bins appropriately.
- ✓ Where possible material considered as waste may be re-used or recycled or be given to who may consider them useful for others uses.
- \checkmark Order materials according to needs to reduce waste.
- ✓ Maintain and repair equipment rather than replacing it to reduce waste.
- ✓ Regularly collect solid wastes and dispose them to prevent accumulation at collection areas.
- ✓ Cover the solid waste collection areas to minimize invasion by pests and rodents or other animals.

7.6 Anticipated negative social impacts during operation phase

7.6.1 Gender based violence and sexual harassment (GBV/SH)

This impact is triggered during project operation phase when the Proponent or project management fail to comply with the following provisions:

- Gender Inclusivity requirements in hiring of workers and entire project management as required by Gender Policy 2011 and 2/3 gender rule; and
- Failure to protect human risk areas associated with, disadvantaged groups, interfering with participation rights, and interfering with labour rights.

The proposed Mitigation Measures of Human Rights and Gender Requirements are:

- a) Ensure clear human resources policy against sexual harassment that is aligned with national law.
- b) Integrate provisions related to sexual harassment in the employee Code of Conduct.
- c) Ensure appointed human resources personnel to manage reports of sexual harassment according to policy.

- d) Ensure all employees and any personnel thereof engaged in the project implementation to individually sign and comply with a Code of Conduct with specific provisions on protection from sexual exploitation and abuse.
- e) Implement provisions that ensure that GBV at the community level is not triggered by the project, including:
- f) Effective and on-going community engagement and consultation, particularly with women and girls.
- g) Review of specific project components that are known to heighten GBV risk at the community level, e.g. compensation schemes; employment schemes for women; etc.
- h) Develop specific plan for mitigating these known risks, e.g. sensitization around gender-equitable approaches to compensation and employment.
- i) Ensure adequate referral mechanisms are in place if a case of GBV at the community level is reported related to project implementation.

7.6.2 Sexual Exploitation and Abuse by project workers against community members

This impact refers to sexual exploitation and abuse (SEA) committed by project staff against communities and represents a risk at all stages of the project, especially when employees and community members are not clear about prohibitions against SEA in the project.

The proposed mitigation measures to risks of SEA include:

- a) Develop and implement a SEA action plan with an Accountability and Response Framework as part of the C-ESMP. The SEA action plan will follow guidance on the World Bank's Good Practice Note for Addressing Gender-based Violence in Investment Project Financing involving Major Civil Works (Sept 2018).
- b) The SEA action plan will include how the project will ensure necessary steps are in place for:
- c) Prevention of SEA: including Code of Conducts and ongoing sensitization of staff on responsibilities related to the Code of Conducts and consequences of non-compliance; project-level IEC materials.
- d) Response to SEA: including survivor-centered coordinated multi-sectoral referral and assistance to complainants according to standard operating procedures; staff reporting mechanisms; written procedures related to case oversight, investigation and disciplinary procedures at the project level, including confidential data management.
- e) Engagement with the community: including development of confidential communitybased complaints mechanisms discrete from the standard GRM; mainstreaming of PSEA awareness-raising in all community engagement activities; community-level IEC materials; regular community outreach to women and girls about social risks and their PSEA-related rights.
- f) Management and Coordination: including integration of SEA in job descriptions, employments contracts, performance appraisal systems, etc.; development of contract policies related to SEA, including whistle blower protection and investigation and disciplinary procedures; training for all project management; management of coordination mechanism for case oversight, investigations and disciplinary procedures; supervision of dedicated PSEA focal points in the project and trained community liaison officers.

7.6.3 Risk of Increased incidences of HIV/AIDS and STIs

The influx of people may bring communicable diseases to the project area, including sexually transmitted infections (STIs), or the incoming workers may be exposed to diseases to which they have low resistance. This can result in an additional burden on local health facilities and resources. Local health and rescue facilities may also be overwhelmed and/or ill-equipped to address the industrial accidents that can occur in a large construction site.

- a) Sensitize workers and community members on HIV/AIDS awareness and other communicable diseases to be instituted and implemented as part of the contractor's Health and Safety Management Plan to be enforced by the Supervising Engineer. This will involve periodic HIV/AIDS and other communicable diseases Awareness Workshops for Contractor's Staff.
- b) Controlled access to private offices and working places by outsiders.
- c) Provide standard quality condoms at the site at all times.

7.6.4 Grievances/conflicts

Common grievances expected to arise during the proposed project implementation include:

- a) Disruption of income streams, physical harm, and nuisance from operation activities;
- b) Health and safety risks;
- c) Socially-unacceptable omena drying facility staff relations with the communities and other stakeholders;
- d) Conflicts over water sources; and
- e) Pollution and other environmental related impacts.

Mitigation measures

The following are possible mitigation measures to manage grievances:

- a) Establish a grievance redress mechanism (GRM) for the proposed project;
- b) Seek to establish amicable relationships with stakeholders and manage the impact of the project activities on affected communities;
- c) Put in place a pre-emptive community liaison structure aimed at identifying potential issues arising from project-related impacts and addressing them before they become grievances;
- d) Establish a grievance redress mechanism targeting communities and other project stakeholders but not applicable to commercial and employee-employee relationships, and which will allow stakeholders to easily put forth their concerns relating to the project, implementation and have them addressed in a prompt and respectful manner;
- e) Ensure the grievance redress mechanism is available to the affected community members and stakeholders at no cost;
- f) Address all raised grievances, real or imagined and take reasonable steps to maintain confidentiality of the parties to the mechanism and regardless of the complainants' participation in this process, give a guarantee that the complainant's statutory rights to undertake legal proceedings remain unaffected; and
- g) Educate all project stakeholders on the availability and use of the grievance redress mechanism in a manner that is understandable to all, before, during and after construction of the proposed project.

7.6.5 Child abuse

Children within the project area will be exposed to risks associated with interaction between them and workers in the drying facility. This includes child labour and sexual abuse which coherently leads to teenage pregnancies and exposure to communicable diseases such as HIV/AIDS.

- a) The Proponent will develop and implement a Children Protection Strategy that will ensure minors are protected against negative impacts associated with the project.
- b) All staff must sign, committing themselves towards protecting children, a contract which clearly defines what is and is not acceptable behaviour.
- c) Children under the age of 18 years will not be hired at the site as provided by Child Rights Act (Amendment Bill) 2014.
- d) Refrain from hiring children for domestic or other labour, which is inappropriate given their age, or developmental stage, which interferes with their time available for education and recreational activities, or which places them at significant risk of injury.
- e) Comply with all relevant local legislation, including labour laws in relation to child labour specifically provisions of Kenya's Employment Act, 2007 (Cap. 226) Part VII on protection of children against exploitation.

7.6.6 Risk of increased spread of COVID-19

During project operation, there will be a lot of interactions among different people at the site. The potential for the spread of any infectious disease like COVID-19 is high. There is also the risk that the project may experience large numbers of its workforce becoming ill and will need to consider how they will receive treatment, and whether this will impact on local healthcare services including the project host community.

The proposed Mitigation Measures against spread of COVID-19:

- a) The Proponent shall put in place measures to prevent and manage the spread of the COVID-19.
- b) The Proponent will develop a SOPs for managing the spread of COVID-19 during project operation. The SOPs shall be in line with the World Bank guidance on COVID-19, Ministry of Health Directives and site-specific project conditions.
- c) Mandatory provision and use of appropriate PPE shall be required for all project personnel.
- d) The project shall put in place means to support rapid testing of suspected workers for COVID-19.
- e) Avoid concentrating of more than 15 persons or workers at one location. Where more than one person are gathered, maintain social distancing at least 2 meters. All workers and visitors accessing worksites every day or attending meetings shall be subjected to rapid COVID-19 screening which may include temperature check and other vital signs.
- f) Install handwashing facilities with adequate running water and soap, or sanitizing facilities at entrance to work sites including consultation venues and meetings and ensure they are used.
- g) Ensure routine sanitization of shared social facilities and other communal places routinely including wiping of workstations, door knobs.

7.6.7 Impacts related to occupational and public/community safety and health

There are three main types of occupational health and safety hazards that may be of concern. These are physical, chemical and biological. Potential physical hazards will include noise and accidents. Chemical hazards will involve exposure to harmful gases and chemicals by inhalation, ingestion and skin contact. Biological hazards involve exposure to pathogenic organisms which may cause diseases. Specific areas of concern include fire hazards, noise and vibrations, congestion, body contact, failure to observe social distancing thus exposing other people to COVID-19, poor sanitation, gender-based violence, sexual harassment and accidents at the site. Poor sanitation could result from presence of potential environmental pollutants at the site including wastewater, decomposing solid wastes, dust and exhaust emissions. Accidents including cuts, pricks and bruises; electrocution from naked electrical cables; falling in uncovered holes and/or trenches and from raised places and suffocation from lack of oxygen in confined spaces. Accidents could result from lack of supervision and job training, improper handling of machinery and hand tools and inappropriate carrying out of tasks.

Mitigation measures

Mitigation options to some of the impacts have been discussed. Additional mitigation measures to other impacts are:

- a) Supervise all works at the site e.g. plumbing, masonry, etc.
- b) Support all structures under construction.
- c) Keep all passages clear at all times.
- d) Adopt proper working procedures and when working with chemicals, machines and equipment.
- e) Fence the site for protection, privacy, reduction of trespass and theft, and control of entry by straying animals and therefore avoid conflicts between people at the site and the people in the neighborhood.
- f) Clean all spilt hazardous materials using an appropriate disinfectant.
- g) Ensure that trained first aid personnel are always available on site to handle emergencies.
- h) Have a fully equipped First Aid Kit (containing a first aid manual and is equipped with sterile adhesive bandages, safety pins, cleansing agent/soap, latex gloves; sterile gauze pads triangular bandages, non-prescription drugs, scissors, tweezers and antiseptic amongst others) at the site at all times.
- i) Put in place an appropriate emergency response plan including having emergency contacts (such as ambulance, fire tender and police) conspicuously displayed.
- j) Dispose wastes from the site regularly and ensure high standards of cleanliness of all waste collection and disposal facilities.
- k) Ensure adequate water supply for high standards of sanitation that keeps to the minimum chances of disease outbreaks.
- Frequently undertake workers through refresher courses in order to make them have a basic understanding of the tasks under them, the hazards involved, and how to manage them.
- m) Construct a pit latrine at the site and always keep it clean.
- n) Conduct regular maintenance of the proposed site and facilities thereat to increase the life of the proposed building making it safe for habitation.
- o) Ensure employee welfare including provision of free or subsidized medical attendance if injured on work, making provisions for leaves and offs, and operation of shorter-shift period for workers in highly polluted working areas.
- p) Provide appropriate PPE including face masks, goggles, scarfs, boots and overalls among other protective clothing to all workers and people at the site and sensitize them to use them whenever they are in environments that warrant the use of such PPE especially in all situations where the body and skin are potentially exposed to hazards such as chemicals, harmful dusts, highly infectious wastes, sharp objects, burns and extreme temperature and/or when working in areas that present threatening experiences.
- q) Ensure high standards of construction as recommended in the approved structural designs and regular maintenance to increase the life of the structures at the site.

7.7. Project Lifespan and Decommissioning

In the unlikely event of decommissioning, the BMU shall be expected to demolish the facility and restore the host environment close to its original state. Decommissioning is expected after 50 years, or if the capacity is to be upgraded in future. Possible impacts are primarily related to disposal and handling of the inert and non-biodegradable wastes from demolition. The demolition exercise shall involve the following:

- \checkmark Demolishing and removal of all the concrete works
- ✓ Demolishing and removal of all the sanitary utilities
- ✓ Demolishing and removal of the wooden and roofing materials
- ✓ Carefully removing all the electrical fittings and associated cables
- ✓ Ensuring proper handling of the demolished materials and have an authorized and guided transportation and disposal away from the site
- \checkmark The scrap metal will be sold for recycling

7.4.1 Negative Environmental and Social impacts and Mitigation measures during decommissioning

7.4.1 Negative impacts

The negative impacts during decommissioning are same as those in the construction phase.

7.4.1.1. Noise and Vibration Generation

Continuous exposure to noise levels above 85 dB may cause hearing problems leading to occupational deafness. Noise and vibration produced during decommissioning may have some temporary negative impacts to the immediate residents. These may include impairing verbal communication, temporary hearing problems/temporary threshold shift (TTS), noise annoyance or even interference of the normal behaviour of domestic.

Proposed mitigation and management measures

To control the generation of noise and ensuring compliance with relevant noise standard include:

- \checkmark Avoid idling of machinery or engine when not in use.
- \checkmark Restrict activities that create noise to daytime only.
- ✓ Provision of Personal Protective Equipment and clothing (PPE/C) to those actively engaged in the works at the site

7.4.1.2. Motor vehicle emissions

The potential sources of air pollution include traffic emission from excavator and material transport vehicles. This impact is considered low risk. The proposed mitigation and management measures include;

✓ Ensure that maintenance on all machinery is done regularly to avoid the emission of noxious gases.

✓ Drivers and machine operator to avoid unnecessary running of motor vehicle engines and machinery when not in use

7.4.1.3. Oil spills/Fuels and Lubricants

Oils and grease spillage on the ground may cause contamination to the soil and groundwater. Proposed mitigation and management measures are:

- ✓ Proper maintenance of vehicles and other equipment (using petroleum products) to avoid fuels and lubricants spills at the project site.
- ✓ The contractor should properly handle, storage, and disposal off oils and greases and their wastes during construction by ensuring that servicing is strictly done at designated servicing yard or external petroleum stations

7.4.1.4 Solid Waste Generation

The decommissioning of the project will entail a small scale of removal of structures which is expected to be in small amounts with low impact.

The proposed mitigation measures are as follows

- ✓ The contractor should ensure that construction wastes generated and not reusable or recyclable is cleared from the project site and disposed of accordingly in line with appropriate law by the regulatory agency at the region which is NEMA and County Government.
- ✓ Contractor to practice waste separation to enable easy recycling of re-usable waste materials
- ✓ Contractor to provide temporal waste disposal receptacle in site.
- ✓ Contractor to liaise with licensed waste collector to routinely collect and dispose the waste

7.4.1.5. Occupational Health and Safety (OHS)

During decommissioning of the auxiliary structures (latrine, livestock troughs) and other masonry structures accidents may also be caused by falls in excavated sites or injuries by wrongly placed equipment, tools or other construction materials.

The proposed mitigation measures

The Contractor to place labels and warning signs in areas posing risk of injury or accidents

- \checkmark The contractor labels and warn the public on the danger of decommissioning activities
- \checkmark The contractor to provide all workers with full protective gear
- ✓ The contractor to train and provide First-aid Kit to the workers
- ✓ The contractor to have a book (Incident and Accident Registers) on site for recording of injuries or any OHS incidence
- ✓ Contractor to prepare a contingency / emergency management and preparedness plan for accident response
- ✓ Ensure the availability of Emergency contacts for police, ambulance

7.4.1.6 Spread of COVID-19 amongst workers

(ii) Avoid concentrating of more than 15 community members at one location. Where two or more people are gathered, maintain social distancing of at least 2 meters.

(iii) The team carrying out engagements within the communities on one-on-one basis will be provided with appropriate PPE for the number of people they intend to meet.

7.5. Positive Impacts during decommissioning

Several positive impacts include short term employment and recycling of the material for income generation. Should decommissioning involve expansion of the site, then the positive impact will include expansion of area under horticultural production and amount of water available to wider area of the location.

CHAPTER EIGHT

8.0. ENVIRONMENTAL AND SOCIAL MANAGEMENT AND MONITORING PLAN

Environmental and Social Management& Monitoring Plan (ESMMP) provides a logical framework within which the negative environmental and social impacts identified during the ESIA study can be mitigated and any beneficial environment effects can be enhanced. Monitoring and management practices as well as monetary compensation are considered and cost estimates included as applicable. Responsibilities and time frames for the implementation of the various aspects of the ESMMP will be identified. The ESMMP will be provided to prospective bidders for the construction contracts to ensure that environmental mitigation costs are factored into their costing. The Contractor(s) will also be required to prepare a separate and specific ESMMP for their works in order to control construction impacts and ensure compliance with applicable environmental and health and safety legislation and standards. KCSAP especially the County Project Coordination Unit and County Environment and Social Officers (CESOs) will ultimately be responsible for ensuring that the ESMMP is implemented on site via reviewing the Contractor's ESMMP and ensuring its implementation on site via audits.

The ESMMP has been developed to assist in prioritization of key findings of the ESIA mitigation measures. The ESMMP is based on ISO 14001 principles comprising the following: The environmental issues of concern, recommended mitigation measures, responsibilities, timeframes and costs (Table 4). The Environmental and Social Management& Monitoring Plan also includes environmental monitoring measures with the following objectives:

- \checkmark To verify the execution of the measures proposed in the ESIA and to evaluate the effectiveness of these measures.
- \checkmark To verify if the impacts anticipated in the ESIA have occurred and to detect environmental problems that could not have been identified previously, in order to adopt solutions adequate for the conservation of the environment.
- ✓ To provide reliable information to be used for the verification of environmental impacts with the purpose of improving the techniques of prediction of environmental impacts and the quality and opportunity of application of corrective measures. The monitoring program will cover, among others, the following aspects.

TABLE 8: SUMMARY OF ENVIRONMENTAL AND SOCIAL MANAGEMENT MONITORING PLAN- KAMARIGA BMU FISH DRYING PLANT

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
Clearance of	Replant the area with grass and indigenous		Design Engineer	2 months	No. of	Requisition invoices
vegetation, soil	tree species to replace the lost plants	Contractor	and contractor		seedlings	Site observation
erosion and loss of	Protect the lake shore at the landing site	cost	Contractor, CPU		replanted and	Interviews
biodiversity				2months	surviving	
						Site observations
Waste management	Provide solid waste collection facilities and	Contractor	BMU and	1 month	Number of	Records of tonnage
	encourage waste segregation through	cost	contractor		licences waste	of waste
	sensitizing workers and community on				handlers	management
	waste management practices					facilities
					Number and	
	engage licensed waste handler to regularly				type of	Reports/Observation
	collect and dispose the wastes				suitable waste	
	ensure recycling of recyclable wastes such				disposal	
	as paper, metals, plastics				facilities	
					installed	
	BMU to apply waste reduction strategy					
	that collects marketable waste such as				Amount in	
	flakes and market them as chicken feed and				tonnes of	
	organic fertilizer				flakes that	
					have been	
					marketed as	
					chicken feed	

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
Hydrological	install roof harvesting to reduce	Contractor	Contractor	Constructor	Roof area	Reports
Impacts		cost			designated for	
	allow for natural seepage of water by				rain water	
	replanting disturbed areas with grass and				harvesting	
	indigenous trees					
Health Operation	Comply with places of work regulation and	Contractor	Contractor BMU	Continuous	Number and	DOSH licenses
and safety of the	ensure health and safety of the workers and	cost			length of side	Audit report
processing facility	community				drains installed	
	Install side drains				F · 1	D 1 1
	Establish an environmental audit and				Environmental	Records and
	protocol and schedule as per the EIA/Audit				Audit Report	observations
	regulations					
	Secure the factory, through construction of				Length of	Reports/site visit
	a chain link fence and locked at all time				chain-link	Reports/site visit
	a chain hink fonce and focked at an time				fence	
	Provide appropriate protective clothing					Observation
	(overalls, head covers/caps, gloves, nose				Number and	
	muffs) and training of workers on use of				types of PPE	
	these PPEs.				provided	
					-	

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
Effluent offload	Sweep up solid materials for use as a by-	Contractor	Contractor	Continuous	Amount of	Reports
	product, instead of washing them down the	cost			Solid materials	
	drain				kept for use	
	Fit drains with screens and/or traps to					
	prevent solid materials from entering the				Length of	
	effluent system.				screens and	Reports
					traps installed	
					in the drainage	
					systems	
Surface runoff		Contractor	Contractor	1 Month	Length of	Reports
	Stone pitch the waterways around the	cost			stone pitches	_
	building				around the	
	Sensitization and afforestation of the				proposed	
	landscape/catchment				building	
					Number of	
					trees grown	
					around the	
					landscape	
Oil spills/Fuels and	Proper maintenance of vehicles and other	Contractor	Contractor	Continuous	Number and	Maintenance record
Lubricants	equipment (using petroleum products) to	cost			types of	register of vehicles
	avoid fuels and lubricants spills at the				vehicles and	
	project				machines	
					Date of due	
	properly handle, storage, and disposal off				service	
	oils and greases and their wastes during					Stores Ledger cards
	construction by ensuring that servicing is				Amount and	for the oils and
	strictly done at designated servicing yard				types of oils	grease
	or external petroleum stations				and greases	

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
Air, dust and noise	\checkmark People participating in the	Contractor	Contractor	Continuous	Number and	
pollution	construction should be provided	cost			types of PPEs	
	with Personal Protective				i.e ear	
	Equipment (PPE) such as ear				protectors	
	muffles for ear protection					
	\checkmark Sprinkling to minimize dust					
	emission during construction				Amount of	
	\checkmark Restrict construction activities to				litres of water	
	day time				sprinkled	
Effluents and	provide oil sumps at the construction yard	Contractor	Contractor	Continuous	Number of	Stores ledger register
pollution of water	and/or service and fuel at registered oil	cost			containers for	
sources	dealers yards necessary measures to prevent				storing oil	
	oil and grease spills and soil erosion which				sumps at the	
	may pollute the water				yard	
	~					Reports/observation
	Community sensitization and awareness on				Number and	
	sanitation				size of septic	
	Construct ablution block with bathrooms				tank	Reports/observation
	Construct a septic tank				constructed	•
					Number of	
					ablution block	
					constructed for	
					males and	
					females	

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
Increased Energy	Install Solar powered energy equipment's		Contractor, and	1 month	Number ad	Reports/Observation
consumption		0.4	BMU		types of	
_	Implement switch-off programs and install				Energy	
	sensors to turn off or power down lights				efficiency	
	and equipment when not in use				equipments	
					For example	
	Ensure regular maintenance as to optimise				Number of	
	energy efficiency of equipment and				energy saving	
	schedule maintenance activities on regular				bulbs,	
	basis to avoid inefficiencies and					
	breakdowns				Number of	
					watts of solar	
					panels	
					installed	
Aerial emissions	Cover all transfer systems, wastewater	Contractor	BMU/Contractor	1 month	Amount of raw	Stores ledger in the
	drainage systems , and wastewater	cost			materials and	warehouse
	treatment facilities to reduce the escape of				waste by	
	foul odours				products in the	
					warehouse	
					Length of	Reports
					cover of the	
					drainage	
					systems	
Anticipated negative	Social and health impacts during construct	ction phase				

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
Labour Influx Effects	Ensure effective community engagement and strong grievance mechanisms on matters related to labour, with a discrete mechanism for safely and confidentially reporting issues of SEA and GBV at the community level triggered by the Project The contactor should institute a security plan e.g. through a register for all visitors and workers The contractor will Adopt and adapt Nyumba Kumi strategies	Contractor cost	Contractor/BMU	Continuous	Number of immigrants seeking jobs Number of workers and vsitors registered in the work premise	Register of the construction worker
Child Abuse	The contractor will develop and implement a Children Protection Strategy that will ensures minors are protected against negative impacts associated by the Project including on SEA Children under the age of 18 years should not be hired on site as provided by Child Rights Act (Amendment Bill) 2014.	.Contractors cost	Contractor/BMU		Number of persons below the age of 18 employed in the project	Reports observation
Risks of Increased HIV and Aids transmission in the area	The Contractor shall institute HIV/AIDS awareness and prevention campaign amongst his workers for the duration of the contract, contracting and implementing organisation, with preference for an organisation already working on this issue in the Project area;	.Contractors cost	Contractor/BMU	continuous	Number of Trainings on HIV/AIDs Held Number of cases reported on HIV/A infections	Ministry of health records on HIV/AIDs cases reported during the construction period

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
GBV at community level	The contractor will create awareness on gender-based violence at the community level and for the workers The contractor will ensure adequate referrals mechanisms are in place if a case of GBV at the community level is reported related to project implementation Separate toilets for different gender	0.2M	Contractor/BMU	continuous	Number of Training on GBV Number of GBSV cases reported in the GRM registers	Minutes and records of training Invitation register Proceedings of the training Interviews with trainees GRM register
Sexual Exploitation and Abuse by project workers against community members	Response to SEA: including survivor- centered coordinated multi-sectoral referral and assistance to complainants according to standard operating procedures; staff reporting mechanisms; written procedures related to case oversight, investigation and disciplinary procedures at the project level, including confidential data management. Engagement with the community: including development of confidential community-based complaints mechanisms discrete from the standard GRM;	-	Contractor/BMU		Number of incidences of SEA reported Number of confidential community- based complaints mechanisms discrete from the standard GRM;	Proceedings from Interviews with communities Reports GRM Register
Grievance Redress	Address all raised grievances, real or	-	Contractor/	continuous	Number of	Minutes and records
Mechanisms	imagined and take reasonable steps to		CPCU and BMU		grievances	of meetings with

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
	maintain confidentiality of the parties to the mechanism and regardless of the complainants' participation in this process , give a guarantee that the complainant's statutory rights to undertake legal proceedings remain unaffected;				raised during the project phase	names, telephone numbers of the aggrieved parties
Spread of COVID- 19 amongst workers	Mandatory provision and use of appropriate Personal Protective Equipment (PPE) shall be required for all project personnel including Avoid concentrating of more than 15 persons or workers at one location. Where	0.2M	BMU Supervising Eng. & Contractor(s)	continuous	Availability of SOP(s), Training material, PPE, sanitising facilities etc;	Requisition invoices Site observation Interviews Archived records
	more than one person is gathered, maintain social distancing at least 2 meters All workers and visitors accessing worksites every day or attending meetings shall be subjected to rapid Covid-19 screening which may include temperature check and other vital signs; The project shall put in place means to support rapid testing of suspected workers for covid-19; Install handwashing facilities with adequate running water and soap, or sanitizing facilities at entrance to work sites including consultation venues and meetings and ensure they are used; Ensure routine sanitization of shared social facilities and other communal places routinely including wiping of workstations, door knobs, hand rails etc;					

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
Issue						
15540						
Anticipated Negative	e Environmental impacts during operation	phase				
Contamination of	Formulate an IPM guideline	0.3 M	BMU, CPU,	Continuous	Number of	Minutes and records
Omena and the by	Train farmers on safe use of pesticides		Farmers		Trainings on	of training
products on	Procure PPE demo kits and hold				IPM held	Invitation register
pesticides	demonstrations at farm level					Proceedings of the
					Committees on	training
					IPM formed	Interviews with
		0.0	DIG			trainees
Effluent discharge	Use pressurised spray nozzles to increase	0.2	BMU	Continuous	Number of	Reports
	reduce water concurrentian				enecuve	
	Determine the required amount or				pressurized	
	concentration for effective cleaning				installed in the	
	scraping followed by an initial rinse to				fish facility	
	reduce the consumption of detergents				non racinty	
	Use dry cleaning techniques where					
	possible, by scraping equipment before					
	cleaning, pre-cleaning with air guns and					
	cleaning floor spills with squeegees					
Health risks and	Proponent shall promote use of	0.05	BMU work	Continuous	Amount in	Reports
Environmental	Integrated Pest Management in an		closely with the		litres and types	
contamination from	event that there will be pests		county dept of		of bio	
pesticides	\checkmark If absolutely necessary then		public health		pesticides	
	application of bio-pesticides					
	\checkmark Sensitization and awareness				Number of	
	training on pesticide risks				trainings on	
	\checkmark Contiguous sensitization on the				the use of IPM	
	proper use of pesticides				and	
	✓ Put in place a PMP				biopesticides	
					1	

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
					PMP	_
Fire	 ✓ Declare places with flammable construction materials as "NO SMOKING ZONES" and display conspicuous notices of the same. ✓ Train workers and the management on emergency (fire) preparedness and management. Provide fire extinguishers at the site e.g. at the fish hatchery, at the fish feeds formulation unit, at the office etc. 	0.02	BMU	Continuous	Number and size of fires extinguishers installed Number of trainings on fire management and preparedness	Reports
Solid wastes	Provide bins for separate collection of wastes into appropriate sorts such as recyclable and non-recyclable, general wastes and infectious wastes e.g. used masks and label the collection bins appropriately. Cover the solid waste collection areas to minimize invasion by pests and rodents or other animals.	0.02	BMU	Continuous	NumberandtypeofsuitablewastedisposalfacilitiesfacilitiesinstalledNumberoflicenceswastehandlersAmountintonnesofwastecollectedatanyone time	Records of tonnage of waste management facilities Reports/Observation Reports/Observation
Anticipated negat	ive social impacts during operation					
Phase						

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
Gender based	✓ .	0.05	BMU	Continuous	Number of	Reports
violence and sexual	\checkmark Implement provisions that				reported cases	
harassment	ensure that GBV at the				on GBV	GBV Register
(GDV/SII)	community level is not					
	triggered by the project,				Number of	
	including:				Number of	
	✓ Effective and on-going				engagement	
	community engagement and				and	
	consultation, particularly with				consultations	
	women and girls.					
	✓ Ensure adequate referral				Number of	
	mechanisms are in place if a				referral	
	case of GBV at the community				mechanisms	
	level is reported related to					
	project implementation.					
Sexual Exploitation	✓ Response to SEA: including	0.05	BMU	Continuous	Number of	Reports
and Abuse by	survivor-centered coordinated				reported cases	SEA Register
project workers	multi-sectoral referral and				on SEA	
against community	assistance to complainants					
members	according to standard operating					
	procedures; staff reporting					
	mechanisms; written					
	procedures related to case					
	oversight, investigation and					
	disciplinary procedures at the					
	project level, including					
	confidential data management.					
	Engagement with the community:					
	including development of confidential					

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
	community-based complaints					
	mechanisms discrete from the standard					
	GRM;					
Risk of Increased	\checkmark Sensitize workers and	0.01	BMU/CPCU	Continuous	Number of	Reports
incidences of	community members on				workers	
HIV/AIDS and STIs	HIV/AIDS awareness and other				sensitized	
	communicable diseases to be					
	instituted and implemented as				Number of	
	part of the contractor's Health				Condoms 01	
	and Safety Management Plan				availed to	
	\checkmark Controlled access to private				workers	
	offices and working places by				Workers	
	outsiders.					
	Provide standard quality condoms at					
	the site at all times.					
Grievances/conflicts	\checkmark Establish a grievance redress	-	BMU/CPCU	Continuous	Number of	Reports
	mechanism (GRM) for the				GRM cases	
	proposed project;				reported	
	\checkmark Seek to establish amicable					
	relationships with stakeholders					GRM Register
	and manage the impact of the				Number of	
	project activities on affected				cases	
	communities;				auuresseu	
	\checkmark Address all raised grievances,					
	real or imagined and take					
	reasonable steps to maintain					
	confidentiality of the parties to					
	the mechanism and regardless					

Environmental and social	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
Issue	of the complainants' participation in this process, give a guarantee that the complainant's statutory rights to undertake legal proceedings remain unaffected;					
Child abuse	 ✓ The Proponent will develop and implement a Children Protection Strategy that will ensure minors are protected against negative impacts associated with the project. ✓ All staff must sign, committing themselves towards protecting children, a contract which clearly defines what is and is not acceptable behaviour. ✓ Children under the age of 18 years will not be hired at the site as provided by Child Rights Act (Amendment Bill) 2014. 	-	BMU	Continous	Number of reported incidences of child abuse One Child Protection strategy in place	Reports
Risk of increased spread of COVID- 19	 ✓ The project shall put in place means to support rapid testing of suspected workers for COVID-19. ✓ Avoid concentrating of more than 15 persons or workers at one 	0.2	BMU	Continuous	Number of people in the Omena facility plant at any one time	Reports

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
	location. Where more than one					
	person are gathered, maintain social				Number	
	distancing at least 2 meters. All				Reported cases	
	workers and visitors accessing				of high	
	worksites every day or attending				temperatures	
	meetings shall be subjected to rapid				among	
	COVID-19 screening which may				workers	
	include temperature check and					
	other vital signs				Amount in	
	\checkmark Install handwashing facilities with				Amount m	
	adequate running water and soan or				running water	
	sanitizing facilities at entrance to				and soan	
	work sites including consultation				und soup	
	venues and meetings and ensure					
	they are used					
	Enguna nouting conitization of					
	Ensure routine samuzation of					
	snared social facilities and other					
	communal places routinely including					
-	wiping of workstations, door knobs.			~ .		-
Impacts related to	✓ Keep all passages clear at all	0.15	BMU	Continuous	Number of	Reports
occupational and	times.				fully equipped	
safety and health	✓ BMU to adopt and implement				IIIST and KITS	
Safety and fication	Good management Practice (GMP) and				Number of	
	Hazard Analysis and Critical Control				DDE _c	
	Points (HACCP) food safety programs				11123	
	• Adopt proper working					
	procedures and when working with					
	chemicals, machines and equipment.					

Environmental and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social						
Issue						
	\checkmark Clean all spilt hazardous					
	materials using an appropriate					
	disinfectant.					
	\checkmark Ensure that trained first aid					
	personnel are always available on site					
	to handle emergencies.					
	✓ Have a fully equipped First Aid					
	Kit (containing a first aid manual and is					
	equipped with sterile adhesive					
	bandages, safety pins, cleansing					
	agent/soap, latex gloves: sterile gauze					
	pads triangular bandages, non-					
	prescription drugs, scissors, tweezers					
	and antiseptic amongst others) at the					
	site at all times					
	\checkmark Conduct regular maintenance of					
	the proposed site and facilities thereat					
	to increase the life of the proposed					
	building making it safe for habitation					
	\checkmark Provide appropriate PPF					
	including face masks goggles scarfs					
	hoots and overalls among other					
	protective clothing to all workers and					
	protective clothing to an workers and					
	people at the site and sensitize them to					
	Ensure high standards of construction					
	Ensure high standards of construction					
	as recommended in the approved					
	structural designs and regular					

Environmental	and	Specific Mitigation plan	Cost (Kes)	Responsibility	T/Frame	Indicators	Sources of data
social							
Issue							
		maintenance to increase the life of the					
		structures at the site.					

Total is KES 1.03M

Decommissioning phas	Se la	
Solid wastes	 Use of an integrated solid waste management system (recycling, reuse, combustion and sanitary land filling) Remove from site and/or recycle/re-use at and/or away from site all machinery, equipment, structures and partitions that will not have been used up Ensure source separation and collection of wastes into recyclable and non-recyclable wastes by installation of double waste collection bins at each collection point Hire licensed waste handlers to regularly collect and dispose-off the wastes 	Contractor
Gender based violence and Sexual harassment (GBV/SH)	Make use of the project policy against sexual harassment, Code of Conduct and grievance redress mechanism	Contractor
Risk of spread of Covid 19	 Where one-on-one engagements for the PAPs are needed, observe social distance and adhere to PPE (masks) wearing Avoid concentrating of more than 15 community members at one location and where more than one person are gathered, maintain social distancing of at least 2 meters 	Contractor,
Risk of increased HIV/AIDS and STIs	 Sensitize workers and community members on HIV/AIDS and STIs, and create awareness on other communicable diseases as part of the Contractor's Health and Safety Management Plan to be enforced by the Supervising Engineer Enforce controlled access to Contractor's Workforce Camps by outsiders 	Contractor

	• Provide standard quality condoms at the site during the decommissioning period		
Sexual Exploitation and Abuse (SEA) by project workers against community members	• Develop and implement a SEA Action Plan with an Accountability and Response Framework as part of the decommissioning ESMP in accordance with the World Bank's Good Practice Note for Addressing Gender-based Violence in Investment Project Financing involving Major Civil Works (Sept 2018)	Contra	ctor
Incidents and accidents	 Report any incidents and accidents using prescribed forms obtainable from the Occupational Health and Safety Office Design suitable documented emergency preparedness and evacuation procedures for emergencies Have on site stocked First Aid Kits which are easily available and accessible Have on-site persons trained in first aid and issued with a certificate by a recognized body Prepare a contingency plan for emergency response Maintain an accident/incident register at the site 	Contra	ctor
Sub-total for decomm	issioning phase		191,000

Total cost of ESMP implantation	1,221,000
Total cost of ESIVIE implantation	1,221,000

CHAPTER NINE

9.0 CONCLUSION AND RECOMMENDATIONS

9.1 Conclusion

This study has ascertained that the construction of the proposed Kamariga fish drying facility will have both negative and positive impacts on the physical and the surrounding human environment. Positive impacts include resilient livelihoods, increased income, employment during construction and operation phases, improved sanitation, as well as catchment conservation. Negative impacts include increased incidences of water pollution, noise and dust pollution during construction, removal of vegetation to create space for drier and associated facilities, risks of occupational hazards and risk of soil erosion in all phases. The project design has integrated measures to mitigate some of the adverse impacts with a view to ensuring compliance with applicable laws and procedures. Additional and more detailed measures are provided in the ESMP and will help in mitigating the impacts. Overall, the ESIA study concludes that excavation and construction of a water pan will not generate significant negative and irreversible impacts that can compromise the ecological and environmental wellbeing of the area as well as health and safety of the residents. It is thus recommended that on submission of this report to NEMA a conditional approval for the proposed sub Project activities is granted through issuing an EIA license.

9.2 Recommendations

It is recommended that the proposed project proponent be allowed to go ahead provided the outlined mitigation measures are implemented to as outlined in the ESMP. The ESMP should be translated into Contractor-Specific Environmental and Social Management Plan (CESMP) and shared with the contractor who wins the subproject bid. The CESMP is binding on the contractor. Accordingly, the contractor is required to engage a qualified Environmental and Social Safeguards specialist as well as food Safety and Health consultant to oversee implementation of the satisfactory implementation HACCP and GMP food systems the ESMP. On approval, it is recommended that the proponent should implement the proposed project based on the proposed plans and if alterations are necessary, advice should be sought from the CESSCO and subsequently environmental expert.

- ✓ The proponent should share the ESMP with the Contractor and other responsible stakeholders and that the ESMP form part and parcel of the Contractor's contract to ensure that their obligations as outlined in the ESMP are executed
- ✓ The proponent will be required to undertake annual environmental and social audit pursuant to the provisions of the Act and World Bank ESS guidelines, as well food safety audits

In case of future closure of the fish drying facility, key stakeholders and the community should be involved in the planning and execution of the closure as to ensure that direct and indirect users are well prepared and able to adjust to the decommissioning, all environmental and socioeconomic impacts arising are addressed, alternative sources of livelihoods for the community are created and the land is reclaimed and restored to achieve its initial ecological and ecosystem functions.

9.2.1 Compliance with the Water Quality Regulations

The proponent as a registered BMU has complied with the standards set out in the fisheries Management and Development Act No.35 of 2016 by applying a licence for the establishment of a fish processing plant. The contractor will commence construction upon written authorization to go ahead with the construction works as the process for processing the permit proceeds. The proponent must ensure compliance with NEMA Environmental Management and Co-ordination (Water Quality) Regulations, 2006; the Water Act (No. 43 of 2016), and other related regulations. Regular testing of the water should be carried out (every 3 months or as may be agreed upon by the relevant authority) for the relevant parameters and mitigatory

action thereof. Food safety audits and testing of workers in compliance with food safety regulations will also be undertaken regularly.

9.2.3 Emergency Response Plan (ERP)

During the project construction, commissioning, operation and decommissioning, sustainable environmental management practices and adherence to stipulated structural designs and regulations will be observed. The proponent is committed to working closely with NEMA, environmental experts and relevant government agencies in adherence and implementation of the ESMP. In particular the proponent will institute regular monitoring of the work space particularly the machinery room, electrical wiring and testing the machinery. This is a risk reduction measure. The proponent should train the project management committee on detecting and responding to any risk situation and be part of the ERP team together with the local administration. All relevant departments should be alerted early enough for proper measures to be taken. The proponent should establish proper channel of information and risk communication.

References

- FAO, "Agro-Ecological zoning guidelines: FAO Soils Bulletin No. 73." Land and Water Development Division, FAO, Rome, Italy, 1996.
- Jaetzold, R., Schmidt, H., Hornetz, J., and Shisanya, C. (2007).Farm Management Handbook Of Kenya Vol. II: Natural conditions and Farm Management Information, 2nd Ed., Part A, West Kenya. Subpart A1, Western Province. Kenya.Ministry of Agriculture/GTZ (Germany Technical Agency. Nairobi, Kenya

G.o.K, "Agricultural Sector Transformation and Growth Strategy 2019-2029," Nairobi, Kenya,2019.

- G.o.K, "Climate Change Action Plan 2018-2022," Nairobi, Kenya 2018.
- Kenya gazette supplement No. 56. Environmental impact assessment and auditregulations 2003. Government printer-Nairobi
- UN, "Transforming the world: the 2030 Agenda for Sustainable Development," New York, A/RES/70/1, 2015

Kenya gazette supplement Acts 2000, EMCA 1999, Government printer- Nairobi Kenya gazette supplement Acts land planning Act. Government printer-Nairobi

Kenya gazette supplement Acts fand planning Act. Government printer-Narool

United Nations Environment Programme Division of Technology, Industry and Economics Cleaner Production Assessment in Fish Processing

- United Nations Environment Programme (UNEP), 2012, Avoiding future famines. Strengthening the ecological foundation of food security through sustainable food systems, UNEP, Nairobi, Kenya
- Republic of Kenya (2006) The Fisheries Management and Development Act NO.35. Government Printer, Nairobi
- Republic of Kenya (1999) The Environmental Management and Co-ordination Act, 1999 (EMCA) . Government Printer, Nairobi
- Republic of Kenya (2003) EMCA requirements for Environmental Assessment and Audits. Government Printer, Nairobi
- Republic of Kenya (2006) Environment management and coordination (waste management) regulations 2006. Government Printer, Nairobi
- Republic of Kenya (2006) Environment management and coordination, conservation of Biological Diversity (BD) Regulations 2006. Government Printer, Nairobi
- Republic of Kenya (2006) Environment management and coordination (Fossil fuel emission control) Regulations 2006. Government Printer, Nairobi

Republic of Kenya (2007) Environment management and coordination (controlled substances) Regulations 2007. Government Printer, Nairobi

Republic of Kenya (2009) Environment management and coordination (Noise and excessive vibration pollution) control Regulations 2009. Government Printer, Nairobi

- Republic of Kenya (2007) the occupational safety and health Act, 2007. Government Printer, Nairobi
- Republic of Kenya (2007) Fire Risk Reduction rules (2007). Government Printer, Nairobi
- Republic of Kenya (2007) Hazardous substances rules (2007). Government Printer, Nairobi
- Republic of Kenya (2004) Health and Safety Committee Rules, May 2004. Government Printer, Nairobi

Republic of Kenya The Public Health Act. Government Printer, Nairobi

APPENDIXES

Appendix I: Environmental Social Management Framework Checklist and Appraisal

Annex 12A: Environmental and Social screening Cheek list ESM Sub-projects Screening Checklist (Prototype) (Sub-projects screening process by benefitting communities/Agencies) Section A: Background information Name of County SLATA Name of CPCU/Researcher FAITtl CHERONDH Sub-project location KAMARI G.A. Name of CBO/Institution KAMARICA. BEACH, MANAGEMENT UNIT Postal Address 110-40601 BONDO Contact Person WALTER JURA Cell phime 0121704-205 Sub-project name KAMARIGA BART Omona bry Approximate size of land area available for the sub-project. 1 H9 Objectives of the sub project. TREE NURSER-HOR IICULTURAL PRODUC 110 Activities/enterprises undertaken Fish diging How was the sub-project chosen?... Commun Expected sub project duration. ONE TONE Section B: Environmental Issues Will the sub-project: Create a risk of increased soil erosion? Yes No Create a risk of increased deforestation? Create a risk of increasing any other soil degradation soil degradation? 4 L Affect soil solinity and alkalinity? 2 Divert the water resource from its natural course/location? L Cause pollution of aquatic ecosystems by sedimentation and agrochemicals, oil spillage, effluents, etc.? introduce exotic plants or animals Involve drainage of wetlands or other permanently flooded areas? Cause poor water drainage and increase the risk of water-related ω disenses such as malaria? Reduce the quantity of water for the downstream users? Result in the lowering of groundwiner level or depletion of 6 groundwater? Create waste that could adversely affect local tails, vegetation, rivers Reduce various types of livestock production? Affect any watershed?

Will the sub-project:	Yes	No
Displace people from their current settlement?	1	V
interfere with the normal health and safety of the worker/employee?	V	1
Reduce the employment opportunities for the surrounding communities?		V
Reduce settlement (no further area allocated to settlements)?	1	V
Reduce income for the local communities?	-	1
increase insecurity due to introduction of the project?	-	17
Increase exposure of the community to HIV/AIDS?	17	V
Induce conflict?	-	1
Have machinery and/or equipment installed for value addition?	1	V
Introduce new practices and habits?	1	-
Lead to child delinquency (school drop-outs, child share, shild labour	1	
cte?		1
Lead to gender disparity?	-	-
l ead to poor diets?		1
Lead to social evils (drug shure) accessing deuted	-	V
etc.)?	1	

Section D: Natural Habitats Will the sub-project: YES NO He located within or near environmentally sensitive areas (e.g. intact natural forests, mangroves, wetlands) or threatened species? Adversely affect environmentally sensitive areas or critical habitats wetlands, woodlots, natural forests, rivers, etc.)? Affect the indigenous biodiversity (Flora and fauna)? Cause any loss or degradation of any natural habitats, either directly (through project works) or indirectly? Affect the aesthetic quality of the landscape? Reduce people's access to the pasture, water, public services or other 12 resources that they depend on? Increase human-wildlife conflicts? V Agrochemical use Will the sub-project: Involve the use of pesticides or other agricultural chemicals, or increase existing use? Cause contamination of watercourses by chemicals and pesticides? Cause contamination of soil by agrochemicals and pesticides? Experience effluent and/or emissions discharge? Export produce? Involve annual inspections of the producers and ununnounced inspections? Require scheduled chemical applications? Require chemical application even to areas distant away from the focus? Require chemical application to be done by vulnerable group (regnant mothers, chemically allergic persons, elderly, etc.)?

Use irrigation system in its implementation?

6. Training			
a second se	Innine	substant a	in more
 a) Have you ever received any training on any of use following herei-no 	; topics	tenned t	o crot
Internated Pest Manusement Yes			
No. of times /past year.			
b).Pesticide Usage Yes No			
No. of times /past year			
c).Pesticide Safety: Yes No			
No, of times /pust year.			
d).Insect Identification Yes No			
No.of times /past year.			
e).Disease Identification Yes No			
No. of times /pass year.			
1).Quality aspects of production res			
No.01 unies plat year			
7) Is there anything else that you want us to know about your o	crop pr	oduction	?
N/At r			
If the answer to the above is 'yes', please consult the IPM	that ha	s been pi	repare
If the answer to the above is "yes", please consult the IPM for the project.	that ha	s been pi	repare
If the answer to the above is "yes", please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet	that ha	s been pi quireme	repare nts fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10	that ha	s been pi quireme	repart
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginatized Groups meet OP 4.10 re there:	that has ting re Yes	s been pi quireme NO	repart ints fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries	that has	s been pi quireme NO	nts fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project?	that has ting re Yes	s been pr quireme NO	nts fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries f, or near the project? tembers of these VMGs in the area who could benefit from the project?	that has ting re Yes	s been pr quireme NO	nts fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? iembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project?	that has ting re Yes	s been pr quireme NO	nts fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Combers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the	that has fing re Yes VMGI	s been pr quireme NO	ats fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project.	that has ting re Yes V VMGI	s been pr quireme NO	ents fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project. Section G: Land Acquisition and Access to Resources	that has ting re Yes V VMGI	s been pr quireme NO	epare ents fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project. Section G: Land Acquisition and Access to Resources Vill the sub-project:	that has ting re Yes VMGI	s been pr quireme NO P that ho s No	ents fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project. Section G: Land Acquisition and Access to Resources Vill the sub-project: cquire that land (public or private) be acquired (temporarily conserved) for its Amediacource?	that has fing re Yes VMGI	s been pr quireme NO S No S No	ents fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project. Section G: Land Acquisition and Access to Resources Vill the sub-project: course that land (public or private) be acquired (temporarily emanently) for its development?	that has ting re- Yes VMGI or Ye	s been pr quireme NO V s No S No V	mts fo
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project. Section G: Land Acquisition and Access to Resources Vill the sub-project: equire that land (public or private) be acquired (temporarily emanently) for its development? be land that is currently occupied or regularly used for producti memory for an area the project for producti	that has ting re Yes VMGI or ive	s been pr quireme NO S No S No	epare
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project. Section G: Land Acquisition and Access to Resources Will the sub-project: require that land (public or private) be acquired (temporarily emanently) for its development? Ise land that is currently occupied or regularly used for producti arposes (e.g. gardening, farming, pasture, fishing locations, forests)	that has ting re Yes VMGI vMGI ive	s been pr quireme NO S No S No S No	as bei
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project. Section G: Land Acquisition and Access to Resources Vill the sub-project: bequire that land (public or private) be acquired (temporarily ermanently) for its development? be land that is currently occupied or regularly used for producti a rposes (e.g. gardening, farming, pasture, fishing locations, forests) Displace individuals, families or businesses?	that has ting re Yes V VMGI ve ive	s been pr quireme NO S No S No S No S No V	epare
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: sople who meet requirements for OP 4.10 living within the boundaries (; or near the project? tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project Section G: Land Acquisition and Access to Resources Will the sub-project: tequire that land (public or private) be acquired (temporarily emanently) for its development? be land that is currently occupied or regularly used for producti arposes (e.g. gardening, farming, pasture, fishing locations, forests) Displace individuals, families or businesses? Cesult in temporary or permanent loss of crops, fruit trees and pastu and?	that has ting re- Yes VMGI VMGI ive ure	s been pr quireme NO V V that h s No V V V V V	epare
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project. Section G: Land Acquisition and Access to Resources Will the sub-project: Equire that land (public or private) be acquired (temporarily emanently) for its development? Ise land that is currently occupied or regularly used for producti criposes (e.g. gardening, farming, pasture, fishing locations, forests) Displace individuals, families or businesses? Result in temporary or permanent loss of crops, fruit trees and pasta and? Miversely affect small communal cultural property such as faneral a wird sites, or sacred proves?	that has ting re Yes VMGI ve ure und	s been pr quireme NO S No S No S No V V V V V	repare
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project. Section G: Land Acquisition and Access to Resources Vill the sub-project: Equire that land (public or private) be acquired (temporarily emanently) for its development? Is land that is currently occupied or regularly used for producti projects (e.g. gardening, farming, pasture, fishing locations, forests) Displace individuals, families or businesses? Result in temporary or permanent loss of crops, fruit trees and pastured? Meresely affect small communal cultural property such as funeral a urial sites, or sacred groves?	that has ting re Yes VMGI VMGI ive ure und ted	s been pr quireme NO S No S No S No V V V V V V V	repare
If the answer to the above is 'yes', please consult the IPM for the project. Section F: Vulnerable and Marginalized Groups meet OP 4.10 re there: cople who meet requirements for OP 4.10 living within the boundaries t, or near the project? Tembers of these VMGs in the area who could benefit from the project? MGs livelihoods to be affected by the sub project? If the answer to any of the above is 'yes', please consult the prepared for the project. Section G: Land Acquisition and Access to Resources Vill the sub-project require that land (public or private) be acquired (temporarily emanently) for its development? Is land that is currently occupied or regularly used for producti projects (e.g. gardening, farming, pasture, fishing locations, forests) Displace individuals, families or businesses? Result in temporary or permanent loss of crops, fruit trees and pasta and? Moversely affect small communal cultural property such as faneral a urial sites, or sacred groves? Result in involuntary restriction of access by people to legally designar as ks and protected areas?	that has ting re Yes VMGI VMGI ive ure und ted	s been pr quireme NO S No S No S No V V V V V V V V V V V V V V V V	repare

If the answer to any of the above is 'yes', please consult the mitigation measures in the ESMF, and if needed prepare a (Resettlement Action Plan) RAP.

(i) Summarize the above:	(ii) Guidance
All the above answers are 'No' There is at least one 'Yes'	 If all the above answers are 'No', there is no need for further action;
	 If there is at least one 'Yes', please describe your recommended course of action (see below)

(iii) Recommended Course of Action

If there is at least one 'Yes', which course of action do you recommend?

CPCUs and County Director of Environment (CDE) will provide detailed guidance on mitigation measures as outlined in the ESMF; and

EAspecific advice is required from CDE and CPCUs regarding sub-project specific EIA(s) and also in the following area(s)

All sub-project applications/proposals MUST include a completed ESMF checklist. The KCSAP-CPCU and CDE will review the sub-project applications/proposals and the CDEs will sign off;

The proposals will then be submitted to NPCU for clearance for implementation by communities in the proposed subprojects.

Expert Advice

The National Government through the Department of Monuments and Sites of the National Museums of Kenya can assist in identifying and, mapping of monuments and archaeological sites; and Sub-project specific ESIAs, if recommended, must be carried out by expense

L Sub-project specific ESIAs, if recompended, must be carried out by experisi registered with NEMA and be followed by monitoring and review. During the process of conducting an EIA the proponent shall seek views of persons who may be affected by the sub-project. The WB policy set out in OP 4.01 requires consultation of sub-project affected groups and disclosure of EIA's conclusions. In seeking views of the public after the approval of the sub-project, the proponent shall avail the draft ESIA report at a public place accessible to project-affected groups and local NGOs/CSOs.

Date:	20/02/20	20 11	1.15			
Field Apprais: Slavature:	il Officer (CDE	20	JILLIA	MC	DE-10	1.14
Signature:	UA	22-	D		an no pa	air
Date:	20/02	2020		201	20 1029	147

Appendix II : Assessment Checklist and Findings

	Permits	Standar ds	Len	Length of impact			Effect on Environment		Effect on Environment		Mitigation Required
Assessment factors			Т	Р	None	MA	SA	Beneficial			
Regulatory Factors											
1.Air Pollution Control (including CFCs			X	-		X	-	X	Maintenance of machinery		
Drinking Water Management			X	-		X	-		Chlorination		
Water Pollution			X			X			Water quality and effluent management		
Hazardous Waste Management			X	-		-			Effluent management system		
Solid Waste Management			X	-		X	-		Recycling and waste minimisation		
PCB Management			X	-		X	-				
Radioactive Materials Management			-			-					
Environmental factors											
Natural Factors 1. Fish and Wildlife 2. Vegetation			- X			X -			Habitat restoration		
 Endangered Species Water and Hydrology 			- X			X X					

5. Air and Noise	X		X		
6. Physiography	х		-		
7. Soils and Erosion	х		-		
8.Historical,	-				
Archaeological,	-		-		
Paleontological	-		-		
Resources	-		-		
9. Prime Farmlands	-		-		
10. Wetlands					
11. Floodplains	-		-		
12. Wild and Scenic					
Rivers	-		-		
15. National Wilderness	-		-		
B. Human Factors					
1. Demography	-		-		
2. Housing	-		-		
3. Utilities	-		-		
4. Fire hazards	Х		Х		
5. Social Services	Х		Х		
6. Recreation and	Х		Х		
Aesthetics	х				
7. Land Use	х		Х		
8.Traffic and	-			Х	
Transportation					
9. Quality of Life					
C.Socio-economic					Code of practice and policies on GBV, SEA and
Factors					Child protection
1.Residential Dwellings					
2. Local Employment	-			Х	
3. Public Health and				Х	
Well-Being					

4. Relocation of Public	-				
Utilities			х		
5.Traffic and Congestion	-				
6. Safety	Х				
7. Effect on Population	-				
Trends					
8 Adverse Community	-		-		
Reaction to the Project	Х				
9. GBV	Х				
10.SH	Х				
11. Child Labour					

MA = Moderately adverse; SA Significantly adverse; () Temporal; (P) Permanent; (-) absence of the issue (benefit), (x) presence of the issue

Appendix III: Practising Licences

FORM 7



NATIONAL ENVIRONMENT MANAGEMENT AUTHORITY(NEMA) THE ENVIRONMENTAL MANAGEMENT AND CO-ORDINATION ACT

ENVIRONMENTAL IMPACT ASSESSMENT/AUDIT (EIA/EA) PRACTICING LICENSE License No : NEMA/EIA/ERPL/11751

Application Reference No:

NEMA/EIA/EL/15517

(r.15(2))

M/S DR. VOLENZO TOM ELIJAH (individual or firm) of address

P.O. Box 41-30105, SOY

is licensed to practice in the

capacity of a (Lead Expert/Associate Expert/Firm of Experts) Lead Expert registration number 7332

in accordance with the provision of the Environmental Management and Coordination Act Cap 387.

Issued Date: 2/13/2020	Expiry Date	12/31/2020
(Signature
		the (Real)
	The Natio	Director General onal Environment Management Authority
	1. 2. 2. 2.	
11.11.14	P.T.O.	
and a start of the	ISO 9001: 2008 Certified	



Appendix IV : Ground Layout Of The Proposed Kamariga Fish Drying



Appendix V: Evidence of landing ownership title deed



Appendix VI: Record of Issues (FGD) Meetings

Agenda Prayer Opening Remarks Introductory remarks Participatory Assessment in ESIA Filling of questionnaires Community reaction and approval Closing remarks and way forward

The meeting was opened with a word of prayer and self-introduction. The **CESSCO** gave introductory remarks mostly touching on the KCSAP Project and the history of the subproject. The Agricultural officer in charge of the ward also sensitized on the Agricultural component of the project. The CESSCO then invited the NEMA registered expert to give a few remarks on the ESIA process, in particular the importance of public participation. Filling questionnaires

Prepared questionnaires were introduced with the Nema Expert going in through all the items on the questionnaire to clarify all the issues. Translation of the questionnaire items into DHOLUO language (Though majority of the attendees could read in English). The questionnaire was distributed and members given the opportunity to fill them. The respondents were given leeway to consult whenever issues were not clear.

Community reaction

- ✓ The need for catchment conservation and other agricultural productivity projects
- ✓ Sex for fish practices among of the fishermen
- \checkmark Managing the lake shore erosion
- ✓ HIV/AIDS menace

The CESSCO and the Project engineer gave satisfactory responses to all the raised matters as follows;

- ✓ Catchment conservation shall be taken care through another Project component on natural resource management and that the BMU should come up with proposals on the same
- ✓ Sensitisation on HIV/AIDS is integrated into the implementation framework
- ✓ Sex for fish is considered as sexual exploitation World Bank takes it as serious social issue. The BMU should come with code of conduct for its members and explicitly have a policy of zero tolerance on sexual exploitation
- ✓ Community members would practice integrated Pest Management and Agroecology principles to reduce surface contamination
- ✓ Community would be encouraged to build latrines in homesteads

On conservation and income generating activities, the CPU will assess any submitted proposal and fund in line with the Project funding guidelines but in the meantime capacity building on pan management committees, Environmental social safeguard committees and social accountability integrity committees will be undertaken

Voting and proposal approval

The community unanimously voted for the project asking KCSAP to expedite the process as it would greatly alleviate the water scarcity in the area

Closing remarks

The area chief was invited to make a comment. In his remark, he welcomed the initiative and pledged his support for the implementation process. The meeting adjourned with a word of prayer from one of the BMU members.

Kenya Climate Smart Agriculture Project (KCSAP) Office of the CPCU - Siaya Defice of the CPCU - Siaya NMME Recent and	1552	ACTIVITY:	NO	+ D	2	ун 5	*	5 WA	s. Mo	N N	1	9 24	10,	
Kenya Climate Smart Agriculture Project (KCSAP) Office of the CPCU - Siaya Registration rom Marticulture Project (KCSAP) Nome Registration rom Context Nome Context Nome Context Nome Context Nome Context Nome None <th< td=""><td></td><td>Public</td><td></td><td>AVID</td><td>3 Nos</td><td>NWA-</td><td>HARLES</td><td>ubn (</td><td>(Lutri)</td><td>hander</td><td>ostura</td><td>ICLIPE</td><td>HILES</td></th<>		Public		AVID	3 Nos	NWA-	HARLES	ubn ((Lutri)	hander	ostura	ICLIPE	HILES	
Climate Smart Agriculture Project (KCSAP) Office of the CPCU - Siaya NEGISTRATION FORM I de mé na ng de de Maile registration Intel de mé na ng de de Intel de mé ng de de Intel de mé na ng de de Intel de mé ng de de Intel de mé ng de de Intel de mé ng de de de Intel de mé ng de de de de Intel de mé ng de	Kenya (Participation	NAME	AVM USI CBENNI	Marko	IN DOMISHIA	ONGACY	Tich ago wint	Ohneria Talance	BILITY ALOW	Ouma	OFFICE NYMAND	00000	
Intron FORM CPCU - Siaya OREANIZATION DESIG Contest DATE DATE <th co<="" td=""><td>Office of</td><td>- Hama</td><td>Gender (Male/Female)</td><td>male</td><td>THM</td><td>MALE</td><td>VALE</td><td>MARG</td><td>MARE</td><td>Mak-</td><td>male .</td><td>MALE</td><td>Fernance</td></th>	<td>Office of</td> <td>- Hama</td> <td>Gender (Male/Female)</td> <td>male</td> <td>THM</td> <td>MALE</td> <td>VALE</td> <td>MARG</td> <td>MARE</td> <td>Mak-</td> <td>male .</td> <td>MALE</td> <td>Fernance</td>	Office of	- Hama	Gender (Male/Female)	male	THM	MALE	VALE	MARG	MARE	Mak-	male .	MALE	Fernance
DATE Siaya Siaya DATE DATE III III III III IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	art Agricult the CPCU -	20 C	ORGANIZATION	Benest	1 . W. W.	13 mil	BANN	B.mu-CALAIR	Bmu vie char	b.m.	Bm	B-MI-U	(Control	
ect (KCSAP) Contact Mobilie number 07:5528511 07:5528511 07:15528511 07:25:28:2719 07:25:427:49 07:25:427:49 07:25:427:49 07:25:427:49 07:25:427:49	ure Proj Siaya		DESIG	11/ 11	4. 5	N 82	MENALR	1	Office	themas	PF-	Contraction of the	- Marches	
	ect (KCSAP)	DATE	Contact Mobile number	07:552851	PITERSTRED	136L90 MILO	0720402705	506402HL9	0723 018 445	0734727294	octop was	PERT 2HS STO	010135140	

Appendix VII: Evidence of Public Consultation (Attendance and Sample Questionnaire)
10,		90	2	9	57	2	39	2	+	NO	ACTIVIT	- on Sam
LEMAN ALLOTH ODOI	LEAMERY OTION	JANG OGUIL	GENTA ATHRO	Emiles ONTANGS	DURAN OTIGNO	MART ALIANISO	LEDNARD OPUL NO	MILLIAM OKED ADD	ALICE ABHIAMED 06	NAME	M2	K Republic of Kenye
C31	Arine C.						nue	(mmdis	e-bashing			enya C
7	m	P	D	D	Г	٦	m	m	D	Gender (Male/Female)	REGISTRA	limate Sm Office of
11	11	11	n	11	10	11	ULENGA PA	UNINE PAN	M. PAN	ORGANIZATION	TION FORM	art Agricult the CPCU -
н	11	200	iii	11	NEWSO	maniboli	A. Farvey	Mamida	memlo	DESIG		ure Pro Slava
07142073 6h	0729998112		0708753765	07175465M	× 11866640 7	856412 0 540	2/07368370/2	Equility 0-4	H25/8 abtop	Contact Mobile number	DATE	ject (KCSAP)
Ş	Aug-	XX	R	ð	Real	mary.	Bulltoner	AD	7	SIGN		Konya Chan

10.	9	ġ#	7.	, PA		* V	- LE	" Vu	00	ACTIVITY:	1.52
			amond Agaya	TRICE OFICIA	35rey Odany	Olenza Tom	MIGHER ACOCA	wert Mills Othe	NAME		Kenya Is d Konya
			1 M	M	3	00	n	И	Gender (Male/Female)	REGISTR	a Climate Sm Office of
			trant	N/cener	Detruit	1. Shund	MONLE	CGIS	ORGANIZATION	ATION FORM	hart Agricult
			CETTO	Alcher	Nees	El AVE	MUNIC DEFECT	WARD ADMINISTOR	DESIG		ture Proj - Slaya
			paraferro	0724969612	HOMENHE CO	Astroche 40	0727615178	69LANERLO 20	Contact Mobile number	DATE	ect (KCSAP)
			ATT.		t	W	ale	A	SIGN		Name Units a

	10,	9	09	7.	9	7	4	ļ.	2	٢	NO	ACTIVIT	- 000
				*		ARIHUR OSIMBO	MUHAEL W. ANIVLLA	BETHWARD OGUSO HITAMILA	the same shows arke	UPONARD DEEWE Chiems	NAME		Kenya Kenya
		e)				m	3	151	m	N	Gender (Male/Female)	REGISTRA	Office of
						MCP	ADDIN		11	WHER NGA	ORGANIZATION	TION FORM	art Agricult the CPCU -
						orthic ere	17552	h	MEMRE	MEMBER	DESIG		ure Proj∈ Siaγa
						LZINTSLID	S889484044		CCLINERIA C	6666KERTCHO	Contact Mobile number	DATE	ect (KCSAP)
1						All of the	-ak	p	Alama	13	SIGN		- Keepa tiim



.

.

Project name Article Article Article Article Location Theory Heat Article Arti	1.12	Kanmines (Dansile	Sin	bourt	
Virage Control of the sub-sub-sub-sub-sub-sub-sub-sub-sub-sub-	Project nam	DUA 51 VO	L. MILINS	Kaapt	Company With	
Network Separture Barbon of the second of the se	Village.	Scrat Odha	ud location.	Th has in	20659110	
Date 26/14m Render 164/24mm Alee 46/14m Alee 1 Alee 1 Alee 1 III. New copins what are the likely impacts of the project on the community and the environment (Yes/No) Impact category Positive Regative 1 Impact category Positive Regative Comment 1 Impact category Impact category <td< td=""><td>TEL</td><td></td><td>E.</td><td>mail</td><td>tellismbile parts (Spheril Cane</td><td></td></td<>	TEL		E.	mail	tellismbile parts (Spheril Cane	
Age As Gender Horizon In how long have you lived in this place? Horizon Horizon III) Any our opinion what are the likely impacts of the project on the community and the environment Impact category Positive Regative Comment III) Any our opinion what are the likely impacts of the project on the community and the environment Impact category Positive Regative Comment III) Any our opinion what are the likely impacts of the project on the community and the environment Impact category Positive Regative Comment III our opinion what are the likely impacts of the project on the community and the environment Impact category Impact category Impact category III our opinion Asthetic/partial Impact category Impact category Impact category Social HIV/MDS AND STIS Impact category Impact category Impact category Social HIV/MDS AND STIS Impact category Impact category Impact category Social HIV/MDS AND STIS Impact category Impact category Impact category Social HIV/MDS AND STIS Impact category Impact category Impact category Social <	Date	30/JA- 2	220	. 5	ignature 1811 Source	
II. Water you award this place? 12 III. Water you award this project before? 12 III. Water you award this project before? 12 III. Biodiversity 1 1 III. Biodiversity	Age	4.5		er		
III. Were you awars of this project before? III. Impact Category Positive Negative Comment Impact Category Positive Negative Comment I. Biodiversity III. Negative Comment I. Biodiversity III. III. Negative Comment I. Biodiversity III. III. III. III. Social III. III. III. III. III. III. Social III. III	I). How long	g have you lived in this	place?			
Minute certain are used in the project of the proj	iii) te rei yo	waware of this project	kely impacts	of the proje	ect on the community and the environment	
Impact Category Positive Negative Comment Interact Category Positive Negative Comment I. Biodiversity Enclose Enclose B. Second Enclose Enclose W. Water/Air pollution Interaction Interaction Social Interaction Interaction Interaction B. Cathure Interaction Interaction Interaction Social Interaction Interaction Interaction Interaction Social Interaction Interaction Interaction Interaction Social Interaction Interaction Interaction Interaction Biosciness and trade Interaction Interaction Interaction Interaction With your opinion, are there better alternatives to the project? Interaction Interaction Interaction Interaction Vith your opinion, can the project design be improved ? Yes? No No Interaction Interaction Interaction Interaction Interaction Interaction Interaction Interaction Interaction Interacti	(Yes/No)	opinion when one one is	wery in ignored	ar the prop		
Environmental + Biodiversity ii. Asthetic(c)natural beauty - - iii. Eresion - - iv. Water/Air polliution - - Social - - - - iii. Culture - - - - iii. Culture - - - - - iii. Gender relations -	Impact cata	ngary	Positive	Negative	Comment	
I. Biodiversity II. Aesthetics[natural basaty] II. Eration N. Water/Air pollution N. Culture N. Culture N. Culture N. Business and trade T. Business and trade N. Househald income Value of Land Total M. Househald income Voln your opinion, are there better alternatives to the project? Tationg Total Technology Total M. Househald income V.In your opinion, can the project design be improved ? Yes? No No. Based metrial V.In your opinion (ii to v), do you support project implementation? Yel/ No Will. Briefly give your comment on why you support project implementation? Yel/ No Will. Briefly give your comment on why you support for the project? Tather Mann Areabare <t< td=""><td>Environmen</td><td>rital</td><td>1</td><td></td><td></td><td>1</td></t<>	Environmen	rital	1			1
II. Aesthetics[instural beauty) III. III. Erosion IV. Water/Air pollution Social III. III. Gender relations	1.	Biodiversity	10			
Beautyl Image: Consumeration of the second of the seco	II.	Aesthetics(natural				
Image: Enclose Image: Enclose No. Water/Air pollution III Culture IIII Culture IIII Culture IIII Culture IIIII Culture IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII		beauty)		1		
Nr. Water/Air pollution 30cial HIV/AIDS AND STIS 1) Culture 10) Gender relations 11) Culture 12) Technology 13) Business and trade 14) Household income 17) Value of Land 17) Technology 18) Business and trade 19) No Technology Technology 17) Technology 18) Business and trade 19) Business and trade 10) Nour opinion, can the project design be improved ? Yes/ No 11) Bused on your opinion (iii to v), do you support project implementation? Yes/ No 11) Bused on your opinion (iii to v), do you support/Oppose the project? 11) Bused on your opinion (iii to v), do you support/Oppose the project?	18L	Erosion	10.5			1.1
Social III. HIV/AIDS AND STIS III. III. III. III. III. III. III. III	ity.	Water/Air pollution	-	-		-
ii) Culture The second se	Social	unitable tale on	705			1
Mill Gender relations Trial Bit Economic 191 Bit Business and trade 191 Mill Household income 191 VIn your opinion, are there better alternatives to the project? No Category Yes No Technology 191 191 VIn your opinion, can the project design be improved ? Yes? No No Mill. Brieffy explain your answer 115 Project Mill. Breef on your opinion (iii to VI. do you support project limplementation? Yes? No No Mill. Brieffy give your comment on why you support project limplementation? Yes? No No Mill. Brieffy give your comment on why you support/Oppose the project? Trace Project Mill. Brieffy give your comment on why you support/Oppose the project? State Project Mill. Brieffy give your comment on why you support/Oppose the project? State Project Mill. Brieffy give your comment on why you Project	11	HIV/AIDS AND STE	Her			
mini General Fieldon 1744 ii) Business and trade 1744 iii) Value of Land 1744 iv) Househald income 1744 iv) Repert the better alternatives to the project? Category 1745 Technology 1744 Yes No Technology 1744 Stilling 1744 Raw material 1744 V) In your opinion, can the project design be improved ? Yes/ No wi) Briefly explain your answer 154 The material 164 Wi) Based on your opinion (iii to v), do you support project implementation? Yes/ No wii). Briefly give your comment on why you support/Oppace the project? Still Will Briefly give your comment on why you support/Oppace the project? Taupport the implementation? Yes/ No will Briefly and the implementation? Still District The child the implementation? Still Distone Implementation? </td <td>104.9</td> <td>Gender relations</td> <td>TPS</td> <td></td> <td></td> <td></td>	104.9	Gender relations	TPS			
1) Employment 1711 11) Business and trade 1711 11) Household income 1715 Win your opinion, are there better alternatives to the project? Category Yes No Technology 1711 Alternatives to the project? Category Yes No Technology 1713 Alternatives to the project? No	Economic	server (Cabobia)		-		
ii) Business and trade Image: Contract of the second	B	Employment	122			
iii) Value of Land The Win your opinion, are there better alternatives to the project? Category Ves No Technology Tes No Sting Tes No Sting Tes No No Tes No No Tes No No No Tes No Tes No No No Tes No No No No No	ΠŠ	Business and trade	ry DA			
M Household income 105 Wiln your opinion, are there better alternatives to the project? Image: Category Ves No Technology 12.5 No Image: Category Ves No Technology 12.5 No Image: Category Ves No Sting 17.5 No Image: Category Ves No Sting 17.5 No Image: Category Ves No Sting 17.5 Image: Category Ves No Sting 17.5 Image: Category Ves No Sting 17.5 Image: Category Ves No V) In your opinion, can the project design be improved ? Yes/ No Ves Ves Ves VI) Based on your opinion (III to v), do you support project implementation? Yes/ No Ves Ves VII) Based on your opinion (III to v), do you support/Oppose the project? Tes Image: Project Sting 11.6 Made: Project Project Project VIII 11.6 Made: Project Project Project VIII 11.6 Made: Project Project Project VIII 11.6 11.6 11.6 Project VIIII <td< td=""><td>(11)</td><td>Value of Land</td><td>191</td><td></td><td></td><td>1</td></td<>	(11)	Value of Land	191			1
Wiln your opinion, are there better alternatives to the project? Category Ves No Technology 12.5 Raw material 12.1 No 12.1	IM)	Household income	105			
Raw material Test v) In your opinion, can the project design be improved ? Yes! No Improved ? Yes! No will, Briefly explain your answer Its bout 15 Shall improved that the prosess include the prosest include the prosess incline the proses include the pro	Category	IV	Yes	ES ID DIE P	No	
v) In your opinion, can the project design be improved ? Yes? No will Briefly explain your answer. The project will be send of the process higher the terminated the project design be improved ? Yes? No will Briefly give your opinion (iii to v), do you support project implementation? Yes? No will Briefly give your comment on why you support/Oppose the project? Taupport the Implementation of the door project Taupport the Implementation of the door project Taupport the Implementation of the door project Taupport the Implementation of the door project The active option that employee the project of the door project The active option that make the make project the the terminated of the Westlered Changes	Category Technolog Sitling	ly	Yes	es to die p Te s	No	
Changes.	Category Technolog Siting Raw mate v).In your vi).Brjefly	rial opinion, can the proj explain your answer	Yes	ES TES TES improved	Prez No Shall Impers living of	
	Category Technolog Siting Raw mate vi).Briefly vii).Briefly vii).Baset viii).Briefl Viii).Briefl Viii).Briefl Category The Category The Category	appinion, can the project of the pro	Yes ret design be This bis bis v), do you s an why you s Thyp(thyp CYP) bis Dis thyp(thyp) CYP)	ES TEJ TEJ MEJ Improved I Improved I Import SERA upport/Op Em IETES I Import/Op Em IETES I Improved I Improved	No Yes/No Stall Internet Internet of at Internet as the at Internet process pose the project? The Char project? And the Stalerty Units of make Augusti and the Stalerty Units of Make Augusti and the Stalerty	

I

CONTRACTOR OF A DESCRIPTION OF THE REAL PROPERTY OF	Sub location	A Condection WEST D'LANUS.	20.00
Name DO RST TIM DCA	ALE THE	10.00 21 882587	- 11
EL 0 11406797	Email	Northern Martines	
Jate 3 m m 1020	Gender	wave Statute	-
Here lane have you lived in th	ic niace?	SE THAT! LETAS	
I). Were you aware of this profe	ct before?	ES	
ii).In your opinion what are the	likely impacts of the	project on the community and the environme	ent.
(Yes/No)	a na ann an t-a an t-a f-a tais	Step As a prostant of the own the task	10.00
Impact category	Positive Nega	tive Comment	
Environmental	1		
i. Biodiversity	1	-	
E. Aesthetics(natural	1		
Deauty)			1.1
m. Erosion	V		1.0
w. water/Air posution			-
II HOVAIDS AND ETH	~		
II) Culture			
iii) Gender relations			
Economic			
i) Employment			-
ii) Business and trade			
iii) Value of Land			
(v) Household income			
Siting Raw material	1		1 1
naw maderial			-0
of Realization and the same sufficient	and the second of the	DAM Local Conserved	5-1
v() Briefly explain your answer.	to v), do you support t on why you support t on why you support	project implementation? Yes No t/Oppose the project? to the the total for the second to she the second for the second for she the second for	2 Aue45
v() Briefly explain your answer.	to v), do you support to v), do you support on why you support the	project implementation? Yes No t/Oppose the project? USD THE LO THE WIN CO STOR HE COMPANY Produceds	2 Aue45
v() Briefly explain your answer.	to v), do you support t on why you support t on why you support A Second	project implementation? Yes No VOppose the project? USB THE LATH BIN CO SIGN HEART	2 18: 14: 14: 14: 14: 14: 14: 14: 14: 14: 14
v() Briefly explain your answer Stuck House vii). Based on your opinion (iii) viii). Briefly give your comment CSS COMPANY CSS COM	to v), do you support t on why you support t on why you support	project implementation? Yes No VOppose the project? USB AT LOT AND AND AT A DECAST REALIST	ie ducts
v() Briefly explain your answer.	to v), do you support t on why you support t on why you support	project implementation? Yes No t/Oppose the project? Use the construction of the second of the second of the secon	2 duces
v() Briefly explain your answer.	to v), do you support t on why you support	project implementation? Yes No t/Oppose the project? Into the constant of the second Constant of the second Produced &	e duess
v() Briefly explain your answer.	to v), do you support t on why you support	project implementation? Yes No t/Oppose the project? to the training for pro- to great the project produced a	e duess
v() Briefly explain your answer.	to v), do you support to n why you support	project implementation? Yes No t/Oppose the project? to the the country for the second to she the second se	e duets
v() Briefly explain your answer.	to v), do you support t on why you support the construction of the	project implementation? Yes No t/Oppose the project? US STOR DE US HILL SIN CS STOR DE US PER END JUISSE	2 Auets
v() Briefly explain your answer Stuck Harden your opinion (ii) viii), Based on your opinion (ii) viii), Briefly give your commen CSS COMPANY	to v), do you support t on why you support	project implementation? Yes No t/Oppose the project? USA THE LAND BUN PER END JULIE	27 28 24 24 24 25 27 27 27 27 27 27 27 27 27 27 27 27 27
v() Briefly explain your answer.	to v), do you support t on why you support	project implementation? Yes No t/Oppose the project? Into the construction of the construction of the construction of the construction for the construction of the con	2 Buess
v() Briefly explain your answer.	to v), do you support t on why you support	project implementation? Yes No t/Oppose the project? In the the complete Complete the complete Produced &	e duess
v() Briefly explain your answer.	to v), do you support to n why you support	project implementation? Yes No t/Oppose the project? In the training for pro- to grad the project Produced	e duess
vi) Briefly explain your answer.	to v), do you support t on why you support an why you support an access an access and access and access acc	project implementation? Yes No t/Oppose the project? In the training the project Co. Stea theory of the pro- produced a	e duces
vi) Briefly explain your answer.	to v), do you support t on why you support the construction of the	project implementation? Yes No t/Oppose the project? Log the country of the series to such the series produced a	e duces

Project name Kannange Ordene Drien Vilage Cristin Sub location Katsurn Location wES1 Offerna Name Kithman OEDNISS also BORD USE DO DO 1130 2073 TEL 07 1924 1943 Email Khamadles 120 Offernal COLOR.). How long have you lived in this place? . 4.9. 723 ii). Were you aware of this project before? 7.423 Iii].In your opinion what are the likely impacts of the project on the community and the environment (Yes/No) $-\frac{1}{2}e_{s'}$ Impact category Positive Negative Comment Environmental will reduce Biodiversity й, Aesthetics(natural beauty! 2 111 Erosion hr. Water/Air pollution Social Ì HIV/AIDS AND STIS 13 remain the fame 莳 Culture iii) Gender relations will incase Economic i) Employment ii) Business and trade ~ 10 Value of Land 1 (w) Household income lv)In your opinion, are there better alternatives to the project? Category Technology Yes No Siting Raw material v). In your opinion, can the project design be improved ? Yes/ No vi) Briefly explain your answer (2011) 15 million (1000) 100 million (1000) 1000 million (vii). Based on your opinion (iii to v), do you support project implementation? Yes/ No-(ii) Eased on your opinion (iii to v), do you support project implementation? Yes Na-viii). Briefly give your comment on why you support/Oppose the project? <u>Emetical content of the state of the protect of the Property contents</u>, closels else <u>reduced</u>. <u>Interference of the protect of the product</u> <u>hypetics with the the protect of makels, shart i reduced</u> <u>Other halp iter to the protect of makels, shart i reduced</u>

Project nameRCCVVLQ	1140				4444	- 2
village kanana.	iub location.	Ka3:29	Locati	on Kagiwa		- 3
NameCXXIIITSQ	Charren	ID	no			1
TEL COMP IN	EE	mail		G	-	1.11
Date 364/1/201			gnature			2.3
Age		er cr	2 dat			
I). How long have you lived in this	t hefore3 fal	2 0.72	LINGTON	to Rusiness		
III) to your opinion what are the	ikely imparts	of the prois	ect on the	community and the environme	304	
NaciNol	west without	our trie briefs	in the second			
Impact rategory	Positive	Negative	Comme	nt		
Environmental	1					
L Riodiversity	V					
II. Aestheticsinatural		1.2			- L - Y	-
beauty)						- 2
iii. Erosion	-		No	comment		
Iv. Water/Air pollution	~					
Social					-	3 2
i) HIV/AIDS AND STIS	-					
ii) Culture						
iii) Gender relations	14					
Economic						
I) Employment	14				-	
Business and trade	1.1					
III) Value of Land	-					
N/ Household income		-				-
lide usur onlation, and there but	tor alternatio	ei to the or	steele			
Catagory	Vec.	nur to the pr	alees	No		
Technology					-	
Siting	12				-	3.1
Raw material						
FILM THREE SHIT						
villo your opinion, can the proj	ect design be	improved ?	Yes/No			
vi).Briefly explain your answer	-	100000				
					11-1	
vii). Basea on your opinion (iii t	o v), do you s	upport proje	ect imple	nentation? Yes/ No		
viii). Briefly give your comment	on why you:	support/Opp	nose the p	roject?		
BECARSE DJ	C. Link x3.	-First				

enel Project na 4.700 Sub location.... Village-Incation wahend Sten 100 mg Spin P 1 Name. LA SI O Email. Milke TEL - Riks The second secon (Yes/No) Positive Negative Comment impact category Environmental Biodiversity ١. Aesthetics(natural 16. beauty) 18. Erosion Water/Air pollution īψ. Social 43 HIV/AIDS AND STIS ŵ Cuiture 前 Gender relations Economic Employment. 1} Business and trade H) Value of Land iii) Household income îv) win your opinion, are there better alternatives to the project? Category Yes No Technology ١. Siting t Raw material v).in your opinion, can the project design be improved 7 (car No vi).Briefly explain your answer and In Crease vi) Briefly explain your answer 1 f ter vii). Based on your opinion (iii to v), do you support project implementation? Yes/No viii). Briefly give your comment on why you support/Oppose the project? poile will. tot upport en culli ann

Age 25 hand ord	Gende	er Min	R and	
I). How long have you lived in this	place?	Jargs	ST	
ii).Were you aware of this project	before?	of the surgicity of	the community and th	ne environment
(Yes/No)	wery unpacts	on the birdeet of		
Impact category	Positive	Negative Com	ment	
Environmental	~	1		
L. Biodiversity	12	-		
H. Aesthetics(natural				
Deauty)	V			
iv. Water/Air pollution	1-			
Social				
 HIV/AIDS AND STIs 	15			
ii) Culture	1			
III) Gender relations	100			
i) Employment	N	4		
ii) Business and trade	-			
iii) Value of Land	-			
iv) Household income	-			
	erena protecte			
ivin your opinion, are there bet	ter alternativ	ies to the project	No	
Tachonlow	10		110	
Sting				
Raw material	1			-
v) in your opinion, can the proj vi) Briefly explain your daswor	v), do you si on why you s	upport project in support/Oppose	no groß g algementation? Yes/ N he project? Cr () cr	0 100/
vii), assed on your opinion (iii o viii). Briefly give your comment	un	aty		

Appendix VIII : Minutes Of Community Validation Barazaon Proposed Project Planning

Held at Kamariga Beach West Uyoma Ward Rarieda Sub County on the Validation of Omena Drying plant on 4th March , 2020

1.1 Introduction

The Public Participation forum was held on 4th March 2020 at the Kamariga Beach in collaboration with the Department Of Agriculture (Kenya Climate Smart Agriculture Program) World Bank Departments of Health, Water (National Environmental Management Authority), The Directorate of Communication and Office of The Sub County Administrators.

1.2 Profile of Participants

The forum brought together members of the public from the ward The participants were drawn from different sectors which included Opinion leaders, Residents, Development Committee Representatives, Women Representatives, Youths Groups, People Living With Disability, The Clergy, Projects Management Committee Members and the Beach Management Unite.

1.3 Agenda and Format of Workshops

The fora commenced with a word of prayer and self-introductions were done subsequently. The Ward Administrators welcomed members and gave opening remarks followed by The Head of Public Participation who emphasized the importance of Public Participation before any project the public have the power to approve or reject as the activity is enacted in the constitution, She urged the Beach Management Unit (BMU) to stop lamenting on the project and attend all the public participation, budget cycle and civic education.

Agriculture: Mr. Ben from the Kenya Climate Smart Agriculture said this was a World Bank funded project, Ministry of Fisheries and County Government of Siaya they operates in three Sub Counties in Siaya this are Ugunja, Rarieda and Bondo the four main roles are

Increase Climate Smart Agriculture

Agro-wealth Advisory

Strengthening Research (KALRO)

Coordination and Emergency

Lead agencies in implementation comments

NEMA representative Mr. Samson Ochwiri as the organization responsible for environment we will be monitoring the project and approvals will be based on merits the building should be 30 metres from the water body. The previous building before new ESIA report (this has been solved and the project have the alternative site, recent evidence of the title is attached)

Engineer Agriculture: The first proposal was done in 2013 where solar dryer was to be installed. All the waste products will NOT be diverted to the water body, the dryer will be used specifically for its purpose, the plant can accommodate wherever quantity that comes from the lake and the neighbourhood lakes this includes the Omena and the big fish. General reactions

Been approved by NEMA, the factory should have its own management unit with the manager, Operators and Cleaners, the project should not stall because of luck of electricity,

The Chief Officer Agriculture on his part stated that the participation of the members of public in the validation of the design and modification of the project. This is a food processing plant which will compete in the international market this going to change the face of Kamariga and create employment opportunities and the standards should not be compromised

The CECM for Agriculture asked members of the public to approve the project as they proposed the project, The project will change the life style of the community and will stop

wastage of the Omena, lets fully own the project. The project has a timeline and if not utilized the partner can think otherwise. **Community Reactions and resolutions**

Iane	We need a barrier/wall around	The barrier and roofing are all captured in
Jane		
Gaunya	the project and the roof should	the BQ
	be replaced	
Edmond	Can the solid waste be	Solid waste will be sold to the local
Dima	converted to compost manure	community as manure
	or biogas	
Wycliffe	Any provision of power	This can only be done for the dryer which
Anguka	backup in case of power black	will finally have effect on the cost of
	outs	production
Victor	The location and position of	The public land is limited and the position
Anguka	the project as the lake gets full	was based on that but the NEMA expert to
		advise
Tom	Foundation of the building was	Machines will be put in the building thus
Allo	not stable what are the plans	engineers must work on the foundation
Auto	The PMC needs training on the	The supplier will conduct a two weeks
Chuo	management of the project	training on the operation and maintenance to
		the plant management
Achieng	Any criteria of employment of	ple with qualification based on the
Auma	staff working at the project	ent or The County Government can bring
		nt partners to run the plant



PLATE 12: COUNTY DIRECTOR OF PUBLIC PARTCIPATION ADDRESSING COMMUNITY



Plate 13: CHIEF OFFICER AGRICULTURE ADDRESSING COMMUNITY VALIDATION

Appendix IX : Key Informant (KI) Interview Schedule

The World Bank funded Kamola Water Pan - KCSAP Project is being implemented in Siaya County. As the project falls under the second schedule of EMCA, an Environmental Impact Assessment is to be undertaken. This is in addition to mainstreaming of World Bank ESS guidelines and operational procedures. As a Key informant resource your contribution in decision making and ultimate licensing of the project is critical. Please make your comments regarding the following

Are you aware and been involved in theof the project? Yes...... No......

What are the main statutes in your docket that touch on this project?

Are there any issues of concern (Social and environmental) that you think should be considered in this project?

In your Professional judgement does the implementation of the project require licensing/ permits from your department/ Authority/ Ministry?

Any other comment.

Thank You for your cooperation

Summary of key issues from KII

- ✓ There is no known heritage site in the project area however chance find procedures should be observed
- ✓ Not afforested area
- ✓ permit to construct processing plant in accordance with fisheries Act 26 to be processed though the construction may continue on NEMA approval
- \checkmark Effluent management and need for effluent license during operation
- \checkmark DOSH and registration of plant as work place

1. Purpose of the chance find procedure

The chance find procedure is a project-specific procedure that outlines actions required if previously unknown heritage resources, particularly archaeological resources, are encountered during project construction or operation. A Chance Find Procedure, is a process that prevents chance finds from being disturbed until an assessment by a competent specialist is made and actions consistent with the requirements are implemented.

Scope of the chance find procedure

This procedure is applicable to all activities conducted by the personnel, including contractors, that have the potential to uncover a heritage item/site. The procedure details the actions to be taken when a previously unidentified and potential heritage item/site is found during construction activities. Procedure outlines the roles and responsibilities and the response times required from both project staff, and any relevant heritage authority.

3. Induction/Training

All personnel, especially those working on earth movements and excavations, are to be inducted on the identification of potential heritage items/sites and the relevant actions for them with regards to this procedure during the Project induction and regular toolbox talks.

4. Chance find procedure

If any person discovers a physical cultural resource, such as (but not limited to) archaeological sites, historical sites, remains and objects, or a cemetery and/or individual graves during excavation or construction, the following steps shall be taken:

 Stop all works in the vicinity of the find, until a solution is found for the preservation of these artefacts, or advice from the relevant authorities is obtained;

 Immediately notify a foreman. The foreman will then notify the Resident/Supervising Engineer and the Environment Officer (EO)/Environmental Manager (EM);

3. Record details in Incident Report and take photos of the find;

4. Delineate the discovered site or area; secure the site to prevent any damage or loss of removable objects. In cases of removable antiquities or sensitive remains, a night guard shall be arranged until the responsible local authorities take over;

5. Preliminary evaluation of the findings by archaeologists. The archaeologist must make a rapid assessment of the site or find to determine its importance. Based on this assessment the appropriate strategy can be implemented. The significance and importance of the findings should be assessed according to the various criteria relevant to cultural heritage such as aesthetic, historic, scientific or research, social and economic values of the find;

6. Sites of minor significance (such as isolated or unclear features, and isolated finds) should be recorded immediately by the archaeologist, thus causing a minimum disruption to the work schedule

of the Contractor. The results of all archaeological work must be reported to the National Museums of Kenya (NMK), once completed.

7. In case of significant find the National Museums of Kenya (NMK) should be informed immediately and in writing within 7 days from the find.

8. The onsite archaeologist provides the NMK with photos, other information as relevant for identification and assessment of the significance of heritage items.

9. The NMK must investigate the fact within 2 weeks from the date of notification and provide response in writing.

10. Decisions on how to handle the finding shall be taken by the responsible authorities. This could include changes in the layout (such as when finding an irremovable remain of cultural or archaeological importance) conservation, preservation, restoration and salvage;

11. Construction works could resume only after permission is granted from the responsible authorities.

12. In case no response received within the 2 weeks' period mentioned above, this is considered as authorization to proceed with suspended construction works.

One of the main requirements of the procedure is record keeping. All finds must be registered. Photo log, copies of communication with decision making authorities, conclusions and recommendations/guidance, implementation reports - kept.

5. Additional information

Management options for archaeological site

- Site avoidance. If the boundaries of the site have been delineated attempt must be made to redesign the proposed development to avoid the site. (The fastest and most cost-effective management option)
- b) Mitigation. If it is not feasible to avoid the site through redesign, it will be necessary to sample it using data collection program prior to its loss. This could include surface collection and/or excavation. (The most expensive and time-consuming management option.)
- c) Site Protection. It may be possible to protect the site through the installation of barriers during the time of the development and/or possibly for a longer term. This could include the erection of high visibility fencing around the site or covering the site area with a geotextile and then capping it with fill. The exact prescription would be site- specific.

Management of replicable and non-replicable heritage

Different approaches for the finds apply to replicable and non-replicable heritage.

Replicable heritage¹

⁸ Replicable cultural heritage is defined as tangible forms of cultural heritage that can themselves be moved to another location or that can be replaced by a similar structure or natural features to which the cultural values can be transferred by appropriate measures. Archaeological or historical sites may be considered replicable where the particular eras and cultural values they represent are well represented by other sites and/or structures.

APPENDIX XII: Disclosure of ESMP at the Proposed Site

Sensitisation on Grievances Redress Mechanisms (GRM) at the proposed Kamariga omena drying Subproject, Siaya County on 6.6.2020

Background

The World Bank considers environmental impact assessment (ESIA) as one among a range of instruments for environmental assessment. These are procedures that ensure the proposed development is sustainable and environmentally sound. Although its operational policies and requirements vary in certain respects, the World Bank follows a relatively standard procedure for the preparation and approval of an environmental assessment study, which:

- Identifies and assesses potential risks and benefits based on proposed activities, relevant site features, consideration of natural/human environment, social and trans-boundary issues
- Compares environmental pros and cons of feasible alternatives
- Recommends measures to eliminate, offset, or reduce adverse environmental impacts to acceptable levels (sitting, design, technology offsets)
- Proposes monitoring indicators to implement mitigation measures
- Describes an institutional framework for environmental management and proposes relevant capacity building needs.

The specific objectives were:

- To establish a systematic approach to stakeholder engagement that will help Borrowers identify stakeholders and build and maintain a constructive relationship with them, in particular project-affected parties.
- To assess the level of stakeholder interest and support for the project and to enable stakeholders' views to be taken into account in project design and environmental and social performance.
- To promote and provide means for effective and inclusive engagement with projectaffected parties throughout the project life cycle on issues that could potentially affect them.
- To ensure that appropriate project information on environmental and social risks and impacts is disclosed to stakeholders in a timely, understandable, accessible and appropriate manner and format.
- To provide project-affected parties with accessible and inclusive means to raise issues and grievances, and allow Borrowers to respond to and manage such grievances. Scope of application Information disclosure

Process / approach

A public baraza brought together key stakeholders in the project site with the summary of the just concluded Environmental and social Management Framework (ESMF) providing the grounding for engagement. This was in line with World Bank requirement for early access to information, in informing project appraisal, and in a timeframe that enables meaningful consultations with stakeholders on project design: The mechanism, process or procedure will not prevent access to judicial or administrative remedies. Thewas disclosed to the project-affected parties about the grievance process in the course of its community engagement activities through a documentation responses. The community was thus appraised on the

- a. Stakeholder engagement process highlighting participation procedures
- b. Reporting procedures
- c. grievanceswhich was disclosed through Dholuo language and interpreted in English /Swahili where necessary
- d) Debrief on the summarised ESMP
 - \checkmark The purpose, nature and scale of the project;

- ✓ The duration of proposed project activities;
- ✓ Potential risks and impacts of the project on local communities, and the proposals for mitigating these, highlighting potential risks and impacts that might disproportionately affect vulnerable and disadvantaged groups and describing the various measures taken to avoid and minimize the risks.

2.1 Objectives of the grievance Redress mechanism

The general steps of the grievance process comprise: sensitization of target beneficiaries, constitution of institutional structures, capacity building, receipt of complaints; determining and implementing the redress action; verifying the redress action; amicable mediation and settlement; dissatisfaction and alternative actions; and documentation and communication. The GRM will also include timelines for implementation of redress process. The hierarchy to resolve complaints or grievances arising from implementation of the project will only become clear then. The benefitting communities will be assisted by the County Project Implementation Unit (CPCU) to establish the grievance committees. Grievances may arise from members of communities who are dissatisfied with any aspect of the implementation process. The management process will be elaborated after training

- ✓ to receive and facilitate resolution of such concerns and grievances. The ESMP form the core basis and baseline in the grievance process. Where feasible and suitable for the project, the grievance mechanism will utilize existing formal or informal grievance mechanisms, supplemented as needed with project-specific arrangements.
- ✓ address concerns promptly and effectively, in a transparent manner that is culturally appropriate and readily accessible to all project-affected parties, at no cost and without retribution
- ✓ provide affected people with avenues for making a complaint or resolving any dispute that may arise during the course of the implementation and determination of entitlements of compensation and implementation of the project
- ✓ ensure that appropriate and mutually acceptable redress actions are identified and implemented to the satisfaction of complainants; and avoid the need to resort to forceful actions and/or judicial proceedings as well as providing a transparent and accountable implementation process.
- ✓ Handling of grievances will be done in a culturally appropriate manner and be discreet, objective, sensitive and responsive to the needs and concerns of the project-affected parties. The organizational capacity and commitment will be finalised after training of the sub project management committees

Scope of the Grievances Redress Mechanisms (GRM) sensitisation

- ✓ The scope, scale and type of grievance mechanism required will be proportionate to the nature and scale of the potential The grievance mechanism may include the following:
- ✓ Different ways in which users can submit their grievances such as in person, by phone, text message, mail, e-mail or via a web site;
- \checkmark A log where grievances are registered in writing and maintained as a database;
- ✓ Publicly advertised procedures, setting out the length of time users can expect to wait for acknowledgement, response and resolution of their grievances;
- Transparency about the grievance procedure, governing structure and decision makers; and
- ✓ An appeals process (including the national judiciary) to which unsatisfied grievances may be referred when resolution of grievance has not been achieved. It was emphasised that mediation is an option where users are not satisfied with the proposed resolution.

4.0. Outcomes and recommendations

- To allow for meaningful and informed participation the County planning unit (CPU) will organise for in depth training of the sub project management committee on environmental issue vis avis the identification and setting of objectively verifiable indicators. This will allow for informed reporting and effective grievance mechanisms.
- The CPCU will continue to engage with, and provide information to, project-affected parties and other interested parties throughout the life cycle of the project, in a manner appropriate to the nature of their interests and the potential environmental and social risks and impacts of the project.
- The CPU will continue to conduct stakeholder engagement, and will build upon the channels of communication and engagement already established with stakeholders.
- In particular, the CPU will seek feedback from stakeholders on the environmental and social performance of the project, and the implementation of the mitigation measures through an Environmental Audit within the first year of completion.
- If there are significant changes to the project that result in additional risks and impacts, particularly where these will impact project-affected parties, the CPU will provide information on such risks and impacts and consult with project-affected parties as to how these risks and impacts will be mitigated. This will be by means of setting out any additional mitigation measures.
- Code of practice to mitigate the risks of sexual exploitation as it is a common practice along the beaches. The BMU will have zero tolerance policy on sexual exploitation

5.0. Monitoring and Evaluation

The terms of reference defining clear roles, responsibilities and authority as well as designate specific personnel to be responsible for the implementation and monitoring of stakeholder engagement activities and compliance will be elaborated during tailored training for the project management committee/implementation committee at **CPU** level

6.0. Annexes (Attendance List; Photos)

Key Issues highlighted as part of GRM

• Sexual exploitation (sex for fish), solutions and it's mainstreaming into implementation framework

a	Keny Keny	/a Climate Srr	lart Agrie	culture Proj	ect (KCSAP)	
and a	Republic of Kenya	Office of	the CPC	CU - Siaya		Kanya Climate 8 Agriculture Pro
1 . 1	Disclosure	REGISTRA	VTION FORM	Kommerce	T DATE C	.000
5	oncent betime	しころろ	1			
-	NAME	ORGANIZATION	DESIG	MOBILE	CONTACT	SIGN SIGN
	PROMPLE DRUKE	AGRIC	Var	65286186EO	nchinacontragmation	J.
-	TOUN NOLENZO	MMGT	hight	P312221124	Villension g. gmu	P-may
	MEIR DATAN AKINTI	WATER	Serior hydrologu4	0197031376	mar nongo ang na	ф
-	HERRINI MIHERO	E Duca Tion	Maven	2721363175	Linn Parchesarin	B
	Acrow D. Will	in LACCAP	990	Starstate	attaughtille enail ice	¥,
-	2minut outer	Mof	ef	agree 976000	Zerek ontes Equiles	Å
1	Churles educanter	14CSAP	CPD	22147(97	cluberte Szegm	clat
-		/				
		/				
12						

0	Ke	nya Climate \$	Smart Agri	culture Proje	sct (KCSAP)	
3 - 4	Republic of Kenya	Office	of the CP(CU - Siaya		Kenya Climate Snu Agriculture Projec
INI	The clesure	REGIS	P M	Komperico	DATE	.0000
T	oncean beting	main	an area		CONTACT	SIGN
-	NAME	ORGANIZATE	ON DESIG	MOBILE	EMAIL	(all
	PROMPED DRUKD	AGRIC	4CH	65286186EO	ncherd onthe almost	
ai	TOWA NOLENZE	MMST	high	P31222124	Volension g. gmu	Hours -
	MEIR DATAN AKINTI	WATER	Senior hydrologu4	0197031316	ayang290gmart.com	\$
+	HECRANI MIHERO	EDuca7	TON DRIVEN	2721363175	Linn Parchisadin	- mont
uń.	Atrave D. C.	Nin LACOND	2 dec	07225236	attaughtille ermail ic	THE A
10	ZANNARY ORLEYO	Mof	5	agent ando	Zaux one o cquilo	A.
1º	Churter Colucent	~ ILCSAI	P CPD	22123747198	cluberto S3@gm	L clak-
rá .		/				
m		/				
10.						

l